New York Journal of Mathematics

New York J. Math. 26 (2020) 378–445.

Two approaches to the use of unbounded operators in Feynman's operational calculus

Lance Nielsen

ABSTRACT. In this paper, we investigate two approaches to the use of unbounded operators in Feynman's operational calculus. The first involves using a functional calculus for unbounded operators introduced by A. E. Taylor in the paper [34]. The second approach uses analytic families of closed unbounded operators as discussed in [19]. For each approach, we discuss the essential properties of the operational calculus as well as continuity (or stability) properties. Finally, for the approach using the Taylor calculus, we discussion a connection between Feynman's operational calculus in this setting with the Modified Feynman Integral of M. L. Lapidus ([14, 20]).

Contents

1.	Introduction	378
2.	Necessary definitions and initial constructions	384
3.	Feynman's operational calculus on Banach algebras and some	
	stability theory	401
4.	FOC for unbounded operators using the Taylor calculus	423
5.	FOC for unbounded operators using analytic families	436
References		442

1. Introduction

The subject of this paper is, as the reader has no doubt deduced, Feynman's operational calculus – the formation of functions of several not necessarily commuting operators (originated in [5]). The aim of this paper, again as is clear from its title, is to introduce two methods of using unbounded operators in the abstract approach to Feynman's operational calculus (originated in [8,

Received August 29, 2019.

 $^{2010\} Mathematics\ Subject\ Classification.\ 44A99,\ 47A10,\ 47A13,\ 47A60,\ 47B15,\ 47B25,\ 47B48,\ 46G10.$

Key words and phrases. operational calculus, disentangling, unbounded operators, time-ordering, Taylor calculus, analytic families of operators, stability theory, modified Feynman integral.

9, 10, 11] and discussed in great detail in [15].) The approach to Feynman's operational calculus introduced by G. W. Johnson and B. Jefferies in the late 1990's allows only bounded operators to be used. Later work by Johnson, Jefferies and the author ([12]) introduced the (typically unbounded) generator of a (C_0) semigroup into the abstract approach to the operational calculus. The use of such generators has been, to now, the only way any unbounded operators could be used in the mathematically rigorous approach to the operational calculus. No other unbounded operators could be accommodated. (The use of unbounded semigroup generators was introduced in [12] in a way that was consistent with the abstract approach, but was somewhat informally done. The recent paper [30] incorporated the semigroup generator in a much more formal way via certain Banach algebras of entire functions.)

The first method introduced which enables unbounded operators to be used in the operational calculus comes from the 1951 paper [34] by A. E. Taylor. In this paper, Taylor introduces a functional calculus for unbounded operators using methods from complex analysis. His approach is similar to the approach he and N. Dunford (see [33], [2] and [3]) took when developing an operational calculus for bounded linear operators. This approach for bounded operators A was carried out by choosing the algebra of functions f(z) which are analytic on some open set containing the spectrum $\sigma(A)$ of A. The function f(A) was defined by

$$f(A) = \frac{1}{2\pi i} \int_{\Gamma} f(z)(zI - A)^{-1} dz,$$

the curve Γ being the boundary of a suitable bounded domain containing $\sigma(A)$. In [34], Taylor shows how to develop an operational calculus which can be applied to any closed linear operator whose spectrum does not cover the entire plane (i.e., the resolvent set of the operator must be nonempty). This operational calculus is defined in such a way that the contour integral above is part of the general theory for a closed operator. In fact, the corresponding expression for an unbounded closed operator B takes the form

$$f(B) = f(\infty)I + \frac{1}{2\pi i} \int_{\partial D} f(z)(zI - B)^{-1} dz,$$

where D is a particular domain in \mathbb{C} , called a Cauchy domain and $f(\infty)$ denotes the value of f at the point at infinity, where f is assumed to be nicely behaved on a neighborhood of $\sigma(B)$ which includes a neighborhood of infinity. The operator f(B) is shown to be a bounded linear operator.

In Section 3 of the present paper, we make use of the calculus of A. E. Taylor to obtain bounded linear operators from unbounded closed operators by using certain analytic functions f as above for each closed unbounded operator under consideration. Once we have the bounded linear operators $f_1(A_1), \ldots, f_n(A_n)$, we can then use the mathematically rigorous setting for Feynman's operational calculus in [15] to form functions

$$g(f_1(A_1),\ldots,f_n(A_n))$$

of the not necessarily commuting bounded operators $f_1(A_1), \ldots, f_n(A_n)$.

In Subsection 3.1, we apply the Feynman's operational calculus in the setting of the Taylor calculus to discuss a connection between the operational calculus and the Modified Feynman Integral of M. L. Lapidus (see [20] and [14]). In the "usual" setting of the operational calculus, this connection cannot be made; it requires the use of the Taylor calculus.

Finally, in Subsection 3.2, we investigate how the operational calculus in the Taylor calculus setting behaves when we have appropriately convergent sequences $\{f_{j,k}\}_{k=1}^{\infty}$, $j = 1, \ldots, n$, of analytic functions. It is shown that (to state the conclusion informally)

$$g(f_{1,k}(A_1),\ldots,f_{n,k}(A_n)) \longrightarrow g(f_1(A_1),\ldots,f_n(A_n)).$$

This result is in the spirit of the stability theory for Feynman's operational calculus developed by the author in the papers [15], [16], [21], [22], [23], [24], [25], [27], [28], [29] as well as [13], [14]. See Subsection 4.4 below for a short discussion of what we mean by "stability" in the context of Feynman's operational calculus. Also, see [15] for a detailed discussion of the stability theory for Feynman's operational calculus.

The second method we introduce in this paper – Section 5 – for the incorporation of unbounded operators into Feynman's operational calculus involves the use of Kato's analytic families of closed unbounded operators (see [19, Chapter 7]). Somewhat informally, an analytic family of closed unbounded operators is an analytic operator-valued function $T(z), z \in D_0 \subseteq \mathbb{C}$. The definition of an analytic family gives analytic operator-valued functions U(z) and V(z) on D_0 into $\mathcal{L}(X)$ (i.e., "boundedly analytic families" or "boundedly analytic function") for which

$$T(z)U(z) = V(z),$$

where U(z) is injective onto the domain $\mathcal{D}(T(z))$ of T(z). Consequently, we are able to deal with the unbounded operator T(z) by working with the bounded operator V(z). Hence, if we have analytic families $T_1(z), \ldots, T_n(z)$ on D_0 we obtain boundedly analytic operator-valued functions $U_1(z), V_1(z)$, $\ldots, U_n(z), V_n(z)$ where $T_j(z)U_j(z) = V_j(z)$. The boundedly analytic function $U_j(z)$ is injective onto the domain of $T_j(z)$. Once we have the boundedly analytic operator-valued functions $V_1(z), \ldots, V_n(z)$, we can then apply the rigorous setting for Feynman's operational calculus to obtain functions of the operators $V_1(z), \ldots, V_n(z)$ and so, indirectly, of the closed unbounded operators $T_1(z), \ldots, T_n(z)$. See Subsections 2.6 and 2.7 for a brief discussion of analytic families of operators. (For a detailed discussion, see [19, Chapter 7].)

As we did with the Taylor calculus setting, after we show how to use analytic families on the operational calculus, we turn to stability issues. Here, we choose sequences $\{T_{j,k}(z)\}_{k=1}^{\infty}$, $j = 1, \ldots, n$, of analytic families of closed unbounded operators. What is needed is the appropriate idea of convergence, as with unbounded operators we cannot consider norm convergence. The convergence we use is Kato's "generalized convergence" of closed unbounded operators [19, Chapter 4, Section 2]. If we have a sequence $\{T_n\}_{n=1}^{\infty}$ of closed and unbounded operators, we say that $\{T_n\}_{n=1}^{\infty}$ converges to the closed unbounded operator T if the "gap" between the graphs of T_n and the graph of T goes to zero as $n \to \infty$. (See Subsections 2.4 and 2.5 for a brief review of these ideas.) By using this idea of generalized convergence of sequences of unbounded closed operators, we obtain a stability-like result for the operational calculus in the setting of analytic families of operators. (We note that the generalized convergence of sequences of closed unbounded operators in question. However, this approach was not found to be as useful as the idea of the gap between graphs of the operators.)

We now turn to a brief discussion of Feynman's operational calculus. This operational calculus dates back to the work of R. P. Feynman, in particular to his 1951 paper [5]. In his 1951 paper, Feynman discussed the computation (or "formation") of functions of several not necessarily commuting operators using three heuristic "rules":

- (i) Attach time indices to the operators to specify the order of operators in products.
- (ii) With time indices attached, compute the functions of these operators by treating them as though they were commuting.
- (iii) Finally, "disentangle" the resulting expressions; that is, restore the conventional ordering of the operators.

Of the disentangling process, Feynman states [5, p. 110], "The process is not always easy to perform and, in fact, is the central problem of this operator calculus." One ought to note as well that Feynman did not try to supply rigorous (or even heuristic) proofs of his results.

The obvious question to ask, when considering the heuristic rules above, is how one goes about attaching time indices to operators. Of course, it is possible that one or more of the operators under consideration may come with time indices naturally attached, which would be the case, for instance, when operators of multiplication by time-dependent potentials are present and also in connection with the Heisenberg representation in quantum mechanics. However, if an operator does not depend on time, as happens most often in quantum mechanics and in the mathematical literature, we need a mechanism for attaching time indices to such operators. Given a time independent operator A, Feynman, nearly without exception, used Lebesgue measure to attach time indices by writing

$$A = \frac{1}{t} \int_0^t A(s) \, ds,$$

where $A(s) \equiv A$ for all $s \in [0, t]$. Even though it appears artifical, this way of attaching time indices is extraordinarily useful and is crucial to the

approach we will take to Feynman's operational calculus. (See [15], [12], [8, 9, 10, 11].)

To familiarize the reader with the basic ideas of the operational calculus, we will take the time to present two elementary examples. In these examples, X will be a Banach space with norm $\|\cdot\|_X$ and $\mathcal{L}(X)$ will denote the Banach space of bounded linear operators on X equipped with the norm

$$||A||_{\mathcal{L}(X)} = \sup \{ ||Ax||_X : x \in X, ||x||_X \le 1 \}$$

Example 1.1. Take f(x,y) = xy and let $A, B \in \mathcal{L}(X)$. We associate Lebesgue measure ℓ on [0,1] to both operators. More specifically,

$$A = \int_0^1 A(s) \, ds$$
 and $B = \int_0^1 B(s) \, ds$,

where $A(s) \equiv A$ and $B(s) \equiv B$, for all $s \in [0, 1]$. We will compute f(A, B) labeling the result as $f_{\ell,\ell}(A, B)$ to make explicit the role that Lebesgue measure is playing. We have, computing heuristically,

$$\begin{split} f_{\ell,\ell}(A,B) &= \left\{ \int_0^1 A(s) \, ds \right\} \left\{ \int_0^1 B(s) \, ds \right\} \\ &= \int_0^1 \int_0^1 A(s_1) B(s_2) \, ds_1 ds_2 \\ &= \int_{\{(s_1,s_2): s_1 < s_2\}} B(s_2) A(s_1) \, ds_1 ds_2 + \\ &\int_{\{(s_1,s_2): s_2 < s_1\}} A(s_1) B(s_2) \, ds_1 ds_2 \\ &= \frac{1}{2} BA + \frac{1}{2} AB = \frac{1}{2} \left(BA + AB \right). \end{split}$$

The first equality above follows from how we've attached time-indices to the operators. The second equality is the result of writing the product of the integrals as an iterated integral over $[0, 1]^2$. It is after the third equality that the time-ordering is carried out. We write

$$[0,1]^2 = \{(s_1, s_2) : 0 < s_1 < s_2 < 1\} \cup \{(s_1, s_2) : 0 < s_2 < s_1 < 1\} \cup \{(s_1, s_2) : s_1 = s_2\}.$$

Since ℓ is a continuous measure, $(\ell \times \ell) (\{(s_1, s_2) : s_1 = s_2\}) = 0$; this is why there is no third integral in the computation above. In the union just above, because the time index s_1 is smaller (or *earlier*) than s_2 , the operator product is written as $B(s_2)A(s_1)$. On the second set in the union, because the time index s_2 is smaller (or *earlier*) than s_1 , the operator product is written as $A(s_1)B(s_2)$. (We are using "rule" (*ii*) here.) Finally, we evaluate the integrals and obtain the last expressions.

Example 1.2. We take f(x, y) = xy and $A, B \in \mathcal{L}(X)$. We associate Lebesgue measure ℓ on [0, 1] to A and we associate the Dirac point mass δ_{τ} , $\tau \in (0, 1)$, to B. So,

$$A = \int_0^1 A(s) \, ds$$
 and $B = \int_0^1 B(s) \, \delta_\tau(ds)$,

where, as before, $A(s) \equiv A$ and $B(s) \equiv B$, for all $s \in [0, 1]$. Following Feynman's rules leads to

$$\begin{split} f_{\ell,\delta_{\tau}}(A,B) &= \left\{ \int_{0}^{1} A(s) \, ds \right\} \left\{ \int_{0}^{1} B(s) \, \delta_{\tau}(ds) \right\} \\ &= \left\{ \int_{(0,\tau)} A(s) \, ds + \int_{(\tau,1)} A(s) \, ds \right\} \left\{ \int_{\{\tau\}} B(s) \, \delta_{\tau}(ds) \right\} \\ &= \left\{ \int_{\{\tau\}} B(s) \, \delta_{\tau}(ds) \right\} \left\{ \int_{(0,\tau)} A(s) \, ds \right\} + \\ &\qquad \left\{ \int_{(\tau,1)} A(s) \, ds \right\} \left\{ \int_{\{\tau\}} B(s) \, \delta_{\tau}(ds) \right\} \\ &= B \left[\tau A \right] + \left[(1-\tau)A \right] B = \tau B A + (1-\tau) A B. \end{split}$$

We follow Feynman's rules in the same way as in the first example; however, there are differences here. Since δ_{τ} is supported on $\{\tau\}$, we break the integral for A at τ , leading to the expression after the second equality. To obtain the expression after the third equality, we follow Feynman's second rule as well as the time-ordering. Since τ comes after all time-indices in $(0, \tau)$ and before all time-indices in $(\tau, 1)$, we obtain the expression after the third equality. Evaluating the integrals leads to the expression after the fourth equality.

The two examples presented above illustrate the essential ideas of the operational calculus, but the calculations carried out in both examples are not at all rigorous. To see how Feynman's ideas can be made mathematically rigorous, we briefly sketch the approach taken by G. W. Johnson and B. Jefferies (initiated in the late 1990's, appearing in print in [8, 9, 10, 11]). For a more detailed discussion, see Subsections 2.1 and 2.2 below. We take A_1, \ldots, A_n to be noncommuting bounded linear operators on a Banach space X. Associate to each operator A_j a time-ordering measure λ_j on [0, 1] (a probability measure on [0, 1]). We may then write

$$A_j = \int_{[0,1]} A_j(s) \,\lambda_j(ds),$$

where $A_j(s) \equiv A_j$ for all $s \in [0, 1]$. Let $f(z_1, \ldots, z_n)$ be a function analytic on some polydisk P centered at the origin in \mathbb{C}^n and write

$$f(z_1,...,z_n) = \sum_{m_1,...,m_n=0}^{\infty} a_{m_1,...,m_n} z_1^{m_1} \cdots z_n^{m_n}.$$

Define

$$P^{m_1,\ldots,m_n}(z_1,\ldots,z_n) := z_1^{m_1}\cdots z_n^{m_n}.$$

With the examples above in mind, we can compute, using Feynman's rules, the disentangling of $P^{m_1,\ldots,m_n}(A_1,\ldots,A_n)$ using the time-ordering directions supplied by the time-ordering measures $\lambda_1,\ldots,\lambda_n$. One expects (and this is indeed the case) that the disentangled operator corresponding to the function f to take the form

$$f_{\lambda_1,\dots,\lambda_n}(A_1,\dots,A_n) = \sum_{m_1,\dots,m_n=0}^{\infty} a_{m_1,\dots,m_n} P^{m_1,\dots,m_n}_{\lambda_1,\dots,\lambda_n} \left(A_1,\dots,A_n\right).$$

Obviously, questions of convergence arise, but the series above will turn out to converge in operator norm on $\mathcal{L}(X)$. It is also the case that the disentangled monomial

$$P^{m_1,\ldots,m_n}_{\lambda_1,\ldots,\lambda_n}(A_1,\ldots,A_n)$$

is written as a sum of time-ordered products, much as we saw in the simple examples above. Jefferies and Johnson made these ideas mathematically rigorous by first supplying a "commutative world" – the disentangling algebra – in which the time-ordering (or, disentangling) calculations can be done in a rigorous way. (The disentangling algebra is, in fact, a commutative Banach algebra; see Subsection 2.1.1.) Once these computations are complete, the result is then mapped to the noncommuting environment of $\mathcal{L}(X)$ using the so-called disentangling map, see Subsection 2.2. It is this abstract approach which will be followed in the current paper. A comprehensive discussion of this approach can be found in [15] (as well as in the fundamental papers [8, 9, 10, 11]. Furthermore, the paper [12] (see also [21]) extends these ideas to the setting where the operators are time-dependent (or, operator-valued functions).

2. Necessary definitions and initial constructions

In this section we take the time in, Subsections 2.1 and 2.2, to outline the essential ideas of the abstract approach to Feynman's operational calculus which is used in this paper. In Subsection 2.3, we outline the operator theory which will play a role in our investigations. Subsection 2.4 gives a short discussion of the Taylor calculus which was introduced in the 1951 paper [34]. This calculus will play a crucial role in Section 3. In Subsection 2.5, Kato's *generalized convergence* of sequences of closed operators is discussed. The main properties of analytic families of operators are introduced in Subsections 2.6 and 2.7. Finally, in Subsection 2.8 we present a detailed discussion of the interplay between analytic families of (closed unbounded) operators and generalized convergence. This discussion will be used to obtain a stability-like result for the operators (see Subsection 4.1).

2.1. The operational calculus. In this sub-section, we describe, briefly, Feynman's operational calculus as developed in [8, 9, 10, 11], [12], [17] and [21]. A detailed discussion of this abstract approach to the operational calculus can be found in [15].

2.1.1. The commutative Banach algebras. In this subsection we will define the commutative Banach algebras that will be of use in our approach to Feynman's operational calculus. To begin, we will take X to be a separable Banach space and we will denote by $\mathcal{L}(X)$ the Banach space of bounded linear operators on X. Let $A_j : [0,T] \to \mathcal{L}(X)$, $j = 1, \ldots, n$, be strongly measurable; i.e., $A_j^{-1}(E)$ is a Borel set in [0,T] for every strongly open $E \subset \mathcal{L}(X)$. Associate to each $A_j(\cdot)$, $j = 1, \ldots, n$, a Borel probability measure λ_j on [0,T] and decompose λ_j as $\lambda_j = \mu_j + \eta_j$, where μ_j is a continuous Borel measure on [0,T] and η_j is a finitely supported discrete measure on [0,T]. We refer to the measures λ_j , $j = 1, \ldots, n$, as time-ordering measures. As mentioned in the introduction, the time-ordering measures attached to a given operator-valued function (or operator) serve to determine where (or when) the operator acts in operator products. Let $\{\tau_1, \ldots, \tau_h\}$ be the union of the supports of the discrete measures η_1, \ldots, η_n and assume that

$$0 < \tau_1 < \dots < \tau_h < T. \tag{2.1}$$

We can then write

$$\eta_j = \sum_{i=1}^h p_{ji} \delta_{\tau_i} \tag{2.2}$$

for each j = 1, ..., n. Note that, with this definition, it may be that many of the p_{ji} are zero. However, this observation will not play any role in this paper.

With our operator-valued functions and their associated time-ordering measures in hand, we make the assumption that

$$r_{j} := \int_{[0,T]} \|A_{j}(s)\|_{\mathcal{L}(X)} \lambda_{j}(ds)$$
$$= \int_{[0,T]} \|A_{j}(s)\|_{\mathcal{L}(X)} \mu_{j}(ds) + \sum_{i=1}^{h} p_{ji} \|A_{j}(\tau_{i})\|_{\mathcal{L}(X)} < \infty$$
(2.3)

for each j = 1, ..., n. The first commutative Banach algebra we require is defined as follows.

Definition 2.1. Let R_1, \ldots, R_n be positive real numbers. We define $\mathbb{A}(R_1, \ldots, R_n)$ to be the family of all functions analytic on the open polydisk

$$P_{R_1,\ldots,R_n} := \{(z_1,\ldots,z_n) \in \mathbb{C}^n : |z_j| < R_j, j = 1,\ldots,n\},\$$

and continuous on its boundary. For $f \in \mathbb{A}(R_1, \ldots, R_n)$, we define

$$||f||_{\mathbb{A}} := \sum_{m_1,\dots,m_n=0}^{\infty} |a_{m_1,\dots,m_n}| R_1^{m_1} \cdots R_n^{m_n}$$
(2.4)

where

$$f(z_1, \dots, z_n) = \sum_{m_1, \dots, m_n=0}^{\infty} a_{m_1, \dots, m_n} z_1^{m_1} \cdots z_n^{m_n}$$
(2.5)

is the Taylor series expansion of f at the origin in \mathbb{C}^n . The family $\mathbb{A}(R_1, \ldots, R_n)$ is a Banach algebra under pointwise operations. (See [15, Chapter 2].) We will often denote this algebra by \mathbb{A} .

With the real numbers r_1, \ldots, r_n from (2.3), we construct the commutative Banach algebra $\mathbb{A}(r_1, \ldots, r_n)$. We now define a second, but closely related, commutative Banach algebra.

Definition 2.2. The disentangling algebra $\mathbb{D}(A_1(\cdot)^{\sim}, \ldots, A_n(\cdot)^{\sim})$ is defined to be the family of all expressions

$$f(A_1(\cdot)^{\sim}, \dots, A_n(\cdot)^{\sim}) := \sum_{m_1, \dots, m_n=0}^{\infty} a_{m_1, \dots, m_n} (A_1(\cdot)^{\sim})^{m_1} \cdots (A_n(\cdot)^{\sim})^{m_n}$$
(2.6)

for which

$$||f||_{\mathbb{D}} := \sum_{m_1,\dots,m_n=0}^{\infty} |a_{m_1,\dots,m_n}| r_1^{m_1} \cdots r_n^{m_n} < \infty.$$
(2.7)

The objects $A_j(\cdot)^{\sim}$ are formal commuting objects which take the place of the operator functions $A_j(\cdot)$. All function- and operator-theoretic properties of $A_j(\cdot)$ are discarded from $A_j(\cdot)^{\sim}$ with the exception of the operator norm (or L^1 -norm).

This family is a commutative Banach algebra under pointwise operations and with norm (2.7). (See [15, Chapter 2].) We will often denote this algebra by \mathbb{D} .

Remark 2.3. We note that we make no assumptions that the operator-valued functions (or operators) are linearly independent; however, we do assume that the formal objects $A_j(\cdot)^{\sim}$, $j = 1, \ldots, n$, are linearly independent.

Also, upon reflection, it becomes clear that the commutative Banach algebras \mathbb{A} and \mathbb{D} are, in fact, isometrically isomorphic. A detailed discussion of these algebras can be found in [15, Chapters 2 and 6].

It is the disentangling algebra above that will supply the "commutative world" which will allow us to apply Feynman's heuristic rules in a rigorous fashion. More specifically, we carry out the time-ordering computations in the disentangling algebra and then map the resulting expression into $\mathcal{L}(X)$. The resulting operator is referred to as the *disentangled operator*. We make

this very brief remark precise as follows. First, to streamline our notation, for $m_1, \ldots, m_n \in \mathbb{N} \cup \{0\}$, we define

$$P^{m_1,\ldots,m_n}(z_1,\ldots,z_n) := z_1^{m_1}\cdots z_n^{m_n};$$

i.e., $P^{m_1,\ldots,m_n}(z_1,\ldots,z_n)$ is the monomial with exponents m_1,\ldots,m_n . Given $f \in \mathbb{D}$, we may therefore write

$$f(A_1(\cdot)^{\sim}, \dots, A_n(\cdot)^{\sim}) = \sum_{m_1, \dots, m_n=0}^{\infty} a_{m_1, \dots, m_n} P^{m_1, \dots, m_n} (A_1(\cdot)^{\sim}, \dots, A_n(\cdot)^{\sim}),$$

where f has the Taylor series given in (2.5).

2.1.2. Sets of time indices. To carry out the necessary time-ordering calculations required by Feynman's rules, we need to have certain ordered sets of time indices. First, given $t \in [0, T]$, and $m \in \mathbb{N}$, we define

$$\Delta_m(t) = \Delta_m^t := \{ (s_1, \dots, s_m) \in [0, T]^m : 0 < s_1 < \dots < s_m < t \}.$$
(2.8)

Next, given $t \in [0,T]$, $m \in \mathbb{N}$ and $\pi \in S_m$, the group of permutations on m objects, we define

$$\Delta_m^t(\pi) := \left\{ (s_1, \dots, s_m) \in [0, T]^m : 0 < s_{\pi(1)} < \dots < s_{\pi(m)} < t \right\}.$$
 (2.9)

As is shown in [14] and [15],

$$[0,t]^m = \bigcup_{\pi \in S_m} \Delta_m^t(\pi),$$

where the union is a disjoint union. Indeed, if $\lambda_1, \ldots, \lambda_n$ are continuous measures on [0, T], then the equality above holds up to a set of $\lambda_1^{m_1} \times \cdots \times \lambda_n^{m_n}$ - measure zero. (We will use ν^k to denote the k-fold product measure $\nu \times \cdots \times \nu$ and, when the context is clear, often omit the superscript t or T from $\Delta_m^T, \Delta_m^t(\pi)$, etc.)

To accommodate the discrete measures, we need one further set of timeindices which lets us deal with the presence of time-ordering measures with nonzero discrete parts. Let $\tau_1, \ldots, \tau_h \in [0, T], 0 < \tau_1 < \cdots < \tau_h < T$. Let $q \in \mathbb{N}$ and let $\theta_1, \ldots, \theta_{h+1} \in \mathbb{N}$ be such that $\theta_1 + \cdots + \theta_{h+1} = q$. We define

$$\Delta_{q;\theta_1,\dots,\theta_{h+1}} := \{ (s_1,\dots,s_q) \in [0,T]^q : 0 < s_1 < \dots < s_{\theta_1} < \tau_1 < s_{\theta_1+1} < \dots < s_{\theta_1+\theta_2} < \tau_2 < s_{\theta_1+\theta_2+1} < \dots < s_{\theta_1+\dots+\theta_h} < \tau_h < (2.10) \\ s_{\theta_1+\dots+\theta_h+1} < \dots < s_q < T \}.$$

The set $\Delta_{q;\theta_1,\ldots,\theta_{h+1}}(\pi)$, for $\pi \in S_q$, is defined in the same way as (2.9). Also, if ν is a continuous probability measure on [0, T], then

$$\Delta_q = \bigcup_{\theta_1 + \dots + \theta_{h+1} = q} \Delta_{q;\theta_1,\dots,\theta_{h+1}}$$

up to a set of ν^q -measure zero (see [14, Lemma 15.2.7]) and the union above is a disjoint union.

2.1.3. Definitions concerning operators and formal objects. We need the following definitions to carry out the time-ordering calculations required by Feynman's rules. Recall that for operator-valued functions $A_1(\cdot), \ldots, A_n(\cdot)$ we discard all operator-theoretic properties of these functions and create formal commuting objects $A_1(\cdot)^{\sim}, \ldots, A_n(\cdot)^{\sim}$. We take, for every $j = 1, \ldots, n, A_j(s)^{\sim} \equiv A_j(\cdot)^{\sim}$ for all $s \in [0, T]$ and define

$$C_{j}(s)^{\sim} := \begin{cases} A_{1}(s)^{\sim} & \text{if } j \in \{1, \dots, m_{1}\}, \\ A_{2}(s)^{\sim} & \text{if } j \in \{m_{1} + 1, \dots, m_{1} + m_{2}\}, \\ \vdots \\ A_{n}(s)^{\sim} & \text{if } j \in \{m_{1} + \dots + m_{n-1} + 1, \dots, m_{1} + \dots + m_{n}\}. \end{cases}$$

$$(2.11)$$

We also define

$$C_{j}(s) := \begin{cases} A_{1}(s) & \text{if } j \in \{1, \dots, m_{1}\}, \\ A_{2}(s) & \text{if } j \in \{m_{1} + 1, \dots, m_{1} + m_{2}\}, \\ \vdots \\ A_{n}(s) & \text{if } j \in \{m_{1} + \dots + m_{n-1} + 1, \dots, m_{1} + \dots + m_{n}\}. \end{cases}$$

$$(2.12)$$

We note that, if the operator-valued functions $A_j(\cdot)$ are constant-valued; i.e., $A_j(s) \equiv A_j$ for all $s \in [0, T]$, then the definitions still apply, with (2.12) having only the operators A_1, \ldots, A_n on the right-hand side.

2.2. Applying Feynman's rules – disentangling. We will continue with the strongly measurable operator-valued functions $A_j : [0,T] \to \mathcal{L}(X)$ (X a separable Banach space) which appeared in Subsection 2.1.1. Associate with each $A_j(\cdot)$, j = 1, ..., n, the Borel probability measure $\lambda_j = \mu_j + \eta_j$, again just as in Subsection 2.1.1. (We will use the same notation as in Section 2.1.1.)

To begin, we let $f \in \mathbb{D}$ and write (as in Definition 2.2)

$$f(A_1(\cdot)^{\sim}, \dots, A_n(\cdot)^{\sim}) = \sum_{m_1, \dots, m_n=0}^{\infty} a_{m_1, \dots, m_n} P^{m_1, \dots, m_n} (A_1(\cdot)^{\sim}, \dots, A_n(\cdot)^{\sim})$$

We will carry out the time-ordering of the monomial first. However, we will state the result without a detailed calculation as the interested reader can consult [15, Chapters 2 and 8] as well as [17] for a detailed exposition of the time-ordering calculations.

Proposition 2.4. Using the notation developed in this section and given $m_1, \ldots, m_n \in \mathbb{N} \cup \{0\},\$

$$P^{m_1,\dots,m_n}(A_1(\cdot)^{\sim},\dots,A_n(\cdot)^{\sim})$$

$$= \sum_{q_{11}+q_{12}=m_1} \cdots \sum_{q_{n1}+q_{n2}=m_n} \left(\frac{m_1!\cdots m_n!}{q_{11}!q_{12}!\cdots q_{n1}!q_{n2}!}\right) \sum_{j_{11}+\dots+j_{1h}=q_{12}} \cdots$$
(2.13)

$$\sum_{j_{n1}+\dots+j_{nh}=q_{n2}} \left(\frac{q_{12}!\cdots q_{n2}!}{j_{11}!\cdots j_{1h}!\cdots j_{n1}!\cdots j_{nh}!} \right) \sum_{\theta_{1}+\dots+\theta_{h+1}=q_{11}+\dots+q_{n1}} \cdots \sum_{\pi \in S_{q_{11}+\dots+q_{n1}\Delta_{q_{11}+\dots+q_{n1};\theta_{1},\dots,\theta_{h+1}}(\pi)} \int C_{\pi(q_{11}+\dots+q_{n1})} \left(s_{\pi(q_{11}+\dots+q_{n1})} \right)^{\sim} \cdots \right) C_{\pi(\theta_{1}+\dots+\theta_{h}+1)} \left(s_{\pi(\theta_{1}+\dots+\theta_{h}+1)} \right)^{\sim} \left[\prod_{\beta=0}^{n-1} \left\{ p_{\beta h}A_{n-\beta}(\tau_{h})^{\sim} \right\}^{j_{n-\beta,h}} \right] \cdot C_{\pi(\theta_{1}+\dots+\theta_{h})} \left(s_{\pi(\theta_{1}+\dots+\theta_{h})} \right)^{\sim} \cdots C_{\pi(\theta_{1}+1)} \left(s_{\pi(\theta_{1}+1)} \right)^{\sim} \cdot \left[\prod_{\beta=0}^{n-1} \left\{ p_{\beta 1}A_{n-\beta}(\tau_{1})^{\sim} \right\}^{j_{n-\beta,1}} \right] \cdot C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots C_{\pi(1)} \left(s_{\pi(1)} \right)^{\sim} \left(\mu_{1}^{q_{11}} \times \cdots \times \mu_{n}^{q_{n1}} \right) \left(ds_{1},\dots, ds_{q_{11}+\dots+q_{n1}} \right) \cdot C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots C_{\pi(1)} \left(s_{\pi(1)} \right)^{\sim} \left(\mu_{1}^{q_{11}} \times \cdots \times \mu_{n}^{q_{n1}} \right) \left(ds_{1},\dots, ds_{q_{11}+\dots+q_{n1}} \right) \cdot C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots C_{\pi(1)} \left(s_{\pi(1)} \right)^{\sim} \left(\mu_{1}^{q_{11}} \times \cdots \times \mu_{n}^{q_{n1}} \right) \left(ds_{1},\dots, ds_{q_{11}+\dots+q_{n1}} \right) \cdot C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots C_{\pi(1)} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots \times \mu_{n}^{q_{n1}} \right) \left(ds_{1},\dots, ds_{q_{11}+\dots+q_{n1}} \right) \cdot C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots C_{\pi(1)} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots \times \mu_{n}^{q_{n1}} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})} \right)^{\sim} \cdots \times \mu_{n}^{q_{n1}} \left(s_{\pi($$

It follows at once that, given $f \in \mathbb{D}$ with its series representation (2.6),

$$f(A_{1}(\cdot)^{\sim},\ldots,A_{n}(\cdot)^{\sim}) = \sum_{m_{1},\ldots,m_{n}=0}^{\infty} a_{m_{1},\ldots,m_{n}} \sum_{q_{11}+q_{12}=m_{1}}\cdots\sum_{q_{n1}+q_{n2}=m_{n}} \left(\frac{m_{1}!\cdots m_{n}!}{q_{11}!q_{12}!\cdots q_{n1}!q_{n2}!}\right) \cdot \sum_{j_{11}+\cdots+j_{1h}=q_{12}}\cdots\sum_{j_{n1}+\cdots+j_{nh}=q_{n2}} \left(\frac{q_{12}!\cdots q_{n2}!}{j_{11}!\cdots j_{1h}!\cdots j_{n1}!\cdots j_{nh}!}\right) \cdot \sum_{\theta_{1}+\cdots+\theta_{h+1}=q_{11}+\cdots+q_{n1}}\sum_{\pi\in S_{q_{11}+\cdots+q_{n1}\Delta_{q_{11}+\cdots+q_{n1};\theta_{1},\ldots,\theta_{h+1}}} \int \cdot (\pi(q_{11}+\cdots+q_{n1}))^{\sim}\cdots C_{\pi(\theta_{1}+\cdots+\theta_{h}+1)} \left(s_{\pi(\theta_{1}+\cdots+\theta_{h}+1)}\right)^{\sim} \cdot \left[\prod_{\beta=0}^{n-1} \{p_{\beta h}A_{n-\beta}(\tau_{h})^{\sim}\}^{j_{n-\beta,h}}\right] C_{\pi(\theta_{1}+\cdots+\theta_{h})} \left(s_{\pi(\theta_{1}+\cdots+\theta_{h})}\right)^{\sim}\cdots C_{\pi(\theta_{1}+1)} \left(s_{\pi(\theta_{1}+1)}\right)^{\sim} \left[\prod_{\beta=0}^{n-1} \{p_{\beta 1}A_{n-\beta}(\tau_{1})^{\sim}\}^{j_{n-\beta,1}}\right] C_{\pi(\theta_{1})} \left(s_{\pi(\theta_{1})}\right)^{\sim}\cdots C_{\pi(\eta_{1})} \left(s_{\pi(\eta_{1})}\right)^{\sim}\cdots C_{\pi(\eta_{1})} \left(s_{\pi(\eta_{1})}\right)^{\sim$$

With the time-ordering in the disentangling algebra at our disposal, we can now define the *disentangling map* which will take these time-ordered expressions from the commutative setting of the disentangling algebra \mathbb{D} to the noncommutative setting of $\mathcal{L}(X)$. In fact, using the time-ordering from Proposition 2.4 and the definition of the operators $C_j(s)$ from (2.12) it is straightforward to define the disentangling map.

Definition 2.5. Using notation of this section, we define the disentangling map

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n}: \mathbb{D}\left(A_1(\cdot)^{\sim},\dots,A_n(\cdot)^{\sim}\right) \to \mathcal{L}(X)$$

as follows. For $m_1, \ldots, m_n \in \mathbb{N} \cup \{0\}$, we define the action of the disentangling map on

$$P^{m_1,\ldots,m_n}\left(A_1(\cdot)^{\sim},\ldots,A_n(\cdot)^{\sim}\right)$$

by

$$\begin{aligned} \mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}P^{m_{1},\dots,m_{n}}\left(A_{1}(\cdot)^{\sim},\dots,A_{n}(\cdot)^{\sim}\right) \\ &= P_{\lambda_{1},\dots,\lambda_{n}}^{m_{1},\dots,m_{n}}\left(A_{1}(\cdot),\dots,A_{n}(\cdot)\right) \\ &= \sum_{q_{11}+q_{12}=m_{1}}\cdots\sum_{q_{n1}+q_{n2}=m_{n}}\left(\frac{m_{1}!\cdots m_{n}!}{q_{11}!q_{12}!\cdots q_{n1}!q_{n2}!}\right)\sum_{j_{11}+\dots+j_{1h}=q_{12}}\cdots \\ &\sum_{j_{n1}+\dots+j_{nh}=q_{n2}}\left(\frac{q_{12}!\cdots q_{n2}!}{j_{11}!\cdots j_{1h}!\cdots j_{n1}!\cdots j_{nh}!}\right)\sum_{\theta_{1}+\dots+\theta_{h+1}=q_{11}+\dots+q_{n1}}\cdots \\ &\sum_{\pi\in S_{q_{11}+\dots+q_{n1}\Delta_{q_{11}+\dots+q_{n1};\theta_{1},\dots,\theta_{h+1}}(\pi)}\int C_{\pi(q_{11}+\dots+q_{n1})}\left(s_{\pi(q_{11}+\dots+q_{n1})}\right)\cdots \end{aligned}$$
(2.15)

$$C_{\pi(\theta_{1}+\dots+\theta_{h}+1)}\left(s_{\pi(\theta_{1}+\dots+\theta_{h}+1)}\right)\left[\prod_{\beta=0}^{n-1} \{p_{\beta h}A_{n-\beta}(\tau_{h})\}^{j_{n-\beta,h}}\right].$$

$$C_{\pi(\theta_{1}+\dots+\theta_{h})}\left(s_{\pi(\theta_{1}+\dots+\theta_{h})}\right)\cdots C_{\pi(\theta_{1}+1)}\left(s_{\pi(\theta_{1}+1)}\right)\left[\prod_{\beta=0}^{n-1} \{p_{\beta 1}A_{n-\beta}(\tau_{1})\}^{j_{n-\beta,1}}\right]$$

$$C_{\pi(\theta_{1})}\left(s_{\pi(\theta_{1})}\right)\cdots C_{\pi(1)}\left(s_{\pi(1)}\right)\left(\mu_{1}^{q_{11}}\times\cdots\times\mu_{n}^{q_{n1}}\right)\left(ds_{1},\dots,ds_{q_{11}+\dots+q_{n1}}\right).$$

$$C_{\pi(\theta_{1}+\dots+\theta_{n})}\left(s_{\pi(\theta_{1}+\dots+\theta_{n})}\right)\cdots C_{\pi(\theta_{n}+1)}\left(s_{\pi(\theta_{n}+1)}\right)\left(s_{\pi(\theta_{n}+1}\right)\left(s_{\pi(\theta_{n}+1)}\right)\left(s_{\pi(\theta_{n}+1)}\right)\left(s_{\pi(\theta_{n}+1)}\right)\left(s_{\pi(\theta_{n}+1}\right)\left(s_{\pi(\theta_{n}+1)}\right)\left(s_{\pi(\theta_{n}+1}\right)\left($$

Given $f \in \mathbb{D}$ with the representation (2.6), we define the action of the disentangling map on this element by

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n} f\left(A_1(\cdot)^{\sim},\dots,A_n(\cdot)^{\sim}\right) = f_{\lambda_1,\dots,\lambda_n} f\left(A_1(\cdot),\dots,A_n(\cdot)\right)$$
$$= \sum_{m_1,\dots,m_n=0}^{\infty} a_{m_1,\dots,m_n} P_{\lambda_1,\dots,\lambda_n}^{m_1,\dots,m_n} \left(A_1(\cdot),\dots,A_n(\cdot)\right);$$
(2.16)

i.e., we define the action of $\mathcal{T}_{\lambda_1,\dots,\lambda_n}$ on f term-by-term. We often refer to the operator obtained from the action of the disentangling map as the *disentangled operator*.

The following theorem can be found in [15, Chapter 8]. (A thorough discussion of the disentangling map and its properties can be found in Chaper 8 of [15] and [17] when the time-ordering measures have non-zero discrete parts and in Chapter 2 of [15] when the time-ordering measures are continuous.

Theorem 2.6. The disentangling map $\mathcal{T}_{\lambda_1,...,\lambda_n}$ is a bounded linear operator from \mathbb{D} into $\mathcal{L}(X)$. Moreover, $\|\mathcal{T}_{\lambda_1,...,\lambda_n}\| \leq 1$; i.e., $\mathcal{T}_{\lambda_1,...,\lambda_n}$ is a linear contraction on \mathbb{D} .

It is also worth recording the form the disentangling map takes when the time-ordering measures are purely discrete and finitely supported. (See Corollary 8.4.3 of [15].)

Theorem 2.7. If the time-ordering measures $\lambda_1, \ldots, \lambda_n$ are purely discrete and finitely supported with the union of the supports being $\{\tau_1, \ldots, \tau_h\}$ with $0 < \tau_1 < \cdots < \tau_h < T$ and

$$\lambda_j = \sum_{i=1}^h p_{ji} \delta_{\tau_i},$$

then

$$\begin{aligned} \mathcal{T}_{\lambda_{1},...,\lambda_{n}}f(A_{1}(\cdot)^{\sim},\ldots,A_{n}(\cdot)^{\sim}) \\ &= \sum_{m_{1},...,m_{n}=0}^{\infty} a_{m_{1},...,m_{n}} \sum_{j_{11}+\cdots+j_{1h}=m_{1}} \cdots \\ &\sum_{j_{n1}+\cdots+j_{nh}=m_{n}} \left(\frac{m_{1}!\cdots m_{n}!}{j_{11}!\cdots j_{1h}!\cdots j_{n1}!\cdots j_{nh}!} \right) \\ &\left\{ \prod_{\beta=0}^{n-1} [p_{n-\beta,h}A_{n-\beta}(\tau_{h})]^{j_{n-\beta,h}} \right\} \cdots \left\{ \prod_{\beta=0}^{n-1} [p_{n-\beta,1}A_{n-\beta}(\tau_{1})]^{j_{n-\beta,1}} \right\} \end{aligned}$$

Finally, if all of our time-ordering measures are continuous, then we have the following. (See Chapter 2 of [15]. See also [8])

Theorem 2.8. If the time-ordering measures $\lambda_1, \ldots, \lambda_n$ are continuous, then

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n} f\left(A_1(\cdot)^{\sim},\dots,A_n(\cdot)^{\sim}\right)$$

$$= \sum_{m_1,\dots,m_n=0}^{\infty} a_{m_1,\dots,m_n} \sum_{\pi \in S_m} \int_{\Delta_m(\pi)} C_{\pi(m)}(s_{\pi(m)}) \cdots \qquad (2.17)$$

$$C_{\pi(1)}(s_{\pi(1)}) \left(\lambda_1^{m_1} \times \dots \times \lambda_n^{m_n}\right) (ds_1,\dots,ds_m).$$

Note that the disentangled monomial, in the case that our measures are continuous, is

$$\sum_{\pi \in S_m} \int_{\Delta_m(\pi)} C_{\pi(m)}(s_{\pi(m)}) \cdots C_{\pi(1)}(s_{\pi(1)}) (\lambda_1^{m_1} \times \cdots \times \lambda_n^{m_n}) (ds_1, \dots, ds_m).$$

Remark 2.9. While our time-ordering measures in the presentation above have been probability measures, it turns out that the disentangling takes the same form when the time-ordering measures are non-probability measures. See [8] or [15, Section 3.2]. Such measures are of use when considering

evolution problems, for example; see [30]. We do not consider such problems in this paper, and so the reader can safely assume that all time-ordering measures used below are probability measures on [0, T].

2.3. Some basic facts from operator theory. We present here some of the basic ideas of operator theory which will be needed in this paper. A detailed discussion of the ideas presented below can be found in many standard references, for example [19], [31], [32], [4]. Here, we closely follow the presentation of [19].

As the reader is no doubt aware, unlike with bounded linear operators from a Banach space X to a Banach space Y, with unbounded operators having domains not equal to all of X, various difficulties arise. As is wellknown, if S and T are unbounded linear operators from X to Y with domains $\mathcal{D}(S)$ and $\mathcal{D}(T)$, respectively, linear combinations $\alpha S + \beta T$ may not be defined and the operator product ST (or TS) may not be defined. Both of these pathologies arise in Feynman's operational calculus if one attempts to use unbounded operators in a naive way.

We will turn our attention to closed unbounded linear operators from a Banach space X to a Banach space Y. (Below, X and Y will always denote Banach spaces.)

Definition 2.10. Let T be a linear operator from X to Y with domain $\mathcal{D}(T)$. If $\{\phi_n\}_{n=1}^{\infty}$ is a sequence from $\mathcal{D}(T)$ such that $\phi_n \to \phi$ as $n \to \infty$ and such that $T\phi_n \to \psi \in Y$, then $\phi \in \mathcal{D}(T)$ and $T\phi = \psi$, then we say that T is *closed*. In particular, if we define the *graph* of T by

$$\mathcal{G}(T) := \left\{ (\phi, T\phi) : \phi \in \mathcal{D}(T) \right\},\$$

then T is closed if and only if $\mathcal{G}(T)$ is a closed subspace of $X \times Y$.

Remark 2.11. The family of all closed linear operators from X to Y will be denoted by $\mathscr{C}(X, Y)$.

If a linear operator from X to Y is not closed, it may still be *closable*.

Definition 2.12. A linear operator T from X to Y is *closable* if the closure $\overline{\mathcal{G}(T)}$ of $\mathcal{G}(T)$ is a graph; i.e., there is a $T_0 \in \mathscr{C}(X, Y)$ with $\mathcal{G}(T_0) = \overline{\mathcal{G}(T)}$. We call T_0 the closure of T and T_0 is the smallest closed extension of T.

Now, we take the time to remind the reader of the resolvent of an operator and of the resolvent set of an operator.

Definition 2.13. Let T be a closed operator on X. Then the same is true for $\xi - T$, for any $\xi \in \mathbb{C}$. (We follow the convention that, when writing $\xi - T$, we interpret ξ as ξI where I is the identity operator.) If $\xi - T$ is invertible with

$$R(\xi;T) := (\xi - T)^{-1} \in \mathcal{L}(X),$$

 ξ is said to belong to the *resolvent set* of T and we denote the resolvent set by $\rho(T)$. The operator-valued function $R(\cdot;T):\rho(T)\longrightarrow \mathcal{L}(X)$ is called the *resolvent* of T.

As is well-known, an operator T commutes with its resolvent function; i.e., $R(\xi;T)T \subset TR(\xi;T)$ for $\xi \in \rho(T)$. Just as well-known is the fact that $\rho(T)$ is an open subset of \mathbb{C} and $R(\xi;T)$ is a (piecewise) analytic function for $\xi \in \rho(T)$.

Definition 2.14. For a closed operator T on X with resolvent set $\rho(T)$, the spectrum $\sigma(T)$ of T is the complement of $\rho(T)$: $\sigma(T) = \mathbb{C} \setminus \rho(T)$. So, if $\zeta \in \sigma(T), \zeta - T$ is not invertible or is invertible but has range smaller than X.

When we address the use of the Taylor calculus for unbounded operators in Feynman's operational calculus, we will need to refer specifically to the point at infinity. The following theorem is Theorem III.6.13 of [19].

Theorem 2.15. Let $T \in \mathscr{C}(X)$ and let $\rho(T)$ contain the exterior of a circle. Then we have the alternatives:

- (i) $T \in \mathcal{L}(X)$; $R(\xi;T)$ is analytic at $\xi = \infty$ and $R(\infty;T) = 0$.
- (ii) $R(\xi;T)$ has an essential singularity at $\xi = \infty$.

In view of this theorem, we include $\xi = \infty$ in the resolvent set of T if $T \in \mathcal{L}(X)$ and in $\sigma(T)$ otherwise. Hence, an unbounded operator always has $\xi = \infty$ in its spectrum and, if it is an isolated point, it is an essential singularity of the resolvent function.

We now undertake a brief discussion of unbounded quadratic forms, which will be useful to us below. The presentation in [14, Section 10.3] will be followed here. Also, \mathcal{H} will be a complex Hilbert space throughout this discussion.

Definition 2.16. A sesquilinear form is a map $q : Q(q) \times Q(q) \longrightarrow \mathbb{C}$, where Q(q) is a dense subspace of \mathcal{H} , the form domain of q, such that qis linear in the first variable and conjugate linear in the second variable. If $q(\phi, \psi) = \overline{q(\psi, \phi)}$ for every $\phi, \psi \in Q(q)$, the form q is said to be symmetric. For notational symplicity, we will often write $q(\phi)$ for $q(\phi, \phi)$ when $\phi \in Q(q)$. If there is a number $c \geq 0$ such that

$$q(\phi) \ge -c \|\phi\|_{\mathcal{H}}^2, \tag{2.18}$$

for $\phi \in Q(q)$, we say that q is semibounded or bounded below. If we can take c = 0, then q is said to be nonnegative.

Let q be a semibounded quadratic form and let $c \ge 0$ be such that (2.18) holds for all $\phi \in Q(q)$. If we define

$$\langle \phi, \psi \rangle_{+1} := q(\phi, \psi) + (c+1) \langle \phi, \psi \rangle_{\mathcal{H}}, \tag{2.19}$$

then $\langle \cdot, \cdot \rangle_{+1}$ is an inner product on Q(q). The associated norm $\|\cdot\|_{+1}$ is

$$\|\phi\|_{+1} := \{q(\phi) + (c+1)\langle\phi,\phi\rangle_{\mathcal{H}}\}^{1/2}$$
(2.20)

for $\phi \in Q(q)$.

Definition 2.17. A semibounded quadratic form q as above is called *closed* if Q(q) is complete under the norm $\|\cdot\|_{+1}$ given by (2.20). If q is closed, and if D is a subspace of Q(q) which is dense in Q(q) in the norm $\|\cdot\|_{+1}$, then D is called a *form core* for q.

Now, let A be a (unbounded) self-adjoint operator on \mathcal{H} . There is always a quadratic form q_A associated with A (q_A will not in general be semibounded). The form domain $Q(q_A)$ and the form q_A can be described in terms of a spectral representation for A. This representation involves both the Hilbert space \mathcal{H} and a unitary operator U from \mathcal{H} onto the representing space. However, one normally suppresses reference to U and acts as though \mathcal{H} were the representing space.

The form q_A is called the quadratic form associated with A, and we sometimes write $Q(q_A) = Q(A)$. The subspace Q(A) is called the form domain of the operator A. It follows from this definition that $q_A(\phi, \psi) =$ $(\phi, A\psi)$ for all $\phi \in Q(A)$ and all $\psi \in \mathcal{D}(A)$.

We need the definition of a semibounded symmetric operator.

Definition 2.18. A symmetric operator $A : \mathcal{D}(A) \to \mathcal{H}$ is said to be *semibounded* if there is a $c \geq 0$ such that

$$(\phi, A\phi) \ge -c \|\phi\|_{\mathcal{H}}^2$$

for all $\phi \in \mathcal{D}(A)$.

We note that if A is semibounded, then the quadratic form associated to A is semibounded and closed. (Proposition 10.3.8 of [14].)

We now outline the idea of the form sum of two self-adjoint operators. The reader will recall that the ordinary operator sum of self-adjoint operators A and B is defined on $\mathcal{D}(A + B) := \mathcal{D}(A) \cap \mathcal{D}(B)$ and, for $\phi \in \mathcal{D}(A + B)$, $(A + B)\phi := A\phi + B\phi$. As the reader no doubt knows, however, it may happen that $\mathcal{D}(A) \cap \mathcal{D}(B)$ is not dense in \mathcal{H} and, in fact, it may be $\{0\}$. If q_A and q_B are the quadratic forms associated to A and B, respectively, the idea is to consider $q_A + q_B$. Under appropriate conditions, there will be a unique densely defined, semibounded, self-adjoint operator associated to $q_A + q_B$ and this operator, called the *form sum of* A and B and denoted by A + B.

In most applications, the self-adjoint operator A will be nonnegative and so we restrict ourselves to this case. We will write the self-adjoint operator B as $B = B_+ - B_-$, using the spectral theorem for self-adjoint operators. We also appeal to the spectral theorem for the operators $A^{1/2}$, $B^{1/2}$. Indeed, according to the spectral theorem in the multiplication operator form (see Theorem 10.1.8 of [14]), the self-adjoint operator A is unitarily equivalent to the multiplication operator M_f on $L^2(\mu)$; i.e., $A = U^{-1}M_f$ for a unitary operator $U : \mathcal{H} \to L^2(\mu)$. We can then write $A_{\pm} = U^{-1}M_{f+1}U$ with $\mathcal{D}(A_{\pm}) =$

$$\left\{\phi\in L^2(\mu): f_\pm\cdot\phi\in L^2(\mu)\right\}.$$
 Also,

$$A_\pm^{1/2}=U^{-1}M_{f_\pm^{1/2}}U$$

with domain defined similarly. Note that, if A is nonnegative, $A_{+} = A$.

Definition 2.19. We say that B_{-} is relatively form bounded with respect to A (" B_{-} is A-form bounded") with bound less than 1 if $Q(A) \subseteq Q(B_{-})$ and there are positive constants $\gamma < 1$ and δ such that

$$\|B_{-}^{1/2}\phi\|_{\mathcal{H}}^{2} \leq \gamma \|A^{1/2}\phi\|_{\mathcal{H}}^{2} + \delta \|\phi\|_{\mathcal{H}}^{2}.$$

The infimum of all such positive numbers γ is called the *A*-form bound of *B*.

The form sum is determined by the following theorem [14, Theorem 10.3.19].

Theorem 2.20. Let A and B be self-adjoint operators with A nonnegative. Suppose also that B_- is A-form bounded with bound less than 1 and assume that $Q(A) \cap Q(B_+)$ is dense in \mathcal{H} . Then the quadratic form $q_A + q_B$ given by

$$q_A(\phi,\psi) + q_B(\phi,\psi) = (A^{1/2}\phi, A^{1/2}\psi) + (B_+^{1/2}\phi, B_+^{1/2}\psi) - (B_-^{1/2}\phi, B_-^{1/2}\psi)$$

for $\phi, \psi \in Q(A) \cap Q(B_+)$ (this form is bounded below) is closed and semibounded and so there is a unique semi-bounded self-adjoint operator A + B, called the form sum of A and B, such that $q_{A+B} = q_A + q_B$ with $Q(A+B) = Q(A) \cap Q(B) = Q(A) \cap Q(B_+)$. The operator A+B is a self-adjoint extension of the algebraic sum A + B.

2.4. Functions of unbounded operators - The calculus of A. E. Taylor. This section serves to outline the essential details of the functional calculus of A. E. Taylor found in the 1951 paper [34]. This calculus serves as the basis for Section 4 and we follow [34] closely in this subsection.

Let X be a complex Banach space. Throughout this section, T will be a closed operator on X with domain $\mathcal{D}(T)$. We will also let $\rho(T)$ be the resolvent set of T (see Subsection 2.3) and we will denote, for $\lambda \in \rho(T)$, the resolvent operator $(\lambda - T)^{-1}$ by $R(\lambda; T)$. The spectrum $\sigma(T)$ consists of all complex numbers λ not in $\rho(T)$.

We know that $\rho(T)$ is open in \mathbb{C} and so $\sigma(T)$ is closed. When $T \in \mathcal{L}(X)$, $\sigma(T)$ is bounded and nonempty. Just as well-known is the fact that, in the general case, $\sigma(T)$ may be empty, unbounded or equal to all of \mathbb{C} . Also, when $\rho(T) \neq \emptyset$, the resolvent operator $R(\lambda; T)$ is an analytic $\mathcal{L}(X)$ -valued function on $\rho(T)$.

We now need a definition of the certain subsets of the complex plane.

Definition 2.21. A set $D \subseteq \mathbb{C}$ is called a *Cauchy Domain* if the following conditions are satisfied:

- (i) D is open in \mathbb{C} ;
- (ii) D has a finite number of components, the closures of any two of which are disjoint;
- (iii) the boundary of D is composed of a finite number of closed rectifiable Jordan curves, no two of which intersect.

Remark 2.22. We note that a component of a Cauchy domain is also a Cauchy domain. Furthermore, if the Cauchy domain D is unbounded, it has just one unbounded component. This unbounded component contains a neighborhood of the point at infinity (meaning all points outside a sufficiently large disk), and has as its boundary a finite number of closed rectifiable Jordan curves which do not intersect and no one of these curves is inside any other.

If D is a Cauchy domain and if C is one of the curves making up its boundary, we will, as is customary, define the positive orientation of C as part of the boundary ∂D of D.

The following theorem is fundamental. (It is Theorem 3.3 of [34].)

Theorem 2.23. Let F and Δ be subsets of \mathbb{C} . Suppose that F is closed and Δ is open and assume that $F \subseteq \Delta$. Suppose further that $\partial \Delta$ is nonempty and bounded. Then there is a Cauchy domain D such that:

- (i) $F \subset D$,
- (ii) $\overline{D} \subset \Delta$,
- (iii) the curves forming ∂D are polygons and
- (iv) D is unbounded if Δ is unbounded.

We now move onto the functional calculus of Taylor. To this end, we take f(z) to be a complex-valued function of the complex variable z, we denote by $\Delta(f)$ the domain of f and we will assume that $\Delta(f) \neq \emptyset$ (but it may not necessarily be connected) and finally that f(z) is single-valued and analytic on $\Delta(f)$.

Definition 2.24. We let $\mathscr{G}(T)$ be the family of all analytic functions f for which

- (i) $\sigma(T) \subseteq \Delta(f)$,
- (ii) $\Delta(f)$ contains a neighborhood of $z = \infty$ in \mathbb{C} and f is regular at $z = \infty$. (By "regular," we mean that f is analytic in a neighborhood of $z = \infty$; i.e., analytic on the complement of some open disk.)

When $f \in \mathscr{G}(T)$, we denote the limiting value of f(z) as $z \to \infty$ by $f(\infty)$.

Remark 2.25. If $\sigma(T) = \mathbb{C}$, the Taylor calculus cannot be defined. Hence, we will assume from here on that $\rho(T)$ is never empty for any operator under consideration.

The primary theorem which allows the Taylor calculus to be used is Theorem 4.1 of [34], which we will state here for the reader's convenience.

Theorem 2.26. Suppose that $f \in \mathscr{G}(T)$. There is an unbounded Cauchy domain D such that $\sigma(T) \subset D$ and $\overline{D} \subset \Delta(f)$. The integral

$$\frac{1}{2\pi i} \int_{\partial D} f(z) R(z;T) \, dz \tag{2.21}$$

defines an element of $\mathcal{L}(X)$ which is the same for any choice of Cauchy domain D satisfying the conditions stated above.

We now define f(T) using Theorem 2.26.

Definition 2.27. When $f \in \mathscr{G}(T)$ we define

$$f(T) := f(\infty)I + \frac{1}{2\pi i} \int_{\partial D} f(z)R(z;T) \, dz.$$
 (2.22)

2.5. Generalized convergence of closed operators. In this subsection, we follow Kato's monograph [19]. We start with the idea of the "gap" between subspaces. Let X and Y be Banach spaces and let $\mathscr{C}(X,Y)$ be the space of all closed densely defined linear operators from X to Y. If $T, S \in \mathscr{C}(X, Y)$, then their graphs $\mathcal{G}(T)$ and $\mathcal{G}(S)$ are closed subspaces of $X \times Y$. In [19], the "distance" between T and S is measured by the "gap" between the closed subspaces $\mathcal{G}(T)$ and $\mathcal{G}(S)$. In order to measure this gap, we proceed as follows. (We follow Kato's notation below.)

Let Z be a Banach space and let M, N be closed subspaces of Z. Define

$$\delta(M,N) := \sup_{u \in S_M} \operatorname{dist}(u,N), \qquad (2.23)$$

where S_M denotes the unit sphere of M and where dist(u, N) denotes the distance from u to N; i.e., dist $(u, N) = \inf \{ ||u - a||_Z : a \in N \}$. We next define

$$\delta(M, N) := \max\left\{\delta(M, N), \delta(N, M)\right\}, \qquad (2.24)$$

and we define $\delta(0, N) = 0$. Also, $\delta(M, 0) = 1$ if $M \neq \emptyset$. We also note that $\delta(M, N)$ is the smallest nonnegative real number δ for which

$$\operatorname{dist}\left(u,N\right) \le \delta \|u\|,\tag{2.25}$$

for all $u \in M$. (Indeed, (2.25) will play a crucial role in Subsections 2.8 and 4.1.) The number $\hat{\delta}(M, N)$ is called the *gap* between M and N. Kato makes note of the following properties:

$$\begin{split} \delta\left(M,N\right) &= 0 \text{ if and only if } M \subseteq N,\\ \hat{\delta}\left(M,N\right) &= 0 \text{ if and only if } M = N,\\ \hat{\delta}\left(M,N\right) &= \hat{\delta}\left(N,M\right),\\ 0 &\leq \delta\left(M,N\right) \leq 1 \text{ and } 0 \leq \hat{\delta}\left(M,N\right) \leq 1. \end{split}$$

We now use these ideas to discuss the gap between closed operators (see Section IV.4 of [19]). Given $T, S \in \mathscr{C}(X, Y)$, the number $\hat{\delta}(\mathcal{G}(T), \mathcal{G}(S)) =:$ $\hat{\delta}(T, S)$ is called the *gap* between T and S. Next, suppose that $\{T_n\}_{n=1}^{\infty}$ is a sequence from $\mathscr{C}(X, Y)$ and suppose that $T \in \mathscr{C}(X, Y)$. **Definition 2.28.** (See Page 202 of [19]) We say that $T_n \to T$ in the generalized sense if and only if $\hat{\delta}(T_n, T) \to 0$ as $n \to \infty$.

Note that if $\hat{\delta}(T_n, T) \longrightarrow 0$, then $\delta(T_n, T) \longrightarrow 0$. Also, in the case that Y = X, there is a criterion for generalized convergence, Theorem IV.2.25 of [19].

Theorem 2.29. Let $T \in \mathscr{C}(X)$ have a non-empty resolvent set $\rho(T)$. In order that a sequence $\{T_n\}_{n=1}^{\infty}$ from $\mathscr{C}(X)$ to converge to T, it is necessary that $\rho(T) \subseteq \rho(T_n)$ for sufficiently large n and

$$||R(\zeta;T_n) - R(\zeta;T)||_{\mathcal{L}(X)} \longrightarrow 0$$

as $n \to \infty$, while it is sufficient that this be true for some $\zeta \in \rho(T)$.

A sufficient condition for generalized convergence is Theorem IV.2.29 of [19].

Theorem 2.30. Let $T_n, T \in \mathcal{C}(X, Y)$. Let Z be a third Banach space and let $U_n, U \in \mathcal{L}(Z, X)$ and $V_n, V \in \mathcal{L}(Z, Y)$ be such that U_n, U map Z onto $\mathcal{D}(T_n), \mathcal{D}(T)$ injectively, respectively and $T_nU_n = V_n$, TU = V. If $||U_n - U|| \to 0$ and $||V_n - V|| \to 0$ as $n \to \infty$, then $T_n \to T$ in the generalized sense.

2.6. Bounded analytic families of operators. Let X and Y be Banach spaces. We are interested in families T(z) (notation following [19], Chapter VII, Section 1) of operators from X to Y where T(z) is an analytic function in a domain D of \mathbb{C} . First, recall that a X-valued function u(z) on a domain D of \mathbb{C} is analytic if it is differentiable at each $z \in D$; it does not matter whether the derivative is a strong or weak derivative. (See [19] or Section VI.3 of [31].) Therefore, u(z) is analytic if and only if $(u(z), x^*)$ is analytic for every $x^* \in X^*$. (We use the standard notation (\cdot, \cdot) for the duality bracket.)

For operator-valued analytic functions T(z), we say that T(z) is analytic in a domain D of \mathbb{C} if and only if T(z) is norm differentiable for all $z \in D$. We can say that $T(z) \in \mathcal{L}(X, Y)$ is analytic if and only if each z has a neighborhood in which T(z) is bounded and $(T(z)x, y^*)$ is analytic for any x in a fundamental subset of X and every y^* in a fundamental subset of Y^* . Note that, if Y = X, then T(z) is analytic if and only if each z has a neighborhood in which T(z) is bounded and $(T(z)x, x^*)$ is analytic for all xin a fundamental subset of X and every x^* in a fundamental subset of X^* . These statements follow from Theorem III.3.12 of [19]:

Theorem 2.31. Let $T(z) \in \mathcal{L}(X, Y)$ be defined in a domain Δ of \mathbb{C} and suppose that $(T(z)x, y^*)$ is analytic in $z \in \Delta$ for each $x \in X$ and $y^* \in Y^*$. Then T(z) is analytic in Δ in the sense of norm differentiability.

2.7. Analytic families of unbounded operators. Subsection 2.6 states the relevant definitions and criteria for analytic families (functions) from \mathbb{C}

into $\mathcal{L}(X, Y)$. (The so-called *bounded analytic families* or *boundedly analytic families*.) However, norm differentiability will not work when looking at functions from \mathbb{C} into $\mathscr{C}(X, Y)$. Following Section VII.2 of [19], we extend the idea of analytic family to unbounded operators using the idea of generalized convergence of closed operators.

Definition 2.32. A family $T(z) \in \mathscr{C}(X, Y)$ defined in a neighborhood of z = 0 in \mathbb{C} is said to be holomorphic at z = 0 (in the generalized sense) if there is a third Banach space Z and two boundedly analytic functions $U(z) \in \mathcal{L}(Z, X)$ and $V(z) \in \mathcal{L}(Z, Y)$ (in the sense of Subsection 2.2) at z = 0 such that U(z) maps Z onto $\mathcal{D}(T(z))$ injectively and

$$T(z)U(z) = V(z).$$
 (2.26)

The function T(z) is analytic in a domain D of the complex plane if it is analytic at every $z \in D$. Finally, if T(z) is analytic, then it is continuous in the generalized sense that if $z \to z_0$, then $T(z) \to T(z_0)$ in the generalized sense; see Theorem 2.30 above.

When Y = X, we have Theorem VII.1.3 of [19].

Theorem 2.33. Let $T(z) \in \mathscr{C}(X)$ be defined in a neighborhood of z = 0and let $\zeta \in \rho(T(0))$. Then T(z) is analytic at z = 0 if and only if $\zeta \in \rho(T(z))$ and the resolvent $R(\zeta; T(z)) = (\zeta - T(z))^{-1}$ is boundedly analytic for sufficiently small |z|.

2.8. Analytic families of unbounded operators and generalized convergence. In this subsection we state a definition which will be used to assist our investigations of the operational calculus in the setting of analytic families. In particular, this definition and the discussion which follows will be of crucial use in the study of the stability of the operational caluclus with respect to sequences of analytic families.

Definition 2.34. Suppose that T(z) is an analytic family of unbounded operators on the domain $D \subseteq \mathbb{C}$ in the sense of Subsection 2.7. We will say that T(z) is uniform at $z_0 \in D$ if, for any $\epsilon > 0$, there is a $\iota > 0$ such that given any $(\phi, T(z_0)\phi) \in \mathcal{G}(T(z_0))$ and $(\psi, T(z)\psi) \in \mathcal{G}(T(z))$ with $|z-z_0| < \iota$, then

$$\|(\phi, T(z_0)\phi) - (\psi, T(z)\psi)\|_{X \times X} < \epsilon.$$
(2.27)

We will say that T(z) is uniform on $D \subseteq \mathbb{C}$ if is uniform at every point in D.

We apply this definition to a sequence $\{T_n(z)\}_{n=1}^{\infty}$ of analytic families of closed unbounded operators on X on the domain $D \subseteq \mathbb{C}$ each of which satisfy Definition 2.34 on D and which are such that $T_n(z) \longrightarrow T(z)$ in the generalized sense for each $z \in D$. We will also assume that the analytic family T(z) satisfies Definition 2.34 at each point $z \in D$.

Write, as in Subsection 2.7, T(z)U(z) = V(z) where U(z) is boundedly analytic on D and injective from a Banach space Z onto $\mathcal{D}(T(z)) \subseteq X$

and V(z) is boundedly analytic on D and maps Z into X. Similarly, write $T_n(z)U_n(z) = V_n(z)$ for each $n \in \mathbb{N}$, where $U_n(z)$ is boundedly analytic on D and injective from Z onto $\mathcal{D}(T_n(z)) \subseteq X$ and $V_n(z)$ is boundedly analytic on D from Z into X. Note that we may write

$$\mathcal{G}(T(z)) = \{ (\phi, T(z)\phi) : \phi \in \mathcal{D}(T(z)) \}$$
$$= \{ (U(z)\psi, V(z)\psi) : \psi \in Z \}$$
(2.28)

and, similarly,

$$\mathcal{G}(T_n(z)) = \left\{ (U_n(z)\psi, V_n(z)\psi) : \psi \in Z \right\}.$$
(2.29)

Fix a $z \in D$ and let $w(z) \in \mathcal{G}(T(z))$. For each positive integer n, there is a $w_n(z) \in \mathcal{G}(T_n(z))$ for which

$$\|w(z) - w_n(z)\|_{X \times X} < \operatorname{dist} (w(z), \mathcal{G}(T_n(z))) + \frac{1}{n}$$
$$\leq \delta_n(z) \|w(z)\|_{X \times X} + \frac{1}{n} \longrightarrow 0 \qquad (2.30)$$

as $n \to \infty$ (see (2.25) in Subsection 2.5). Since $w(z) \in \mathcal{G}(T(z))$,

$$w(z) = (U(z)\phi(z), V(z)\phi(z))$$
 (2.31)

for some $\phi(z) \in Z$ and since $w_n(z) \in \mathcal{G}(T_n(z))$,

$$w_n(z) = (U_n(z)\psi_n(z), V_n(z)\psi_n(z))$$
(2.32)

for some $\psi_n(z) \in Z$. Because $||w(z) - w_n(z)||_{X \times X} \to 0$ as $n \to \infty$, the definition of the norm on $X \times X$ ($||(a,b)||_{X \times X} := \{||a||_X^2 + ||b||_X^2\}^{1/2}$), tells us that

$$\|U(z)\phi(z) - U_n(z)\psi_n(z)\|_X \to 0$$
(2.33)

and

$$|V(z)\phi(z) - V_n(z)\psi_n(z)||_X \to 0$$
 (2.34)

as $n \to \infty$.

We now ask what happens when we move from $z \in D$ to $z_1 \in D$. Given $\epsilon > 0$, choose $\iota > 0$ according to Definition 2.34. If $z_1 \in D$ satisfies $|z_1 - z| < \iota$, then

$$\|w(z) - w(z_1)\|_{X \times X}$$

= $\|(U(z)\phi(z), V(z)\phi(z)) - (U(z_1)\phi(z_1), V(z_1)\phi(z_1))\|_{X \times X} < \epsilon.$ (2.35)

In particular, then, it follows from the definition of the norm on $X \times X$ that

$$\|U(z)\phi(z) - U(z_1)\phi(z_1)\|_{X \times X} < \epsilon$$
(2.36)

and

$$\|V(z)\phi(z) - V(z_1)\phi(z_1)\|_{X \times X} < \epsilon.$$
(2.37)

If we define $W: D \longrightarrow X$ by

$$W(z) := V(z)\phi(z), \qquad (2.38)$$

we see that W is continuous at every $z \in D$. Proceeding in the same way, we see that, for each $n \in \mathbb{N}$, the map $W_n : D \longrightarrow X$ defined by

$$W_n(z) := V_n(z)\psi_n(z) \tag{2.39}$$

is continuous at every $z \in D$. Moreover, from (2.34), we see that

$$W_n(z) \longrightarrow W(z)$$
 (2.40)

in X-norm as $n \to \infty$ for each $z \in D$.

We will use the continuity and limits discussed above in Subsection 4.1 below, just after the definitions (5.18) and (5.19).

3. Feynman's operational calculus on Banach algebras and some stability theory

In this section we first outline how to carry out Feynman's operational calculus in a Banach algebra setting. Indeed, usual setting for the operational calculus is $\mathcal{L}(X)$, a Banach algebra. We will use this formalism in Section 4 of this paper when we study stability theory when using analytic families of closed unbounded linear operators. We will also use the stability result, Theorem 3.16, when considering joint stability of the operational calculus in the setting of analytic families of operators.

3.1. Comments on measurability. Before proceeding further, we will make some comments concerning measurability of functions taking values in a Banach space. We will follow the notation/terminology found in [7]. In particular, let (S, \mathscr{A}) be a measurable space and let X be a Banach space. The first definition of *measurable* which comes to mind is that, given a function $f: S \to X$, f is measurable provided that $f^{-1}(B) \in \mathscr{A}$ for every Borel set $B \subset X$. As the reader is no doubt aware, this definition is, in general, not too useful as the Borel class $\mathscr{B}(X)$ is 'too large'. Indeed, the σ -algebra generated by all continuous linear functions on X may be strictly smaller than $\mathscr{B}(X)$ and this is an obstruction to the application of the usual tools of functional analysis. Nevertheless, we state the following definition.

Definition 3.1. Let (S, \mathscr{A}) be a measurable space and let X be a Banach space. A function $f: S \to X$ is said to be measurable if $f^{-1}(B) \in \mathscr{A}$ for every Borel set $B \subset X$.

It is the case, however, that if X is a separable Banach space, the problems presented by $\mathscr{B}(X)$ being too large disappear. Indeed, if X is separable and Y is a weak* dense linear subspace of X^* (i.e., Y separates the points of X) then, according to Proposition 1.1.1 of [7],

$$\sigma(Y) = \sigma(X^*) = \mathscr{B}(X),$$

where $\sigma(Y)$ is, by definition, the smallest σ -algebra for which every $y^* \in Y^*$ is measurable. (It can be shown that $\sigma(Y)$ is generated by all sets of the form $\{x \in X : (\langle x, x_1^* \rangle, \dots, \langle x, x_n^* \rangle) \in B\}$ with $n \ge 1, x_1^*, \dots, x_n^* \in Y$ and $B \in \mathscr{B}(\mathbb{C}^n)$.) The following corollary (Corollary 1.1.2 of [7]) is useful. **Corollary 3.2.** If X is separable, then for a function $f : S \to X$ the following are equivalent.

- (i) f is measurable;
- (ii) $\langle f, x^* \rangle$ is measurable (as a scalar function) for all $x^* \in X^*$.

Next, we state the following definition, which is Definition 1.1.4 of [7].

Definition 3.3. A function $f: S \to X$ is strongly measurable if there is a sequence of simple functions $f_n: S \to X$ such that $f_n \to f$ pointwise on S. When we wish to stress the underlying σ -algebra, we will write "strongly \mathscr{A} -measurable."

The result we are interested in is Corollary 1.1.10 of [7] to the Pettis measurability theorem.

Corollary 3.4. For a function $f : S \to X$, the following assertions are equivalent:

(i) f is strongly measurable;

(ii) f is separably valued and measurable.

Consequently, if X is separable, then an X-valued function f is strongly measurable if and only if it is measurable.

We now move to considering X-valued functions on the measure space (S, \mathscr{A}, μ) . The definition of strong μ -measurability is Definition 1.1.14 of [7].

Definition 3.5. A function $f : S \to X$ is strongly μ -measurable if there is a sequence $f_n : S \to X$ of μ -simple functions converging to f μ -almost everywhere, where a μ -simple function is a simple function defined using elements of \mathscr{A} (i.e., of the form $\sum_{j=1}^{N} \alpha_j \chi_{A_j}$, for $A_j \in \mathscr{A}$ and $\mu(A_j) < \infty$).

A proposition relating the ideas of 'strong measurability' and ' μ - measurability' is Proposition 1.1.16 of [7].

Proposition 3.6. Consider a function $f: S \to X$.

(i) If f is strongly μ -measurable, then f is μ -almost everywhere equal to a strongly measurable function.

(ii) If μ is σ -finite and if f is μ -almost everywhere equal to a strongly measurable function, then f is strongly μ -measurable.

Finally, we state the Pettis measurability theorem, Theorem 1.1.20 of [7].

Theorem 3.7. For a function $f: S \to X$, the following are equivalent:

(i) f is strongly μ -measurable;

(ii) f is μ -essentially separably valued and weakly μ -measurable;

(iii) f is μ -essentially separably valued and there exists a weak-* dense subspace Y of X^{*} such that $\langle f, x^* \rangle$ is μ -measurable for all $x^* \in Y$.

Since Feynman's operational calculus involves operator-valued functions, it will be useful to note the relevant definition of strong measurability (respectively, of strong μ -measurability).

Definition 3.8. A function $f: S \to \mathcal{L}(X, Y)$ (X and Y Banach spaces) is called strongly measurable (respectively, strongly μ -measurable) if for all $x \in X$, the Y-valued function $f_x: S \to Y$ defined by $f_x(s) = f(s)x$ is strongly measurable (respectively strongly μ -measurable). The reader will notice that it is more accurate to refer to these functions as being strongly (μ -) measurable with respect to the strong operator topology, however, we will usually omit reference to the strong operator topology.

3.2. Feynman's operational calculus on a Banach algebra. Let X be a Banach algebra and assume, for convenience, that X is separable. As seen in the previous subsection, the assumption that X is separable simplifies discussions of measurability. (See Theorem 3.7 and Corollary 3.4 above.) For each $j = 1, \ldots, n$, let $W_j : [0,T] \to X$ and associate to $W_j(\cdot)$ a Borel probability measure λ_j on [0,T]. We assume that $W_j(\cdot)$, $j = 1, \ldots, n$, is strongly λ_j -measurable. As is well-known, $W_j(\cdot)$ is strongly λ_j -measurable if and only if $W_j(\cdot)$ is measurable and λ_j -almost separably valued. (See Corollary 3.4 and, for a detailed discussion, [4, p. 147] or [7].) It is here that our separability assumption on X plays its role. Since X is assumed to be separable, we can therefore say that $W_j(\cdot)$ is strongly λ_j -measurable if and only if $W_j(\cdot)$ is measurable.

To use Feynman's operational calculus in this setting, we have very little to do – all necessary work has already been done earlier. However, we sketch out some details. To start, define, for each $j = 1, \ldots, n$,

$$R_j := \int_{[0,T]} \|W_j(s)\|_X \,\lambda_j(ds)$$

and assume that this number is finite. Construct the commutative Banach algebras $\mathbb{A}(R_1, \ldots, R_n)$ and $\mathbb{D}((W_1(\cdot), \lambda_1)^{\sim}, \ldots, (W_j(\cdot), \lambda_j)^{\sim})$ just as before. (A moments' reflection should convince the reader that the formal objects $(W_j(\cdot), \lambda_j)^{\sim}, j = 1, \ldots, n$, are defined in essentially the same way as the formal objects for $\mathcal{L}(X)$ -valued functions.) Using the same notation as above, we can write for $f \in \mathbb{A}(R_1, \ldots, R_n) \cong \mathbb{D}(W_1(\cdot)^{\sim}, \ldots, W_n(\cdot)^{\sim})$, written as a Taylor series centered at $0 \in \mathbb{C}^n$,

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n} f\left(W_1(\cdot)^{\sim},\dots,W_n(\cdot)^{\sim}\right) \tag{3.1}$$

$$= f_{\lambda_1,\dots,\lambda_n} \left(W_1(\cdot),\dots,W_n(\cdot) \right)$$
(3.2)

$$=\sum_{m_1,\dots,m_n=0}^{\infty} a_{m_1,\dots,m_n} P^{m_1,\dots,m_n}_{\lambda_1,\dots,\lambda_n} \left(W_1(\cdot),\dots,W_n(\cdot) \right)$$
(3.3)

where $P_{\lambda_1,\ldots,\lambda_n}^{m_1,\ldots,m_n}(W_1(\cdot),\ldots,W_n(\cdot))$ takes exactly the same form as in (2.13), earlier, with $W_1(\cdot),\ldots,W_n(\cdot)$ replacing $A_1(\cdot),\ldots,A_n(\cdot)$, all other notation being exactly the same. The properties of the disentangling map $\mathcal{T}_{\lambda_1,\ldots,\lambda_n}$ also remain the same.

3.3. Two analytic lemmas. We now establish two lemmas which will be crucial to our study of the stability theory of the operational calculus using

analytic families of operators in Section 4, below. However, before stating and proving our lemmas, we take a moment to remind the reader of the definition of weak convergence of sequences of probability measures.

Definition 3.9. Let S be a metric space and suppose that $\{\lambda_k\}_{k=1}^{\infty}$ be a sequence of Borel probability measures on S. We say that $\{\lambda_k\}_{k=1}^{\infty}$ is weakly convergent to the Borel probability measure λ (and write $\lambda_k \rightarrow \lambda$) if

$$\lim_{k \to \infty} \int_S f \, d\lambda_k = \int_S f \, d\lambda$$

for every bounded continuous real-valued function f on S.

The first lemma addresses the first ingredient for joint stability.

Lemma 3.10. Let S be a separable metric space and let $f : S \to X$, X a separable Banach algebra, be strongly λ -measurable where λ is a Borel probability measure on S. Suppose that $\{\lambda_k\}_{k=1}^{\infty}$ is a sequence of Borel probability measures on S for which $\lambda_k \to \lambda$ as $k \to \infty$ and assume that, for every Borel set $E \subseteq S$ and every $k \in \mathbb{N}$, $\lambda_k(E) \leq \lambda(E)$. Let $\{f_k\}_{k=1}^{\infty}$ be a sequence of strongly λ_k -measurable functions for each $k \in \mathbb{N}$ and for which $f_k(\cdot) \to f(\cdot) \lambda$ -almost everywhere on S. Furthermore, we assume that

$$C_1 := \sup \{ \|f(s)\|_X : s \in S \} < \infty$$
(3.4)

and that

$$C_2 := \sup \{ \|f_k(s)\|_X : s \in S, k \in \mathbb{N} \} < \infty.$$
(3.5)

It then follows that

$$\int_{S} f_k(s) \lambda_k(ds) \longrightarrow \int_{S} f(s) \lambda(ds)$$
(3.6)

as $k \to \infty$ in norm on X.

Proof. We may write, for any $k, \ell \in \mathbb{N}$,

$$\begin{split} \left\| \int_{S} f_{k}(s) \lambda_{k}(ds) - \int_{S} f(s) \lambda(ds) \right\|_{X} \\ &\leq \left\| \int_{S} f_{k}(s) \lambda_{k}(ds) - \int_{S} f_{\ell}(s) \lambda_{k}(ds) \right\|_{X} + \left\| \int_{S} f_{\ell}(s) \lambda_{k}(ds) - \int_{S} f_{\ell}(s) \lambda(ds) \right\|_{X} \\ &+ \left\| \int_{S} f_{\ell}(s) \lambda(ds) - \int_{S} f(s) \lambda(ds) \right\|_{X} \\ &=: I + II + III. \end{split}$$

(3.7)

We treat these terms individually. For term I, we first note that, given $\epsilon > 0$, Egorov's theorem (see, for instance, [6]), supplies a Borel set $A \subseteq S$ such that $f_k(\cdot) \to f(x)$ uniformly on $S \setminus A$ and

$$\lambda(A) < \frac{\epsilon}{12\max(C_1, C_2)}.$$
(3.8)

We may now write

$$I = \left\| \int_{S \setminus A} f_k(s) \lambda_k(ds) - \int_{S \setminus A} f_\ell(s) \lambda_k(ds) + \int_A f_k(s) \lambda_k(ds) - \int_A f_\ell(s) \lambda_k(ds) \right\|_X$$

$$\leq \int_{S \setminus A} \|f_k(s) - f_\ell(s)\|_X \lambda_k(ds) + \int_A \|f_k(s)\|_X \lambda_k(ds) + \int_A \|f_\ell(s)\|_X \lambda_k(ds).$$

(3.9)

Because $f_k(\cdot) \to f(\cdot)$ uniformly on $S \setminus A$ as $k \to \infty$, $\{f_k\}_{k=1}^{\infty}$ is uniformly Cauchy. There is, then, a $N_1 \in \mathbb{N}$, for which

$$\sup\left\{\|f_k(s) - f_\ell(s)\|_X : s \in S \setminus A\right\} < \frac{\epsilon}{6}$$
(3.10)

whenever $k, \ell \geq N_1$. Therefore, for $k, \ell \geq N_1$,

$$\int_{S \setminus A} \|f_k(s) - f_\ell(s)\|_X \lambda_k(ds)
\leq \sup \{\|f_k(s) - f_\ell(s)\|_X : s \in S \setminus A\} \lambda_k(S \setminus A)
\leq \frac{\epsilon}{6} \cdot \lambda_k(S \setminus A)
\leq \frac{\epsilon}{6}.$$
(3.11)

Also, using (3.8), we note that

$$\int_{A} \|f_{k}(s)\|_{X} \lambda_{k}(ds) + \int_{[0,T]} \|f_{\ell}(s)\|_{X} \lambda_{k}(ds)$$

$$< C_{1} \left(\frac{\epsilon}{12 \max(C_{1}, C_{2})}\right) + C_{1} \left(\frac{\epsilon}{12 \max(C_{1}, C_{2})}\right)$$

$$\leq C_{1} \left(\frac{\epsilon}{12C_{1}}\right) + C_{1} \left(\frac{\epsilon}{12C_{1}}\right)$$

$$(3.12)$$

$$=\frac{\epsilon}{6}.$$
(3.13)

Therefore, for $k, \ell \geq N_1$, we have

$$I < \frac{\epsilon}{6} + \frac{\epsilon}{6} = \frac{\epsilon}{3}.\tag{3.14}$$

Turning to term II, for any $\ell \in \mathbb{N}$, we can find $N_2 \in \mathbb{N}$ such that if $k \geq N_2$,

$$\left\| \int_{S} f_{\ell}(s) \lambda_{k}(ds) - \int_{S} f_{\ell}(s) \lambda(ds) \right\|_{X} < \frac{\epsilon}{3}.$$
 (3.15)

Note that the integer N_2 depends on epsilon and on ℓ . However, just below, we will be fixing a value of ℓ .

Finally, for term III, we follow our analysis of term I with a change to accommodate the last term. Indeed, we may use (3.8) to write

$$III = \left\| \int_{S\setminus A} f_{\ell}(s) \lambda(ds) - \int_{S\setminus A} f(s) \lambda(ds) + \int_{A} f_{\ell}(s) \lambda(ds) - \int_{A} f(s) \lambda(ds) \right\|_{X}$$

$$\leq \int_{S\setminus A} \|f_{\ell}(s) - f(s)\|_{X} \lambda(ds) + \int_{A} \|f_{\ell}(s)\|_{X} \lambda(ds) + \int_{A} \|f(s)\|_{X} \lambda(ds)$$

$$< \int_{S\setminus A} \|f_{\ell}(s) - f(s)\|_{X} \lambda(ds) + C_{1} \left(\frac{\epsilon}{12 \max(C_{1}, C_{2})}\right)$$

$$+ C_{2} \left(\frac{\epsilon}{12 \max(C_{1}, C_{2})}\right)$$

$$\leq \int_{S\setminus A} \|f_{\ell}(s) - f(s)\|_{X} \lambda(ds) + \frac{\epsilon}{12} + \frac{\epsilon}{12}$$

$$= \int_{S\setminus A} \|f_{\ell}(s) - f(s)\|_{X} \lambda(ds) + \frac{\epsilon}{6}.$$
(3.16)

Choose $N_3 \in \mathbb{N}$ such that if $\ell \geq N_3$,

$$\sup\left\{\|f_{\ell}(s) - f(s)\|_{X} : s \in S \setminus A\right\} < \frac{\epsilon}{6}.$$
(3.17)

Hence, for $\ell \geq N_3$,

$$III \le \frac{\epsilon}{6} + \frac{\epsilon}{6} = \frac{\epsilon}{3}.$$

We now put things together. We start by fixing $\ell = \max(N_1, N_3)$ in the sum I + II + III and let $k \ge \max(N_1, N_2)$. It then follows that

$$I + II + III \le 3\left(\frac{\epsilon}{3}\right) = \epsilon$$

and the proof is complete.

Our next lemma is the second ingredient which will allow us to establish joint stability for the operational calculus in the setting of analytic families of operators.

Lemma 3.11. Let X be a separable Banach algebra. Further, let $A : [0,T] \longrightarrow X$ and let, for $k \in \mathbb{N}$, $A_k : [0,T] \longrightarrow X$. Assume that A is strongly λ -measurable where λ is a Borel probability measure on [0,T] and assume that A_k is strongly λ_k -measurable where λ_k is a Borel probability measure on [0,T] for each $k \in \mathbb{N}$. Assume that $\lambda_k \rightarrow \lambda$ as $k \rightarrow \infty$ and that $A_k(\cdot) \rightarrow A(\cdot)$ pointwise in X-norm on [0,T]. Finally, assume that

$$C_1 := \sup \{ \|A(s)\|_X : s \in [0, T] \} < \infty$$
(3.18)

and that

$$C_2 := \sup \{ \|A_k(s)\|_X : s \in [0, T], k \in \mathbb{N} \} < \infty.$$
(3.19)

For any $m \in \mathbb{N}$, we have

$$\int_{\Delta_m(T)} A_k(s_m) \cdots A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m) \rightarrow \int_{\Delta_m(T)} A(s_m) \cdots A(s_1) \lambda^m(ds_1, \dots, ds_m) \quad (3.20)$$

in X-norm as $k \to \infty$.

Proof. Given $m \in \mathbb{N}$, we may write, using the definition of the set $\Delta_m(T)$,

$$\int_{\Delta_m(T)} A_k(s_m) \cdots A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m)
= \int_{\Delta_m(T)}^T A(s_m) \cdots A(s_1) \lambda^m(ds_1, \dots, ds_m)
= \int_0^T \int_0^{s_m} \cdots \int_0^{s_3} \int_0^{s_2} A_k(s_m) A_k(s_{m-1}) \cdots A_k(s_2) A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m)
- \int_0^T \int_0^{s_m} \cdots \int_0^{s_3} \int_0^{s_2} A(s_m) A(s_{m-1}) \cdots A(s_2) A(s_1) \lambda^m(ds_1, \dots, ds_m).$$
(3.21)

We use (3.21) to write, successively,

$$\begin{split} &\int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) A_{k}(s_{1}) \lambda_{k}^{m}(ds_{1}, \dots, ds_{m}) \\ &\quad - \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A(s_{m}) A(s_{m-1}) \cdots A(s_{2}) A(s_{1}) \lambda^{m}(ds_{1}, \dots, ds_{m}) \\ &= \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) A_{k}(s_{1}) \lambda_{k}^{m}(ds_{1}, \dots, ds_{m}) \\ &\quad - \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) A(s_{1}) \\ \lambda(ds_{1}) \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m}) \\ &\quad + \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) A(s_{1}) \\ \lambda(ds_{1}) \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m}) \\ &\quad - \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) A(s_{1}) \lambda^{m}(ds_{1}, \dots, ds_{m}) \\ &= \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) \left\{ \int_{0}^{s_{2}} A_{k}(s_{1}) \lambda_{k}(ds_{1}) \\ &\quad - \int_{0}^{s_{2}} A(s_{1}) \lambda(ds_{1}) \right\} \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m}) \end{split}$$

$$+ \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{3}) \left\{ A_{k}(s_{2}) \int_{0}^{s_{2}} A(s_{1}) \lambda(ds_{1}) \right\} \cdot \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m}) \\ - \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A(s_{m}) A(s_{m-1}) \cdots A(s_{3}) \left\{ A(s_{2}) \int_{0}^{s_{2}} A(s_{1}) \lambda(ds_{1}) \right\} \cdot \lambda^{m-1}(ds_{2}, \dots, ds_{m}).$$
(3.22)

Define

$$A_k^{(1)}(s_2) := A_k(s) \int_0^{s_2} A(s_1) \,\lambda(ds_1) \tag{3.23}$$

and

$$A^{(1)}(s_2) := A(s_2) \int_0^{s_2} A(s_1) \,\lambda(ds_1). \tag{3.24}$$

It is clear that $A_k^{(1)}(s_2) \to A^{(1)}(s_2)$ in X-norm pointwise on [0,T]. The expression after the final equality in (3.22) can then be written as

$$\int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) \left\{ \int_{0}^{s_{2}} A_{k}(s_{1}) \lambda_{k}(ds_{1}) - \int_{0}^{s_{2}} A(s_{1}) \lambda(ds_{1}) \right\} \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m}) + \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{3}) A_{k}^{(1)}(s_{2}) \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m}) - \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} A(s_{m}) A(s_{m-1}) \cdots A(s_{3}) A^{(1)}(s_{2}) \lambda^{m-1}(ds_{2}, \dots, ds_{m})$$
(3.25)

We continue just as in (3.22), defining

$$A_k^{(j)}(s_{j+1}) := A_k(s_{j+1}) \int_0^{s_{j+1}} A(s_j) \,\lambda(ds_j) \tag{3.26}$$

and

$$A^{(j)}(s_{j+1}) := A(s_{j+1}) \int_0^{s_{j+1}} A(s_j) \,\lambda(ds_j) \tag{3.27}$$

for j = 2, 3, ..., m - 1. We can then the difference of iterated integrals in (3.21) as

$$\int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A_{k}(s_{m}) A_{k}(s_{m-1}) \cdots A_{k}(s_{2}) A_{k}(s_{1}) \lambda_{k}^{m}(ds_{1}, \dots, ds_{m})
- \int_{0}^{T} \int_{0}^{s_{m}} \cdots \int_{0}^{s_{3}} \int_{0}^{s_{2}} A(s_{m}) A(s_{m-1}) \cdots A(s_{2}) A(s_{1}) \lambda^{m}(ds_{1}, \dots, ds_{m})
= \int_{0}^{T} \cdots \int_{0}^{s_{3}} A_{k}(s_{m}) \cdots A_{k}(s_{2}) \left\{ \int_{0}^{s_{2}} A_{k}(s_{1}) \lambda_{k}(ds_{1}) - \int_{0}^{s_{2}} A(s_{1}) \lambda(ds_{1}) \right\} \cdot \lambda_{k}^{m-1}(ds_{2}, \dots, ds_{m})$$
(3.28)

UNBOUNDED OPERATORS IN FOC

$$+ \int_{0}^{T} \cdots \int_{0}^{s_{4}} A_{k}(s_{m}) \cdots A_{k}(s_{3}) \left\{ \int_{0}^{s_{3}} A_{k}^{(1)}(s_{2}) \lambda_{k}(ds_{2}) - \int_{0}^{s_{3}} A^{(1)}(s_{2}) \lambda(ds_{2}) \right\} \cdot \lambda_{k}^{m-2}(ds_{3}, \dots, ds_{m})$$

$$+ \int_{0}^{T} \cdots \int_{0}^{s_{5}} A_{k}(s_{m}) \cdots A_{k}(s_{4}) \left\{ \int_{0}^{s_{4}} A_{k}^{(2)}(s_{3}) \lambda_{k}(ds_{3}) - \int_{0}^{s_{4}} A^{(2)}(s_{3}) \lambda(ds_{3}) \right\} \cdot \lambda_{k}^{m-3}(ds_{4}, \dots, ds_{m})$$

$$+ \int_{0}^{T} \cdots \int_{0}^{s_{6}} A_{k}(s_{m}) \cdots A_{k}(s_{5}) \left\{ \int_{0}^{s_{5}} A_{k}^{(3)}(s_{4}) \lambda_{k}(ds_{4}) - \int_{0}^{s_{5}} A(s_{4}) \lambda(ds_{4}) \right\} \cdot \lambda_{k}^{m-4}(ds_{5}, \dots, ds_{m})$$

$$+ \cdots + \int_{0}^{T} A_{k}(s_{m}) \left\{ \int_{0}^{s_{m}} A_{k}^{(m-2)}(s_{m-1}) \lambda_{k}(ds_{m-1}) - \int_{0}^{s_{m}} A^{(m-2)}(s_{m-1}) \lambda(ds_{m-1}) \right\} \lambda_{k}(ds_{m})$$

$$+ \left\{ \int_{0}^{T} A_{k}^{(m-1)}(s_{m}) \lambda_{k}(ds_{m}) - \int_{0}^{T} A^{(m-1)}(s_{m}) \lambda(ds_{m}) \right\} .$$

To continue, then, we need to look at the norms of each term above:

$$\left\| \int_{0}^{T} \cdots \int_{0}^{s_{j+1}} A_{k}(s_{m}) \cdots A_{k}(s_{j}) \left\{ \int_{0}^{s_{j}} A_{k}^{(j-2)}(s_{j-1}) \lambda_{k}(ds_{j-1}) - \int_{0}^{s_{j}} A^{(j-2)}(s_{j-1}) \lambda(ds_{j-1}) \right\} \lambda_{k}^{m-(j-1)}(ds_{j}, \dots, ds_{m}) \right\|_{X}.$$
 (3.29)

We can bound this term by

$$\int_{0}^{T} \cdots \int_{0}^{s_{j+1}} \|A_{k}(s_{m})\|_{X} \cdots \|A_{k}(s_{j})\|_{X} \left\{ \left\| \int_{0}^{s_{j}} A_{k}^{(j-2)}(s_{j-1}) \lambda_{k}(ds_{j-1}) - \int_{0}^{s_{j}} A^{(j-2)}(s_{j-1}) \lambda(ds_{j-1}) \right\|_{X} \right\} \lambda_{k}^{k-(j-1)}(ds_{j}, \dots, ds_{m})$$

$$\leq C_{2}^{m-(j-1)} \left\| \int_{0}^{s_{j}} A_{k}^{(j-2)}(s_{j-1}) \lambda_{k}(ds_{j-1}) - \int_{0}^{s_{j}} A^{(j-2)}(s_{j-1}) \lambda(ds_{j-1}) \right\|_{X}$$
(3.30)

using (3.5) and the fact that each of our measures is a probability measure on [0, T].

Now, by definition,

$$A_k^{(j-2)}(s_{j-1}) = A_k(s_{j-1}) \int_0^{s_{j-1}} A^{(j-3)}(s_{j-2}) \,\lambda(ds_{j-2})$$

and

$$A^{(j-2)}(s_{j-1}) = A(s_{j-1}) \int_0^{s_{j-1}} A^{(j-3)}(s_{j-2}) \,\lambda(ds_{j-2}).$$

Since $A_k(\cdot) \to A(\cdot)$ pointwise in X-norm and since

$$\int_0^{s_{j-1}} A^{(j-3)}(s_{j-2}) \,\lambda(ds_{j-2})$$

is norm bounded, it follows that $A_k^{(j-2)}(\cdot) \to A^{(j-2)}(\cdot)$ pointwise in X-norm. Using Lemma 3.10, we are able to conclude that

$$\left\| \int_0^{s_j} A_k^{(j-2)}(s_{j-1}) \,\lambda_k(ds_{j-1}) - \int_0^{s_j} A^{(j-2)}(s_{j-1}) \,\lambda(ds_{j-1}) \right\|_X \to 0$$

as $k \to \infty$. The same lemma shows that

$$\left\| \int_0^{s_2} A_k(s_1) \,\lambda_k(ds_1) - \int_0^{s_2} A(s_1) \,\lambda(ds_1) \right\|_X \to 0$$

and that

$$\left\|\int_0^T A_k^{(m-1)}(s_m)\,\lambda_k(ds_m) - \int_0^T A^{(m-1)}(s_m)\,\lambda(ds_m)\right\|_X \to 0$$

as $k \to \infty$. It follows, then, from the triangle inequality and the inequality (3.30), that

$$\left\| \int_{\Delta_m(T)} A_k(s_m) \cdots A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m) - \int_{\Delta_m(T)} A_k(s_m) \cdots A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m) \right\|_X \to 0$$

as $k \to 0$.

We now consider how to apply the lemma above to the disentangling of

$$\left(\int_0^T A(s)\,\lambda(ds)\right)^m,\tag{3.31}$$

where

$$\lambda := \mu + \sum_{i=1}^{h} p_i \delta_{\tau_i}$$

and where μ is a continuous measure. Using the definition of the disentangling map, we may write the disentangling of the expression (3.31) as

$$\sum_{q_{11}+q_{12}=m} \frac{m!}{q_{11}!q_{12}!} \sum_{j_1+\dots+j_h=q_{12}} \frac{q_{12}!}{j_1!\dots j_h!} \sum_{\theta_1+\dots+\theta_{h+1}=q_{11}} \sum_{\pi\in S_{q_{11}}} \cdots (3.32)$$

$$\int_{\Delta_{q_{11};\theta_1,\dots,\theta_{h+1}}(\pi)} A(s_{\pi(q_{11})}) \cdots A(s_{\pi(\theta_1+\dots+\theta_h+1)}) [p_h A(\tau_h)]^{j_h} A(s_{\pi(\theta_1+\dots+\theta_h)})$$

$$\cdots A(s_{\pi(\theta_1+1)}) [p_1 A(\tau_1)]^{j_1} A(s_{\pi(\theta_1)}) \cdots A(s_{\pi(1)}) \mu^{q_{11}} (ds_1,\dots,ds_{q_{11}}).$$

Now, while one does not obtain (3.32) directly from the integral

$$\int_{\Delta_m(T)} A(s_m) \cdots A(s_1) \lambda_k^m(ds_1, \dots, ds_m),$$

in the particular case we are addressing, the integrals in the disentangling of

$$\left(\int_0^T A(s)\,\lambda(ds)\right)^m,$$

can be obtained from the integral

$$\int_{\Delta_m(T)} A(s_m) \cdots A(s_1) \,\lambda^m(ds_1, \dots, ds_m)$$

when careful attention is paid to preserving the time-ordering of the $A(s_i)$. Indeed, when the time-ordering measure λ has the form $\lambda = \mu + \omega \delta_{\tau}$ for $\tau \in (0,T)$, a tedious calculation of

$$\int_{\Delta_m(T)} A(s_m) \cdots A(s_1) \ (\mu + \omega \delta_\tau)^m \ (ds)$$

$$= \sum_{j=0}^m \int_{\Delta_m; j(T)} A(s_m) \cdots A(s_1) \ (\mu + \omega \delta_\tau)^m \ (ds_1, \dots, ds_m)$$
(3.33)

where we modify, for our purposes here, the definition of $\Delta_{m;j}(T)$ to

$$\Delta_{m;j}(T) := \{ (s_1, \dots, s_m) \in [0, T]^m : 0 < s_1 < \dots < s_j \le \tau < s_{j+1} < \dots < s_m < T \} ,$$

for j = 0, ..., m. The "end" cases are j = 0, where then $\tau < s_1$ and j = m, where $s_m < \tau$. One then writes each term of the sum in (3.33) as an iterated integral and evaluate each integral using the expansion of $(\mu + \omega \delta_{\tau})^m$. For instance, when m = 3, we write

$$(\mu + \omega \delta_{\tau})^3 = \mu^3 + \mu \times \mu \times \omega \delta_{\tau} + \mu \times \omega \delta_{\tau} \times \mu + \omega \delta_{\tau} \times \mu \times \mu + \dots + (\omega \delta_{\tau})^3$$

and evaluate the integral over $\Delta_{m;j}(T)$ as an iterated integral, keeping careful eye on the ordering of the integrand. The resulting integrals turn out to be the same integrals as seen in (3.32) although the multinomial coefficients are missing. Nevertheless, our attention is focussed on the integrals using the sequences $\{A_k(\cdot)\}_{k=1}^{\infty}$ and $\{\lambda_k\}_{k=1}^{\infty}$ and these integrals come from decomposing the integral over $\Delta_m(T)$ as in (3.33). By the lemma, we can write, where the limits are in X-norm,

$$\lim_{k \to \infty} \int_{\Delta_m(T)} A_k(s_m) \cdots A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m)$$
$$= \sum_{j=0}^m \lim_{k \to \infty} \int_{\Delta_m; j(T)} A_k(s_m) \cdots A_k(s_1) \lambda_k^m(ds_1, \dots, ds_m)$$

and so the limit of each term of the sum exists. Consequently, we can use the lemma to conclude that

$$P^m_{\lambda_k}(A_k(\cdot)) \to P^m_{\lambda}(A(\cdot))$$

in X-norm as $k \to \infty$, where $P_{\lambda_k}^m(A_k(\cdot))$ and $P_{\lambda}^m(A(\cdot))$ are written as in (3.32).

3.4. The setting and necessary constructions for joint stability theory. In this section, we introduce the necessary setting which will allow us to investigate joint stability of the operational calculus in the setting of analytic families of operators. (This setting also allows for an improvement of the stability theory in the usual setting of the operational calculus – see the discussion at the end of this paper.)

We take X to be a separable Banach algebra. Select maps $Y_j : [0, T] \to X$, $j = 1, \ldots, n$. Associate to each of the maps $Y_j(\cdot)$ a Borel probability measure λ_j on [0, T]. We will assume that each $Y_j(\cdot)$ is strongly λ_j -measurable. Next, assume that there is, for each $j = 1, \ldots, n$, a $R_j > 0$ for which

$$\sigma\left(Y_{j}(s)\right) \subsetneq B\left(0, R_{j}\right),\tag{3.34}$$

for all $s \in [0, T]$, where $\sigma(x)$ denotes the spectrum of the element $x \in X$ and where B(0, r) is the open ball in \mathbb{C} centered at 0 with radius r. Construct, as in Section 2, the commutative Banach algebra $\mathbb{A}(R_1, \ldots, R_n)$.

For each j = 1, ..., n, assume that $Y_j(\cdot)$ is in $L^1([0, T], X, \lambda_j)$ and define positive real numbers $r_1, ..., r_n$ by

$$r_j := \int_{[0,T]} \|Y_j(s)\|_X \,\lambda_j(ds).$$

Observe that, in view of (3.34),

$$\sup \{ \|Y_j(s)\|_X : s \in [0,T] \} \le R_j$$

and so

$$r_{j} = \int_{[0,T]} \|Y_{j}(s)\|_{X} \lambda_{j}(ds)$$

$$\leq \sup \{\|Y_{j}(s)\|_{X} : s \in [0,T]\} \lambda_{j}([0,T])$$

$$\leq R_{j}$$
(3.35)

for j = 1, ..., n. Construct the commutative Banach algebras $\mathbb{A}(r_1, ..., r_n)$ and $\mathbb{D}(Y_1(\cdot)^{\sim}, ..., Y_n(\cdot)^{\sim})$. In view of (3.35), it follows that

$$\mathbb{A}(R_1,\ldots,R_n) \subseteq \mathbb{A}(r_1,\ldots,r_n).$$
(3.36)

We define the disentangling algebra $\mathbb{D}(R_1, \ldots, R_n)$ (note the different notation here) as the family of all expressions

$$f(Y_1(\cdot)^{\sim}, \dots, Y_n(\cdot)^{\sim}) = \sum_{m_1, \dots, m_n=0}^{\infty} a_{m_1, \dots, m_n} (Y_1(\cdot)^{\sim})^{m_1} \cdots (Y_n(\cdot)^{\sim})^{m_n}$$
(3.37)

for which

$$||f||_{\mathbb{D}(R_1,\dots,R_n)} := \sum_{m_1,\dots,m_n=0}^{\infty} |a_{m_1,\dots,m_n}| R_1^{m_1} \cdots R_n^{m_n} < \infty.$$
(3.38)

The proof that $\mathbb{D}(R_1, \ldots, R_n)$ is a commutative Banach algebra is the same as that seen in [15, Chapter 2] as is the proof that $\mathbb{A}(R_1, \ldots, R_n)$ is isometrically isomorphic to $\mathbb{D}(R_1, \ldots, R_n)$. Note that

$$\mathbb{D}(R_1,\ldots,R_n) \subseteq \mathbb{D}(Y_1(\cdot)^{\sim},\ldots,Y_n(\cdot)^{\sim}).$$
(3.39)

Now, let $\ell \in \mathbb{N}$ and let $j \in \{1, \ldots, n\}$. Define

$$p_j^{\ell}(Y_1(\cdot)^{\sim}, \dots, Y_n(\cdot)^{\sim}) := [Y_j(\cdot)^{\sim}]^{\ell}.$$
 (3.40)

It is clear that $p_j^{\ell} \in \mathbb{D}(R_1, \ldots, R_n)$ and that

$$\|p_{j}^{\ell}\|_{\mathbb{D}(R_{1},...,R_{n})} = R_{j}^{\ell} \left(= \left\| (Y_{j}(\cdot)^{\sim})^{\ell} \right\|_{\mathbb{D}(R_{1},...,R_{n})} \right).$$
(3.41)

Knowing this relation, it is clear that, for each j = 1, ..., n,

$$\operatorname{spr}\left(p_{j}^{1}\right) = \lim_{\ell \to \infty} \left\| p_{j}^{\ell} \right\|_{\mathbb{D}(R_{1},\dots,R_{n})}^{1/\ell} = \lim_{\ell \to \infty} \left(R_{j}^{\ell} \right)^{1/\ell} = R_{j},$$

where $\operatorname{spr}(x)$ denotes the spectral radius of an element $x \in X$. From the definition of R_j , $j = 1, \ldots, n$, it follows that

$$R_j \ge \sup \{ \operatorname{spr}(Y_j(s)) : s \in [0, T] \}$$
 (3.42)

and so

$$\operatorname{spr}(Y_j(\cdot)^{\sim}) \ge \operatorname{spr}(Y_j(s))$$
 (3.43)

for all $s \in [0, T]$.

With an eye towards using Cauchy's integral theorm in our formalism, we begin by considering the function

$$h: \rho(Y_1(\cdot)^{\sim}) \times \cdots \times \rho(Y_n(\cdot)^{\sim}) \longrightarrow \mathbb{D}(R_1, \dots, R_n)$$

by

$$h(\xi_1, \dots, \xi_n; Y_1(\cdot)^{\sim}, \dots, Y_n(\cdot)^{\sim}) := \\ \xi_1 \cdots \xi_n (\xi_1 - Y_1(\cdot)^{\sim})^{-1} \cdots (\xi_n - Y_n(\cdot)^{\sim})^{-1}. \quad (3.44)$$

Note that we may write

$$h(\xi_1, \dots, \xi_n; Y_1(\cdot)^{\sim}, \dots, Y_n(\cdot)^{\sim}) = \sum_{m_1, \dots, m_n=0}^{\infty} \xi^{-m_1} \cdots \xi_n^{-m_n} (Y_1(\cdot)^{\sim})^{m_1} \cdots (Y_n(\cdot)^{\sim})^{m_n} \quad (3.45)$$

and this series converges because

$$\xi_j > ||Y_j(\cdot)^{\sim}||_{\mathbb{D}(R_1,...,R_n)} = R_j$$

and

$$R_j > \sup \{ \|Y_j(s)\|_X : s \in [0, T] \}$$

for each j = 1, ..., n. So, for $\xi_1, ..., \xi_n \in \mathbb{C}, |\xi_j| > ||Y_j(\cdot)^{\sim}||_{\mathbb{D}(R_1,...,R_n)}, j = 1, ..., n,$

$$h(\xi_1,\ldots,\xi_n:Y_1(\cdot)^{\sim},\ldots,Y_n(\cdot)^{\sim})\in \mathbb{D}(R_1,\ldots,R_n).$$

It is clear that h is continuous.

Now, let

$$P_0 := \{(z_1, \dots, z_n) : |z_j| \le R_j, j = 1, \dots, n\}$$
(3.46)

and choose sequences $\{\epsilon_{j,k}\}_{k=1}^{\infty}$, j = 1, ..., n, of positive real numbers for which

$$\epsilon_{j,k} \searrow 0 \tag{3.47}$$

as $k \to \infty$ for each j = 1, ..., n. (Note: we use the symbol $a_k \searrow a$ to specify that the sequence $\{a_k\}_{k=1}^{\infty}$ is a decreasing sequence which converges to a.) Given $j \in \{1, ..., n\}$ and $k \in \mathbb{N}$, we define

$$R_{j,k} := R_j + \epsilon_{j,k}. \tag{3.48}$$

We use these numbers to define, for $k \in \mathbb{N}$,

$$P_k := \{ (z_1, \dots, z_n) : |z_j| \le R_{j,k}, j = 1, \dots, n \}.$$
(3.49)

Clearly $P_{k+1} \subseteq P_k$ for all $k \in \mathbb{N}$ and so $\mathbb{A}_k \subseteq \mathbb{A}_{k+1}$, where, for any $\ell \in \mathbb{N}$,

$$\mathbb{A}_{\ell} := \mathbb{A}\left(R_{1,\ell}, \dots, R_{n,\ell}\right). \tag{3.50}$$

For $k, \ell \in \mathbb{N}$ with $k < \ell$, the map $g_{\ell k} : \mathbb{A}_k \to \mathbb{A}_\ell$ defined by

$$g_{\ell k}(f) := f|_{P_{\ell}}.$$
(3.51)

It is clear that $g_{\ell k}$ is linear and bounded. Indeed, linearity is obvious and we have

$$||g_{\ell k}(f)||_{\mathbb{A}_{\ell}} = ||f|_{P_{\ell}}||_{\mathbb{A}_{\ell}} \le ||f||_{\mathbb{A}_{k}}$$

This system of maps and algebras is an inductive system (see [1], for example) and we define

$$\mathbb{A}_{\infty} := \lim g_{\ell k}(\mathbb{A}_k), \tag{3.52}$$

the inductive limit of the inductive system determined by the maps $g_{\ell k}$ and the Banach algebras \mathbb{A}_k . Since the construction of an inductive system is categorical, \mathbb{A}_{∞} is itself a commutative Banach algebra. with norm

$$||f||_{\mathbb{A}_{\infty}} = \sup \{ ||f||_{\mathbb{A}_{k}} : k \in \mathbb{N} \}.$$
(3.53)

Note that, if $f \in \mathbb{A}_{\infty}$, f is analytic on a polydisk containing P_0 and there is a least integer $k_0 \in \mathbb{N}$ for which $f \in \mathbb{A}_{k_0}$ and $f \in \mathbb{A}_k$ for all $k \geq k_0$. The same construction with the associated (and isometrically isomorphic) disentangling algebras allows us to define

$$\mathbb{D}_{\infty} := \lim_{\to} g_{\ell k}(\mathbb{D}_k). \tag{3.54}$$

Definition 3.12. For each $k \in \mathbb{N}$, let $\mathcal{T}_{\mu_1,\ldots,\mu_n}^{(k)}$ denote the disentangling map with domain \mathbb{D}_k using time-ordering directions supplied by the time-ordering measures μ_1, \ldots, μ_n .

Definition 3.13. Given $f \in \mathbb{D}_{\infty}$, let $k_0 \in \mathbb{N}$ be the least integer for which $f \in \mathbb{D}_{k_0}$. Define

$$\mathcal{T}_{\mu_1,\dots,\mu_n} f := \mathcal{T}_{\mu_1,\dots,\mu_n}^{(k_0)} f, \qquad (3.55)$$

with the disentangling is done using time-ordering directions supplied by the time-ordering measures μ_1, \ldots, μ_n . Note that, if $k \ge k_0$, then

$$\mathcal{T}^{(k)}_{\mu_1,...,\mu_n}f = \mathcal{T}^{(k_0)}_{\mu_1,...,\mu_n}f$$

Also, it follows from Proposition IV.5.6 of [1] (concerning inductive systems) that $\mathcal{T}_{\mu_1,\ldots,\mu_n}$ is continuous from \mathbb{D}_{∞} to X.

Remark 3.14. In each of the definitions above, we can use the algebras \mathbb{A}_k and \mathbb{A}_{∞} , which are isometrically isomorphic to the corresponding disentangling algebras \mathbb{D}_k and \mathbb{D}_{∞} .

We now take note of how we can use Cauchy's integral theorem in the setting of this section. Given $f \in \mathbb{D}_{\infty}$ (or \mathbb{A}_{∞}), we let k_0 be the least positive integer for which $f \in \mathbb{D}_{k_0}$. Returning to our X-valued functions $Y_j(\cdot)$ and their associated time-ordering measures λ_j and using [26], we may write

$$\mathcal{T}_{\lambda_{1},\dots\lambda_{n}}f(Y_{1}(\cdot)^{\sim},\dots,Y_{n}(\cdot)^{\sim}) = \mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}^{(k_{0})}f(Y_{1}(\cdot)^{\sim},\dots,Y_{n}(\cdot)^{\sim}) \\
= (2\pi i)^{-n} \int_{|\xi_{1}|=R_{1,k_{0}}} \cdots \int_{|\xi_{n}|=R_{n,k_{0}}} f(\xi_{1},\dots,\xi_{n})\mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}^{(k_{0}+1)} \left((\xi_{1}-Y_{1}(\cdot)^{\sim})^{-1}\cdots (\xi_{n}-Y_{n}(\cdot)^{\sim})^{-1}\right) d\xi_{1}\cdots d\xi_{n}.$$
(3.56)

Define real numbers M_j , $j = 1, \ldots, n$, by

$$M_j := R_j + \epsilon_{j,1} \tag{3.57}$$

and modify, if necessary, $\epsilon_{j,1}$ so that $M_j > 1$. Next, define

 $\mathcal{L}_{M_1,\dots,M_n} := \{ (B_1(\cdot),\dots,B_n(\cdot)) : [0,T] \to X^n : B_j : [0,T] \to X \text{ is strongly} \\ \lambda_j \text{-measurable and } \sup \{ \|B_j(s)\|_X : s \in [0,T] \} < M_j, j = 1,\dots,n \}.$ (3.58)

Construct the commutative Banach algebra $\mathbb{A}(M_1, \ldots, M_n)$. For $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$, let

$$\beta_j^B := \int_{[0,T]} \|B_j(s)\|_X \,\lambda_j(ds) \tag{3.59}$$

for j = 1, ..., n. Construct the commutative Banach algebra $\mathbb{A}(\beta_1^B, ..., \beta_n^B)$. We will, now and in the sequel, associate the Borel probability measure λ_j on [0,T] to $B_j(\cdot)$, the jth component of the *n*-tuple $(B_1(\cdot),\ldots,B_n(\cdot))$. Note that, because $\beta_j^B \leq M_j$ for each $j = 1,\ldots,n$,

$$\mathbb{A}(M_1,\ldots,M_n) \subseteq \mathbb{A}\left(\beta_1^B,\ldots,\beta_n^B\right).$$
(3.60)

This containment holds for any *n*-tuple $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$. Since, for every $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$, $\mathbb{A}\left(\beta_1^B, \ldots, \beta_n^B\right)$ is isometrically isomorphic to

$$\mathbb{D}\left(\left(B_1(\cdot),\lambda_1\right)^{\sim},\ldots,\left(B_n(\cdot),\lambda_n\right)^{\sim}\right),\right.$$

we can consider $\mathbb{A}(M_1, \ldots, M_n)$ to be a subalgebra of every disentangling algebra $\mathbb{D}((B_1(\cdot), \lambda_1)^{\sim}, \ldots, (B_n(\cdot), \lambda_n)^{\sim})$ with $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$. Hence, any $f \in \mathbb{A}(M_1, \ldots, M_n)$ can be taken to be an element of

$$\mathbb{D}\left((B_1(\cdot),\lambda_1)^{\sim},\ldots,(B_n(\cdot),\lambda_n)^{\sim}\right)$$

for every *n*-tuple $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$.

We will now, in the definition (3.58), replace the phrase "strongly λ_j -measurable" with "continuous." To be explicit, we now take

$$\mathcal{L}_{M_1,\dots,M_n} := \{ (B_1(\cdot),\dots,B_n(\cdot)) : [0,T] \to X^n : B_j : [0,T] \to X \text{ is} \\ \text{continuous and } \sup \{ \|B_j(s)\|_X : s \in [0,T] \} < M_j, j = 1,\dots,n \} .$$

Now, let

$$b := (B_1(\cdot), \dots, B_n(\cdot)), c := (C_1(\cdot), \dots, C_n(\cdot))$$

be elements of $\mathcal{L}_{M_1,\ldots,M_n}$. Also, select sequences $\{B_{j,k}(\cdot)\}_{k=1}^{\infty}, \{C_{j,k}(\cdot)\}_{k=1}^{\infty}$ for which $B_{j,k}(\cdot) \to B_j(\cdot)$ and $C_{j,k}(\cdot) \to C_j(\cdot) \lambda_j$ -almost everywhere in norm for each $j = 1, \ldots, n$ and assume that

$$b_k := (B_{1,k}(\cdot), \dots, B_{n,k}(\cdot)), c_k := (C_{1,k}(\cdot), \dots, C_{n,k}(\cdot))$$

are elements of $\mathcal{L}_{M_1,\ldots,M_n}$ for all $k \in \mathbb{N}$. We also select sequences $\{\lambda_{j,k}\}_{k=1}^{\infty}$, $j = 1, \ldots, n$, of Borel probability measures on [0, T] such that $\lambda_{j,k} \rightharpoonup \lambda_j$ as $k \rightarrow \infty$. Our goal is to connect the convergence of the sequences

$$\left\{f_{\lambda_{1,k},\dots,\lambda_{n,k}}\left(b_{k}\right)\right\}_{k=1}^{\infty}$$
 and $\left\{f_{\lambda_{1,k},\dots,\lambda_{n,k}}\left(c_{k}\right)\right\}_{k=1}^{\infty}$

We will continue using the notation $b = (B_1(\cdot), \ldots, B_n(\cdot))$ for elements of $\mathcal{L}_{M_1,\ldots,M_n}$ and we will use \tilde{b} for the corresponding *n*-tuple of formal objects. Define

$$\mathbb{D}_{\oplus} := \bigoplus_{b \in \mathcal{L}_{M_1, \dots, M_n}} \mathbb{D}(\tilde{b}).$$
(3.61)

Then \mathbb{D}_{\oplus} is a commutative Banach algebra under the norm

$$\left\| \{f_b\}_{b \in \mathcal{L}_{M_1,\dots,M_n}} \right\|_{\mathbb{D}_{\oplus}} = \sup \left\{ \|f_b\|_{\mathbb{D}(\tilde{b})} : b \in \mathcal{L}_{M_1,\dots,M_n} \right\}.$$
(3.62)

Next, define

$$\mathscr{C} := \left\{ \theta_f \in \mathbb{D}_{\oplus} : \theta_f = \{f_b\}_{b \in \mathcal{L}_{M_1, \dots, M_n}}, f_b \equiv f \in \mathbb{A} \left(M_1, \dots, M_n \right) \text{ for all} \\ b \in \mathcal{L}_{M_1, \dots, M_n} \right\}.$$
(3.63)

Clearly \mathscr{C} is a subalgebra of \mathbb{D}_{\oplus} and, for $\theta_f \in \mathscr{C}$,

$$\|\theta_f\|_{\mathbb{D}_{\oplus}} = \|f\|_{\mathbb{A}(M_1,\dots,M_n)}.$$
 (3.64)

Define $\Gamma: \mathscr{C} \to \mathbb{A}(M_1, \ldots, M_n)$ by

$$\Gamma(\theta_f) := f. \tag{3.65}$$

Clearly Γ is a linear bijection and, in view of (3.65), is an isometry. The same is true for Γ^{-1} and so Γ is an isometric isomorphism.

To continue, define $\Pi : \mathcal{L}_{M_1,\ldots,M_n} \times \mathscr{C} \to \mathbb{A}(M_1,\ldots,M_n)$ by

$$\Pi(b,\theta_f) := \Gamma(\theta_f)(\tilde{b}) = f(\tilde{b}).$$
(3.66)

For each fixed $b \in \mathcal{L}_{M_1,\ldots,M_n}$, $\Pi(b,\cdot)$ is the canonical projection on \mathbb{D}_{\oplus} restricted to \mathscr{C} . Hence $\Pi(b,\cdot)$ is continuous. For fixed θ_f , in view of the identification of $\mathbb{A}(M_1,\ldots,M_n)$ as a subalgebra of $\mathbb{D}(\tilde{b})$ for every $b \in \mathcal{L}_{M_1,\ldots,M_n}$, $\Pi(\theta_f,\cdot)$ is constant-valued and so continuous.

We now put time-ordering measures into the mix. Let $\mathscr{P}_{\text{Borel}}([0,T])$ be the space of Borel probability measures on [0,T] with the topology of weak convergence. Let $\mathscr{P}_{\text{finite}}([0,T])$ be the subspace of $\mathscr{P}_{\text{Borel}}([0,T])$ consisting of Borel probability measures with finitely supported discrete parts (we will take $\mathscr{P}_{\text{finite}}([0,T])$ to contain all continuous Borel probability measures on [0,T]). Define

$$T: \left[\mathscr{P}_{\text{finite}}\left([0,T]\right)\right]^n \times \mathcal{L}_{M_1,\dots,M_n} \times \mathscr{C} \to X$$

by

$$T\left(\left(\lambda_{1},\ldots,\lambda_{n}\right),b,\theta_{f}\right) := f_{\lambda_{1},\ldots,\lambda_{n}}\left(b\right).$$

$$(3.67)$$

It is clear that

$$T = \mathcal{T}_{\lambda_1, \dots, \lambda_n} \circ \Pi.$$

We now come to the following, crucial, proposition.

Proposition 3.15. For each $1 \leq j \leq n$ let $\{\lambda_{j,k}\}_{k=1}^{\infty}$ be a sequence of Borel probability measures from $\mathscr{P}_{finite}([0,T])$ for which $\lambda_{j,k} \rightarrow \lambda_j$ as $k \rightarrow \infty$, where $\lambda_j \in \mathscr{P}_{finite}([0,T])$. Let $(B_{1,k}(\cdot), \ldots, B_{n,k}(\cdot)) \in \mathcal{L}_{M_1,\ldots,M_n}$ for every $k \in \mathbb{N}$ and let $(C_{1,k}(\cdot), \ldots, C_{n,k}(\cdot)) \in \mathcal{L}_{M_1,\ldots,M_n}$ for every $k \in \mathbb{N}$. Assume that $B_{j,k}(\cdot) \rightarrow B_j(\cdot)$ as $k \rightarrow \infty$ and that $C_{j,k}(\cdot) \rightarrow C_j(\cdot)$ as $k \rightarrow \infty \lambda_j$ -almost everywhere in [0,T]. We will denote our n-tuples as b, b_k, c, c_k . For $f \in \mathbb{A}(M_1,\ldots,M_n)$,

$$f_{\lambda_{1,k},\dots,\lambda_{n,k}}(b_k) \to f_{\lambda_1,\dots,\lambda_n}(b) \quad iff \ f_{\lambda_{1,k},\dots,\lambda_{n,k}}(c_k) \to f_{\lambda_1,\dots,\lambda_n}(c) \,. \tag{3.68}$$

Proof. We may write, for $f \in \mathbb{A}(M_1, \ldots, M_n)$,

$$\begin{split} f_{\lambda_{1,k},\dots,\lambda_{n,k}} \left(b_k \right) &- f_{\lambda_1,\dots,\lambda_n} \left(b \right) \\ &= \mathcal{T}_{\lambda_{1,k},\dots,\lambda_{n,k}} \left(\Pi \left(b_k, \theta_f \right) \right) - \mathcal{T}_{\lambda_1,\dots,\lambda_n} \left(\Pi \left(b, \theta_f \right) \right) \\ &= \mathcal{T}_{\lambda_{1,k},\dots,\lambda_{n,k}} \left(\Pi \left(b_k, \theta_f \right) - \Pi \left(c_k, \theta_f \right) \right) + \mathcal{T}_{\lambda_{1,k},\dots,\lambda_{n,k}} \Pi \left(c_k, \theta_f \right) \\ &- \mathcal{T}_{\lambda_1,\dots,\lambda_n} \Pi \left(c, \theta_f \right) - \mathcal{T}_{\lambda_1,\dots,\lambda_n} \left(\Pi \left(c, \theta_f \right) - \Pi \left(b, \theta_f \right) \right) \\ &= \mathcal{T}_{\lambda_{1,k},\dots,\lambda_{n,k}} \Pi \left(c_k, \theta_f \right) - \mathcal{T}_{\lambda_1,\dots,\lambda_n} \Pi \left(c, \theta_f \right), \end{split}$$

where we've use the fact that $\Pi(\cdot, \theta_f)$ is constant-valued for each θ_f . \Box

We now apply Cauchy's integral theorem (as in [26]) using the definition of the numbers M_1, \ldots, M_n . Let $f \in \mathbb{A}(M_1, \ldots, M_n)$; we know that $f \in \mathbb{A}_{\infty}$ (see (3.52), above). Also, let $b \in \mathcal{L}_{M_1,\ldots,M_n}$. We have

$$f_{\lambda_1,\dots,\lambda_n}(b) = (2\pi i)^{-n} \int_{|\xi_1|=M_1} \cdots \int_{|\xi_n|=M_n} f(\xi_1,\dots,\xi_n) \mathcal{T}^{(2)}_{\lambda_1\dots,\lambda_n} \left((\xi_1 - B_1(\cdot)^{\sim})^{-1} \cdots (\xi_n - B_n(\cdot)^{\sim})^{-1} \right) d\xi_1 \cdots d\xi_n.$$
(3.69)

We set

$$B_0 := \{ (w_1, \dots, w_n) : |w_1| < 1, \dots, |w_n| < 1 \}.$$
Fix $f \in \mathbb{A}(M_1, \dots, M_n) \subseteq \mathbb{A}_{\infty}$ and define
$$(3.70)$$

$$\mathscr{D}_{f}^{B} := \{\mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}f(b): (\lambda_{1},\dots,\lambda_{n}) \in [\mathscr{P}_{\text{finite}}([0,T])]^{n}, b \in \mathcal{L}_{M_{1},\dots,M_{n}}\}.$$
(3.71)

Given $(\lambda_1, \ldots, \lambda_n) \in [\mathscr{P}_{\text{finite}}([0,T])]^n$ and $b \in \mathcal{L}_{M_1,\ldots,M_n}$ we define $g(\lambda_1, \ldots, \lambda_n; b; f; \cdot) : B_0 \to X$

by

$$g(\lambda_{1}, \dots, \lambda_{n}; b; f; w_{1}, \dots, w_{n})$$

$$= (2\pi i)^{-n} \int_{|\xi_{1}|=M_{1}} \cdots \int_{|\xi_{n}|=M_{n}} f(w_{1}\xi_{1}, \dots, w_{n}\xi_{n}) \mathcal{T}_{\lambda_{1},\dots,\lambda_{n}} \left((\xi_{1} - B_{1}(\cdot)^{\sim})^{-1} \cdots (\xi_{n} - B_{n}(\cdot)^{\sim})^{-1} \right) d\xi_{1} \cdots d\xi_{n}.$$
(3.72)

Note that

$$g(\lambda_1, \dots, \lambda_n; b; f; 1, \dots, 1) = \mathcal{T}_{\lambda_1, \dots, \lambda_n} f(b).$$
(3.73)

We can, since f is analytic, differentiate under the integral sign and so, for $(\lambda_1, \ldots, \lambda_n) \in [\mathscr{P}_{\text{finite}}([0, T])]^n$ and for $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$ it follows that $g(\lambda_1, \ldots, \lambda_n; B_1(\cdot), \ldots, B_n(\cdot); f; \cdot)$ is analytic as a function on B_0 . Now define

$$\mathscr{F}_{\mathcal{D}_{f}^{B}} := \{g(\lambda_{1}, \dots, \lambda_{n}; B_{1}(\cdot), \dots, B_{n}(\cdot); f; \cdot) : \\ (\lambda_{1}, \dots, \lambda_{n}) \in [\mathscr{P}_{\text{finite}}([0, T])]^{n}, (B_{1}(\cdot), \dots, B_{n}(\cdot)) \in \mathcal{L}_{M_{1}, \dots, M_{n}}\}$$
(3.74)

We claim that $\mathscr{F}_{\mathcal{D}^B_{\ell}}$ is a normal family of X-valued functions.

To establish this claim, we start by computing, for $(\lambda_1, \ldots, \lambda_n) \in [\mathscr{P}_{\text{finite}}([0,T])]^n$ and $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1,\ldots,M_n}$, a bound for

$$\left\|\mathcal{T}_{\lambda_1,\dots,\lambda_n}^{(2)}\left(\left(\xi_1-B_1(\cdot)^{\sim}\right)^{-1}\cdots\left(\xi_n-B_n(\cdot)^{\sim}\right)^{-1}\right)\right\|_X$$

We write (without yet specifying where the complex numbers ξ_1, \ldots, ξ_n are located)

$$(\xi_1 - B_1(\cdot)^{\sim})^{-1} \cdots (\xi_n - B_n(\cdot)^{\sim})^{-1}$$

$$=\sum_{m_1,\dots,m_n=0}^{\infty}\xi_1^{-m_1-1}\cdots\xi_n^{-m_n-1}P^{m_1,\dots,m_n}\left(B_1(\cdot)^{\sim},\dots,B_n(\cdot)^{\sim}\right)$$

and so

$$\mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}^{(2)}\left((\xi_{1}-B_{1}(\cdot)^{\sim})^{-1}\cdots(\xi_{n}-B_{n}(\cdot)^{\sim})^{-1}\right) = \sum_{m_{1},\dots,m_{n}=0}^{\infty}\xi_{1}^{-m_{1}-1}\cdots\xi_{n}^{-m_{n}-1}P_{\lambda_{1},\dots,\lambda_{n}}^{m_{1},\dots,m_{n}}\left(B_{1}(\cdot),\dots,B_{n}(\cdot)\right).$$
(3.75)

Using the norm on \mathbb{A}_2 (because we're applying $\mathcal{T}^{(2)}$) we have

$$\left\|\mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}^{(2)}P^{m_{1},\dots,m_{n}}\left(B_{1}(\cdot)^{\sim},\dots,B_{n}(\cdot)^{\sim}\right)\right\|_{X} \leq (R_{1}+\epsilon_{1,2})^{m_{1}}\cdots(R_{n}+\epsilon_{n,2})^{m_{n}}$$

It then follows that

$$\left\| \mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}^{(2)} \left((\xi_{1} - B_{1}(\cdot)^{\sim})^{-1} \cdots (\xi_{n} - B_{n}(\cdot)^{\sim})^{-1} \right) \right\|_{X}$$

$$\leq \sum_{m_{1},\dots,m_{n}=0}^{\infty} |\xi_{1}|^{-m_{1}-1} \cdots |\xi_{n}|^{-m_{n}-1} (R_{1} + \epsilon_{1,2})^{m_{1}} \cdots (R_{1} + \epsilon_{n,2})^{m_{n}}.$$

Recalling now the definition of the numbers M_1, \ldots, M_n (see (3.57), above), we take $\xi_1, \ldots, \xi_n \in \mathbb{C}$ such that $|\xi_j| = M_j$, $j = 1, \ldots, n$, we therefore have the claimed norm bound:

$$\left\| \mathcal{T}_{\lambda_{1},\dots,\lambda_{n}}^{(2)} \left(\left(\xi_{1} - B_{1}(\cdot)^{\sim}\right)^{-1} \cdots \left(\xi_{n} - B_{n}(\cdot)^{\sim}\right)^{-1} \right) \right\|_{X} \le \sum_{m_{1},\dots,m_{n}=0}^{\infty} \left| \xi_{1} \right|^{-m_{1}-1} \cdots \left| \xi_{n} \right|^{-m_{n}-1} \left(R_{1} + \epsilon_{1,2} \right)^{m_{1}} \cdots \left(R_{1} + \epsilon_{n,2} \right)^{m_{n}}$$

$$\le \sum_{m_{1},\dots,m_{n}=0}^{\infty} \left(\frac{R_{1} + \epsilon_{1,2}}{R_{1} + \epsilon_{1,1}} \right)^{m_{1}} \cdots \left(\frac{R_{n} + \epsilon_{n,2}}{R_{n} + \epsilon_{n,1}} \right)^{m_{n}}$$

$$=: C_{0}.$$
(3.76)

We can now use the norm bound C_0 to compute, since f is analytic on the interior of the polydisk P_{M_1,\ldots,M_n} ,

$$\|g(\lambda_{1},...,\lambda_{n};B_{1}(\cdot),...,B_{n}(\cdot);f;w_{1},...,w_{n})\|_{X} \leq (2\pi)^{-n}C_{0} \sup_{\substack{(w_{1},...,w_{n})\in B_{0}\\|\xi_{1}|=M_{1}}} \int \cdots \int_{|\xi_{n}|=M_{n}} |f(w_{1}\xi_{1},...,w_{n}\xi_{n})| d\xi_{1}\cdots d\xi_{n}$$
(3.77)

 $\leq (2\pi)^{-n}C_0C_1,$

where C_1 is a positive constant for which

$$\sup_{(w_1,\dots,w_n)\in B_0} \int_{|\xi_1|=M_1} \cdots \int_{|\xi_n|=M_n} |f(w_1\xi_1,\dots,w_n\xi_n)| \, d\xi_1 \cdots d\xi_n \le C_1. \quad (3.78)$$

If follows that $\mathscr{F}_{\mathcal{D}^B_t}$ is a normal family of X-valued analytic functions.

We now assume, if necessary, that $R_j > 1$ for all $j = 1, \ldots, n$. Fix an $f \in \mathbb{A}(M_1, \ldots, M_n)$. Let $\{\lambda_{j,k}\}_{k=1}^{\infty}, j = 1, \ldots, n$, be sequences from $\mathscr{P}_{\text{finite}}([0,T])$ which converge weakly to $\lambda_j \in \mathscr{P}_{\text{finite}}([0,T])$ for each $j = 1, \ldots, n$. Next, let $\{B_{j,k}(\cdot)\}_{k=1}^{\infty}, j = 1, \ldots, n$, be sequences of continuous Xvalued functions which converge pointwise to continuous X-valued functions $B_j(\cdot), j = 1, \ldots, n$. Assume that the *n*-tuple $(B_{1,k}(\cdot), \ldots, B_{n,k}(\cdot))$ is an element of $\mathcal{L}_{M_1,\ldots,M_n}$ for every $k \in \mathbb{N}$ and assume that the *n*-tuple

$$(B_1(\cdot),\ldots,B_n(\cdot)) \in \mathcal{L}_{M_1,\ldots,M_n}.$$

These *n*-tuples induce the sequence

$$\{g(\lambda_{1,k},\ldots,\lambda_{n,k};B_{1,k}(\cdot),\ldots,B_{n,k}(\cdot);f;\cdot)\}_{k=1}^{\infty}$$
(3.79)

from the normal family $\mathscr{F}_{\mathcal{D}_{f}^{B}}$. Because $\mathscr{F}_{\mathcal{D}_{f}^{B}}$ is a normal family of X-valued analytic functions, there is a subsequence

$$\{g(\lambda_{1,k_{\ell}},\ldots,\lambda_{n,k_{\ell}};B_{1,k_{\ell}}(\cdot),\ldots,B_{n,k_{\ell}}(\cdot);f;\cdot)\}_{\ell=1}^{\infty}$$
(3.80)

which converges uniformly on compact subsets of B_0 . Since

$$g(\lambda_{1,k_{\ell}},\dots,\lambda_{n,k_{\ell}};B_{1,k_{\ell}}(\cdot),\dots,B_{n,k_{\ell}}(\cdot);f;w_{1},\dots,w_{n}) = (2\pi i)^{-n} \int_{|\xi_{1}|=M_{1}} \cdots \int_{|\xi_{n}|=M_{n}} f(w_{1}\xi_{1},\dots,w_{n}\xi_{n}) \cdot$$

$$\mathcal{T}^{(2)}_{\lambda_{1,k_{\ell}},\dots,\lambda_{n,k_{\ell}}} \left((\xi_{1}-B_{1,k_{\ell}}(\cdot)^{\sim})^{-1} \cdots (\xi_{n}-B_{n,k_{\ell}}(\cdot)^{\sim})^{-1} \right) d\xi_{1}\cdots d\xi_{n},$$
(3.81)

if we apply Lebesgue's dominated convergence theorem for Bochner integrals, we are able to write

$$\lim_{\ell \to \infty} (2\pi i)^{-n} \int_{|\xi_1| = M_1} \cdots \int_{|\xi_n| = M_n} f(w_1 \xi_1, \dots, w_n \xi_n) \cdot \mathcal{T}_{\lambda_{1,k_\ell}, \dots, \lambda_{n,k_\ell}}^{(2)} \left((\xi_1 - B_{1,k_\ell}(\cdot)^{\sim})^{-1} \cdots (\xi_n - B_{n,k_\ell}(\cdot)^{\sim})^{-1} \right) d\xi_1 \cdots d\xi_n$$
$$= (2\pi i)^{-n} \int_{|\xi_1| = M_1} \cdots \int_{|\xi_n| = M_n} f(w_1 \xi_1, \dots, w_n \xi_n)$$
$$\left[\lim_{\ell \to \infty} \mathcal{T}_{\lambda_{1,k_\ell}, \dots, \lambda_{n,k_\ell}}^{(2)} \left((\xi_1 - B_{1,k_\ell}(\cdot)^{\sim})^{-1} \cdots (\xi_n - B_{n,k_\ell}(\cdot)^{\sim})^{-1} \right) \right] d\xi_1 \cdots d\xi_n$$

where the limit is in X-norm on compact subsets of B_0 . However, this being said, it is easy to see from the limit above, that it therefore must be that the sequence

$$\left\{\mathcal{T}^{(2)}_{\lambda_{1,k_{\ell}},\ldots,\lambda_{n,k_{\ell}}}\left((\xi_{1}-B_{1,k_{\ell}}(\cdot)^{\sim})^{-1}\cdots(\xi_{n}-B_{n,k_{\ell}}(\cdot)^{\sim})^{-1}\right)\right\}_{\ell=1}^{\infty}$$

converges in norm on X. But, by inspecting the disentangling of the Cauchy kernel (see (3.75)), it must be that the sequence

$$\left\{P_{\lambda_{1,k_{\ell}},\ldots,\lambda_{n,k_{\ell}}}^{m_{1},\ldots,m_{n}}\left(B_{1,k_{\ell}}(\cdot),\ldots,B_{n,k_{\ell}}(\cdot)\right)\right\}_{\ell=1}^{\infty}$$

must converge in X-norm for each $m_1, \ldots, m_n \in \mathbb{N} \cup \{0\}$. It remains to determine the limit of this sequence.

Let $m_j \neq 0$ and $m_i = 0$ for $i \neq j, 1 \leq i, j \leq n$. We can write, in view of Lemma 3.11,

$$\lim_{\ell \to \infty} P^{0,\dots,0,m_j,0,\dots,0}_{\lambda_{1,k_{\ell}},\dots,\lambda_{j-1,k_{\ell}},\lambda_{j,k_{\ell}},\lambda_{j+1,k_{\ell}},\dots,\lambda_{n,k_{\ell}}} (B_{1,k_{\ell}}(\cdot),\dots,B_{n,k_{\ell}}(\cdot))$$
$$= P^{0,\dots,0,m_j,0,\dots,0}_{\lambda_{1},\dots,\lambda_{j-1},\lambda_{j},\lambda_{j+1},\dots,\lambda_n} (B_1(\cdot),\dots,B_n(\cdot))$$

in X-norm. To proceed further, we observe that, for any Borel probability measures ν_1, \ldots, ν_n on [0, T] and for any $m_1, \ldots, m_n \in \mathbb{N} \cup \{0\}$,

$$P_{\nu_1,\dots,\nu_{j-1},\nu_j,\nu_{j+1},\dots,\nu_n}^{m_1,\dots,m_j,m_{j+1},\dots,m_n}(I,\dots,I,B_j,I,\dots,I) = P_{\nu_1,\dots,\nu_{j-1},\nu_j,\nu_{j+1},\dots,\nu_n}^{0,\dots,0,m_j,0,\dots,0}(B_1(\cdot),\dots,B_{j-1}(\cdot),B_j(\cdot),B_{j+1}(\cdot),\dots,B_n(\cdot)),$$

and so, in norm on X,

$$\lim_{\ell \to \infty} P^{m_1, \dots, m_{j-1}, m_j, m_{j+1}, \dots, m_n}_{\lambda_{1,k_\ell}, \dots, \lambda_{j-1,k_\ell}, \lambda_{j,k_\ell}, \lambda_{j+1,k_\ell}, \dots, \lambda_{n,k_\ell}} (I, \dots, I, B_{j,k_\ell}(\cdot), I, \dots, I) = P^{m_1, \dots, m_{j-1}, m_j, m_{j+1}, \dots, m_n}_{\lambda_{1,\dots,\lambda_{j-1},\lambda_j, \lambda_{j+1}, \dots, \lambda_n}} (I, \dots, I, B_j(\cdot), I, \dots, I).$$
(3.82)

It therefore follows, from Proposition 3.15 that, in X-norm,

$$\lim_{\ell \to \infty} P^{m_1,\dots,m_n}_{\lambda_{1,k_\ell},\dots,\lambda_{n,k_\ell}} \left(B_{1,k_\ell}(\cdot),\dots,B_{n,k_\ell}(\cdot) \right) = P^{m_1,\dots,m_n}_{\lambda_1,\dots,\lambda_n} \left(B_1(\cdot),\dots,B_n(\cdot) \right).$$

We have therefore shown that, for our *n*-tuples $(B_{1,k}(\cdot), \ldots, B_{n,k}(\cdot)), (B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1,\ldots,M_n}$,

$$\lim_{\ell \to \infty} g\left(\lambda_{1,k_{\ell}}, \dots, \lambda_{n,k_{\ell}}; B_{1,k_{\ell}}(\cdot), \dots, B_{n,k_{\ell}}(\cdot); f; \cdot\right)$$

= $g\left(\lambda_{1}, \dots, \lambda_{n}; B_{1}(\cdot), \dots, B_{n}(\cdot); f; \cdot\right)$

uniformly on compact subsets of B_0 . But the limit is only taken in monomials and is therefore independent of w_1, \ldots, w_n . Consequently this limit is uniform on all of B_0 . However, in view of (3.68) with k replacing k_{ℓ} , it follows from Proposition 3.15 that

$$\lim_{k \to \infty} P^{m_1, \dots, m_n}_{\lambda_{1,k}, \dots, \lambda_{n,k}} \left(B_{1,k}(\cdot), \dots, B_{n,k}(\cdot) \right)$$
$$= P^{m_1, \dots, m_n}_{\lambda_1, \dots, \lambda_n} \left(B_1(\cdot), \dots, B_n(\cdot) \right)$$

uniformly in X-norm on B_0 .

Now fix for the moment $k \in \mathbb{N}$. By definition of the disentangling map we may write

$$g(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; w_1, \dots, w_n) = \sum_{m_1,\dots,m_n=0}^{\infty} a_{m_1,\dots,m_n} w_1^{m_1} \cdots w_n^{m_n} P_{\lambda_{1,k},\dots,\lambda_{n,k}}^{m_1,\dots,m_n} (B_{1,k}(\cdot), \dots, B_{n,k}(\cdot))$$

From the Lebesgue dominated convergence theorem (for Bochner integrals) it follows that

$$\lim_{\substack{(w_1,\dots,w_n)\to(1,\dots,1)\\(w_1,\dots,w_n)\in B_0}} g(\lambda_{1,k},\dots,\lambda_{n,k};B_{1,k}(\cdot),\dots,B_{n,k}(\cdot);f;w_1,\dots,w_n)$$

$$=\sum_{\substack{m_1,\dots,m_n=0\\m_1,\dots,m_n=0}}^{\infty} a_{m_1,\dots,m_n} P_{\lambda_{1,k},\dots,\lambda_{n,k}}^{m_1,\dots,m_n} (B_{1,k}(\cdot),\dots,B_{n,k}(\cdot))$$

$$=g(\lambda_{1,k},\dots,\lambda_{n,k};B_{1,k}(\cdot),\dots,B_{n,k}(\cdot);f;1,\dots,1).$$
(3.83)

This limit is clearly uniform in $k \in \mathbb{N}$. Also, as we have shown,

$$\lim_{k \to \infty} g\left(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; w_1, \dots, w_n\right)$$

= $g\left(\lambda_1, \dots, \lambda_n; B_1(\cdot), \dots, B_n(\cdot); f; w_1, \dots, w_n\right)$

for each $(w_1, \ldots, w_n) \in B_0$. As we have the limit of the g's with respect to the (w_1, \ldots, w_n) being uniform in $k \in \mathbb{N}$ and the limit on $k \in \mathbb{N}$ existing for each (w_1, \ldots, w_n) , we wish to apply Moore's theorem on iterated limits [4, p. 128]. To do this, we select arbitrary sequences $\{\alpha_{j,p}\}_{p=1}^{\infty}, j = 1, \ldots, n$, in B(0, 1) for which $\alpha_{j,p} \to 1$ as $p \to \infty$. Then, because we have

$$g(\lambda_{1,k},\ldots,\lambda_{n,k};B_{1,k}(\cdot),\ldots,B_{n,k}(\cdot);f;\alpha_{1,p},\ldots,\alpha_{n,p})$$
$$\xrightarrow{p\to\infty}g(\lambda_{1,k},\ldots,\lambda_{n,k};B_{1,k}(\cdot),\ldots,B_{n,k}(\cdot);f;1,\ldots,1)$$

uniformly in $k \in \mathbb{N}$ and also, because we have

$$g(\lambda_{1,k},\ldots,\lambda_{n,k};B_{1,k}(\cdot),\ldots,B_{n,k}(\cdot);f;\alpha_{1,p},\ldots,\alpha_{n,p})$$
$$\xrightarrow[k\to\infty]{}g(\lambda_1,\ldots,\lambda_n;B_1(\cdot),\ldots,B_n(\cdot);f;\alpha_{1,p},\ldots,\alpha_{n,p})$$

for each $p \in \mathbb{N}$, we may apply Moore's theorem to show that

$$\lim_{\substack{k \to \infty \\ p \to \infty}} g\left(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; \alpha_{1,p}, \dots, \alpha_{n,p}\right)$$

$$= \lim_{\substack{k \to \infty}} \lim_{p \to \infty} g\left(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; \alpha_{1,p}, \dots, \alpha_{n,p}\right)$$

$$= \lim_{\substack{p \to \infty}} \lim_{k \to \infty} g\left(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; \alpha_{1,p}, \dots, \alpha_{n,p}\right) \quad (3.84)$$

$$= g\left(\lambda_{1}, \dots, \lambda_{n}; B_{1}(\cdot), \dots, B_{n}(\cdot); f; 1, \dots, 1\right)$$

Because the sequences $\{\alpha_{j,p}\}_{p=1}^{\infty}$, j = 1, ..., n, from B(0,1) are arbitrary, we can conclude that

$$\lim_{\substack{k \to \infty \\ (w_1, \dots, w_n) \to (1, \dots, 1) \\ (w_1, \dots, w_n) \to (B_0)}} g(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; w_1, \dots, w_n)$$

$$= \lim_{\substack{k \to \infty \\ (w_1, \dots, w_n) \to (1, \dots, 1) \\ (w_1, \dots, w_n) \in B_0}} \lim_{\substack{g (\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; w_1, \dots, w_n) \\ (w_1, \dots, w_n) \in B_0}} g(\lambda_{1,k}, \dots, \lambda_{n,k}; B_{1,k}(\cdot), \dots, B_{n,k}(\cdot); f; w_1, \dots, w_n)$$

$$= g(\lambda_1, \dots, \lambda_n; B_1(\cdot), \dots, B_n(\cdot); f; 1, \dots, 1).$$
(3.85)

If we translate to the standard notation, (3.85) can be characterized as

$$\lim_{k \to \infty} f_{\lambda_{1,k},\dots,\lambda_{n,k}} \left(B_{1,k}(\cdot),\dots,B_{n,k}(\cdot) \right) = f_{\lambda_{1},\dots,\lambda_{n}} \left(B_{1}(\cdot),\dots,B_{n}(\cdot) \right) \quad (3.86)$$

in norm on X.

Consequently, we have proven the following stability result using the notation introduced here.

Theorem 3.16. Let $(B_1(\cdot), \ldots, B_n(\cdot)) \in \mathcal{L}_{M_1, \ldots, M_n}$ and associate to each $B_j(\cdot), j = 1, \ldots, n, a$ Borel probability measure λ_j on [0, T]. (If λ_j has a discrete part, we assume that it is finitely supported.) Select sequences $\{\lambda_{j,k}\}_{k=1}^{\infty}, j = 1, \ldots, n, \text{ of Borel probability measures on } [0, T]$ (with finitely supported discrete parts, if a discrete part is present) such that $\lambda_{j,k} \rightharpoonup \lambda_j$ as $k \rightarrow \infty$. We also select continuous X-valued functions $B_j(\cdot)$ on $[0, T], j = 1, \ldots, n, and$ sequences $\{B_{j,k}(\cdot)\}_{k=1}^{\infty}, j = 1, \ldots, n, of$ continuous X-valued functions X-valued functions such that $B_{j,k}(s) \rightarrow B_j(s)$ pointwise on [0, T]. It is assumed that the n-tuples $(B_1(\cdot), \ldots, B_n(\cdot))$ and $(B_{1,k}(\cdot), \ldots, B_{n,k}(\cdot)), k \in \mathbb{N}$, are elements of $\mathcal{L}_{M_1,\ldots,M_n}$. Finally, let $f \in \mathbb{A}(M_1,\ldots,M_n)$. Then, in norm on X,

$$\lim_{k \to \infty} f_{\lambda_{1,k},\dots,\lambda_{n,k}} \left(B_{1,k}(\cdot),\dots,B_{n,k}(\cdot) \right) = f_{\lambda_{1},\dots,\lambda_{n}} \left(B_{1}(\cdot),\dots,B_{n}(\cdot) \right).$$
(3.87)

4. FOC for unbounded operators using the Taylor calculus

In this section we address our first approach to using unbounded operators in Feynman's operational calculus. Here, we will use the results of the paper [34], the Taylor calculus and discussed briefly above in Subsection 2.4. We begin by letting A_j , j = 1, ..., n, be closed densely defined unbounded operators on the Banach space X. Fix, for the moment, $j \in \{1, ..., n\}$. Following the paper [34], we let $\mathscr{G}(A_j)$ be the family of all analytic functions f which satisfy (1) $\sigma(A_j) \subseteq \Delta(f)$

and

(2) $\Delta(f)$ contains a neighborhood of infinity and f is regular at infinity. We will denote

$$\lim_{|\xi|\to\infty} f(\xi) =: f(\infty).$$

Furthermore, we will assume that the resolvent set, $\rho(A_j)$, of A_j , is nonempty.

Using Theorem 4.1 of [34], there is an unbounded Cauchy domain D_j (see Subsection 2.4 above) such that $\sigma(A_j) \subseteq D_j \subseteq \overline{D}_j \subseteq \Delta(f)$. The (Bochner) integral

$$\frac{1}{2\pi i} \int_{\partial D_j} f(\xi) R(\xi; A_j) \, d\xi$$

defines a bounded linear operator which is the same for any choice of D_j satisfying the stated conditions.

For $f \in \mathscr{G}(A_j)$, we write, as in Definition 2.27,

$$f(A_j) := f(\infty)I + \frac{1}{2\pi i} \int_{\partial D_j} f(\xi) R(\xi; A_j) \, d\xi.$$
 (4.1)

We fix a $f_j \in \mathscr{G}(A_j)$ and so fix, using (4.1), a bounded linear operator $f_j(A_j)$. Doing this for each j = 1, ..., n, we obtain bounded linear operators $f_1(A_1), ..., f_n(A_n)$ on X. Note that, for each j = 1, ..., n, we have

$$\|f_j(A_j)\|_{\mathcal{L}(X)} \le |f_j(\infty)| + \frac{1}{2\pi} \int_{\partial D_j} |f_j(\xi)| \, \|R(\xi; A_j)\|_{\mathcal{L}(X)} \, d\xi.$$
(4.2)

As noted above in Subsection 2.3, the resolvent operator $R(\xi; A_j)$ is, for each $j \in \{1, \ldots, n\}$ and each $\xi \in D_j \subseteq \rho(A_j)$, a bounded linear operator on X and is analytic on $\rho(A_j)$. Furthermore, D_j , $j \in \{1, \ldots, n\}$, is a Cauchy domain (following [34] and Subsection 2.4) and ∂D_j consists of finitely many rectifiable curves. So, $||R(\xi; A_j)||_{\mathcal{L}(X)}$ is such that

$$\sup\left\{\|R(\xi;A_j)\|_{\mathcal{L}(X)}:\xi\in\partial D_j\right\}=:M_j<\infty\tag{4.3}$$

for each $j \in \{1, \ldots, n\}$. It follows that, for each $j \in \{1, \ldots, n\}$,

$$||f_j(A_j)||_{\mathcal{L}(X)} \le |f_j(\infty)| + \frac{M_j}{2\pi} \int_{\partial D_j} |f_j(\xi)| \, d\xi.$$
 (4.4)

As in Subsection 2.1, we construct the commutative Banach algebra $\mathbb{A}(R_1, \ldots, R_n)$ with radii

$$R_{j} := |f_{j}(\infty)| + \frac{M_{j}}{2\pi} \int_{\partial D_{j}} |f_{j}(\xi)| \, d\xi.$$
(4.5)

4.1. Using the Taylor calculus in Feynman's operational calculus. We are now ready to develop the operational calculus for this setting. Note that, in our setting, using the closed unbounded operators A_1, \ldots, A_n and the functions f_1, \ldots, f_n , we obtain the fixed (i.e., *time independent*) operators $f_1(A_1), \ldots, f_n(A_n)$. This immediately puts us in the setting of the time independent operational calculus. To continue, we associate Borel probability measures $\lambda_1, \ldots, \lambda_n$ on [0, T] to $f_1(A_1), \ldots, f_n(A_n)$, respectively. (They are, of course, the *time-ordering measures*.) Just above, we constructed the commutative Banach algebra $\mathbb{D}(f_1(A_1)^{\sim}, \ldots, f_n(A_n)^{\sim})$ as in Subsection 2.1. Given an element $g \in \mathbb{D}(f_1(A_1)^{\sim}, \ldots, f_n(A_n)^{\sim})$, we write g as a Taylor series centered at $(0, \ldots, 0)$; i.e.,

$$g(f_1(A_1)^{\sim},\ldots,f_n(A_n)^{\sim}) = \sum_{m_1,\ldots,m_n=0}^{\infty} a_{m_1,\ldots,m_n}^g [f_1(A_1)^{\sim}]^{m_1} \cdots [f_n(A_n)^{\sim}]^{m_n}$$

The disentangling map from $\mathbb{D}(f_1(A_1)^{\sim}, \ldots, f_n(A_n)^{\sim})$ to $\mathcal{L}(X)$ is defined by

$$\mathcal{T}_{\lambda_{1},...,\lambda_{n}}g\left(f_{1}(A_{1})^{\sim},\ldots,f_{n}(A_{n})^{\sim}\right)$$

$$:= g_{\lambda_{1},...,\lambda_{n}}\left(f_{1}(A_{1}),\ldots,f_{n}(A_{n})\right)$$

$$= \sum_{m_{1},...,m_{n}=0}^{\infty} a_{m_{1},...,m_{n}}^{g} P_{\lambda_{1},...,\lambda_{n}}^{m_{1},...,m_{n}}\left(f_{1}(A_{1}),\ldots,f_{n}(A_{n})\right),$$
(4.6)

exactly as in Subsection 2.2. Consequently, the operational calculus defined in terms of the Taylor calculus has all of the same properties as the operational calculus developed in [15] and outlined in Subsections 2.1 and 2.2. It is the choice of radii in (4.5) which allows us to show that the disentangling map is a contraction (see Theorem 2.6, above.) The particular form of the disentangled operator depends on whether or not the time-ordering measures have nonzero (finitely supported) discrete parts. See Equations (2.5) and (2.17) above for the form that the disentangled monomial takes depending on whether or not the time-ordering measures have a non-zero discrete part. Though it seems anticlimatic, applying the Taylor calculus in this way allows us to use the operational calculus in ways that the abstract approach of [15] does not permit. So, while the operational calculus in the setting of the Taylor calculus seems to give nothing new, this approach does indeed provide a useful extension of the operational calculus to the incorporation of unbounded operators into the abstract approach to Feynman's operational calculus.

The reader will note that using Taylor's calculus allows us to "hide" the unbounded operators A_1, \ldots, A_n within the functions f_1, \ldots, f_n and use the resulting bounded operators $f_1(A_1), \ldots, f_n(A_n)$ in the operational calculus in the usual way (see Subsection 2.1) to obtain the disentangled operator. However, the approach using Taylor's calculus allows us to see a connection

between the abstract approach to the operational calculus seen in [15] and the "modified Feynman integral" of Lapidus ([20], [14, Chapter 11]). Indeed, this connection is the subject of the next subsection. It is worth noting that this example can not be addressed in the setting of [15]; it is the Taylor calculus which allows the example of the next subsection to be addressed.

4.2. Example: using the Taylor calculus in FOC to obtain the modified Feynman integral. In this subsection we apply the formalism developed above using the Taylor calculus to the so-called Modified Feynman Integral of Lapidus (see [20], [14]). To begin, we outline the essential setup for the "product formula for imaginary resolvents" (see Section 11.3 of [14]) and, in particular, Theorem 11.3.1). Suppose that A and B are unbounded self-adjoint operators on the Hilbert space \mathcal{H} with A nonnegative. Let B_+ and B_- be the positive and negative parts of B (see Subsection 2.3 above) defined by via the spectral theorem; then B_+ and B_- are nonnegative self-adjoint operators and $B = B_+ - B_-$. We assume that B_- is relatively form bounded with respect to A with relative bound smaller than 1; i.e.,

$$Q(A) \subseteq Q(B_{-})$$

and there are constants $\gamma, \delta > 0$ with $\gamma < 1$ such that

$$\|B_{-}^{1/2}\phi\|_{\mathcal{H}}^{2} \leq \gamma \|A^{1/2}\phi\|_{\mathcal{H}} + \delta \|\phi\|_{\mathcal{H}}^{2}$$

for all $\phi \in Q(A)$. We assume, for simplicity, that $Q(A) \cap Q(B_+)$ is dense in \mathcal{H} . For all $\psi \in \mathcal{H}$,

$$\lim_{k \to \infty} \left(\left[I + \frac{it}{k} A \right]^{-1} \left[I + \frac{it}{k} B \right]^{-1} \right)^k \psi = e^{-it(A + B)} \psi \tag{4.7}$$

and this limit is uniform in t on all bounded subsets of \mathbb{R} . This is the "Product Formula for Imaginary Resolvents," Theorem 11.3.1 of [14]. (See also [20].)

We now put this product formula into the setting of Feynman's operational calculus using the Taylor calculus. We take, for $n \in \mathbb{N}$ and for $t \ge 0$,

$$f_k(z) := \left(1 + \frac{it}{k}z\right)^{-1}$$

We note that $f_k(z)$ is regular at infinity and $f_k(\infty) = 0$ for all $k \in \mathbb{N}$. Also, $f_k(z)$ has a simple pole at $z = \frac{ik}{t}$ on the imaginary axis. We further observe that $\sigma(A), \sigma(B) \subseteq \Delta(f_k)$ ($\Delta(g)$ is the domain of g) and $\Delta(f_k)$ contains neighborhoods of infinity. Referring to Subsection 2.4, we see that $f_k \in \mathscr{G}(A)$ and $f_k \in \mathscr{G}(B)$ for all $k \in \mathbb{N}$. Hence, we obtain the bounded linear operators $f_1(A)$ and $f_2(B)$ on \mathcal{H} . We can therefore construct the commutative Banach algebras

$$\mathbb{A}\left(\|f_1(A)\|_{\mathcal{L}(\mathcal{H})}, \|f_2(B)\|_{\mathcal{L}(\mathcal{H})}\right) \text{ and } \mathbb{D}\left(f_1(A)^{\sim}, f_2(B)^{\sim}\right).$$

In order to investigate the modified Feynman integral in the setting of Feynman's operational calculus using the Taylor calculus, we proceed as follows. First, we construct a family of disentangling algebras $\mathbb{D}_1, \mathbb{D}_2, \ldots$. Indeed, we define

$$\mathbb{D}_{1} := \mathbb{D} \left(f_{1}(B)^{\sim}, f_{1}(A)^{\sim} \right), \\
\mathbb{D}_{2} := \mathbb{D} \left(f_{2}(B)^{\sim}, f_{2}(B)^{\sim}, f_{2}(A)^{\sim}, f_{2}(A)^{\sim} \right), \\
\mathbb{D}_{3} := \mathbb{D} \left(f_{3}(B)^{\sim}, f_{3}(B)^{\sim}, f_{3}(B)^{\sim}, f_{3}(A)^{\sim}, f_{3}(A)^{\sim}, f_{3}(A)^{\sim} \right), \qquad (4.8)$$

$$\vdots \\
\mathbb{D}_{k} := \mathbb{D} \left(\underbrace{f_{k}(B)^{\sim}, f_{k}(B)^{\sim}, \dots, f_{k}(B)^{\sim}, f_{k}(A)^{\sim}, f_{k}(A)^{\sim}, \dots, f_{k}(A)^{\sim}}_{f_{k}(A)^{\sim}, \dots, f_{k}(A)^{\sim}} \right) \\
\vdots$$

For each $k \in \mathbb{N}$, the norm on \mathbb{D}_k is

$$\|g(f_{k}(B)^{\sim}, \dots, f_{k}(B)^{\sim}, f_{k}(A)^{\sim}, \dots, f_{k}(A)^{\sim})\|_{\mathbb{D}_{k}}$$

$$= \sum_{m_{1},\dots,m_{k},m_{k+1},\dots,m_{2k}=0}^{\infty} |a_{m_{1},\dots,m_{2k}}| \|f_{k}(B)\|_{\mathcal{L}(\mathcal{H})}^{m_{1}+\dots+m_{k}} \|f_{k}(A)\|_{\mathcal{L}(\mathcal{H})}^{m_{k+1}+\dots+m_{2k}}$$

$$(4.9)$$

where we have used the Taylor series for $g \in \mathbb{D}_k$ in the usual way (see (2.5)). With the family $\{\mathbb{D}_k\}_{k=1}^{\infty}$ in hand, we construct the direct sum Banach algebra (see [18], for instance)

$$\mathbb{D}_{\oplus} := \bigoplus_{k=1}^{\infty} \mathbb{D}_k \tag{4.10}$$

with norm

$$\|\{g_k\}_{k=1}^{\infty}\|_{\oplus} := \sup_{k \in \mathbb{N}} \|g_k\|_{\mathbb{D}_k}.$$
(4.11)

The reader will note that we have not yet associated any time-ordering measures to our operators. (There is no time-dependence in our operators, so we do not need the measures to define the disentangling algebras; see [15, Chapter 2].) Fix a $k \in \mathbb{N}$. Choose real numbers $a_1, b_1, c_1, d_1, \ldots, a_k, b_k, c_k, d_k$ which satisfy

$$0 < a_1 < b_1 < c_1 < d_1 < a_2 < b_2 < c_2 < d_2 < \dots < a_k < b_k < c_k < d_k < T.$$

Next, choose continuous Borel probability measures μ_j on $[a_j, b_j]$, $j = 1, \ldots, k$ and choose continuous Borel probability measures ν_j on $[c_j, d_j]$ for $j = 1, \ldots, k$. (To be explicit, we assume that the support of each μ_j , $S(\mu_j)$, satisfies $S(\mu_j) \subseteq [a_j, b_j]$ and that the support of each ν_j , $S(\nu_j)$, satisfies $S(\nu_j) \subseteq [c_j, d_j]$ for $j = 1, \ldots, n$.) We associate the measures μ_j , $j = 1, \ldots, k$, to the k operators $f_k(B)$ and associate the measures ν_j , $j = 1, \ldots, n$, to the

k operators $f_k(A)$. It is clear that we can carry out this construction for every $k \in \mathbb{N}$.

With the time-ordering measures in hand, we now define the disentangling map on \mathbb{D}_{\oplus} . First, recall that, for each $k \in \mathbb{N}$, we have the disentangling map

$$\mathcal{T}_{\mu_1,\dots,\mu_k,\nu_1,\dots,\nu_k}:\mathbb{D}_k\longrightarrow\mathcal{L}(\mathcal{H})$$

exactly as in Subsection 2.2. We use these disentangling maps to define

$$\mathcal{T}_{\oplus}: \mathbb{D}_{\oplus} \longrightarrow \bigoplus_{k=1}^{\infty} \mathcal{L}(\mathcal{H})$$
 (4.12)

by

$$\mathcal{T}_{\oplus} \left[\{g_k\}_{k=1}^{\infty} \right] := \{\mathcal{T}_{\mu_1,\dots,\mu_k,\nu_1,\dots,\nu_k} g_k\}_{k=1}^{\infty} .$$
(4.13)

It is clear that \mathcal{T}_\oplus is linear and, since

$$\begin{aligned} \|\mathcal{T}_{\oplus} \left[\{g_k\}_{k=1}^{\infty} \right] \| &= \sup_{k \in \mathbb{N}} \|\mathcal{T}_{\mu_1, \dots, \mu_k, \nu_1, \dots, \nu_k} g_k \|_{\mathcal{L}(\mathcal{H})} \\ &\leq \sup_{k \in \mathbb{N}} \|g_k\|_{\mathbb{D}_k} \\ &= \|\{g_k\}_{k=1}^{\infty}\|_{\oplus}, \end{aligned}$$

$$(4.14)$$

 \mathcal{T}_{\oplus} is a linear contraction. (Compare to [15], Chapters 2, 6 and 8.)

We are interested in the elements of \mathbb{D}_k of the form

$$P^{1,\dots,1,1,\dots,1}\left(\overbrace{f_k(B)^{\sim},\dots,f_k(B)^{\sim}}^{k \text{ terms}},\overbrace{f_k(A)^{\sim},\dots,f_k(A)^{\sim}}^{k \text{ terms}}\right).$$
(4.15)

The corresponding element of \mathbb{D}_{\oplus} is

$$\Theta := \left\{ P^{1,\dots,1,1,\dots,1} \left(\overbrace{f_k(B)^{\sim},\dots,f_k(B)^{\sim}}^{k \text{ terms}}, \overbrace{f_k(A)^{\sim},\dots,f_k(A)^{\sim}}^{k \text{ terms}} \right) \right\}_{\substack{k=1\\(4.16)}}^{\infty}$$

and so

$$\mathcal{T}_{\oplus}(\Theta) = \left\{ P_{\mu_1,\dots,\mu_k,\nu_1,\dots,\nu_k}^{1,\dots,1,1,\dots,1} \left(\overbrace{f_k(B),\dots,f_k(B)}^{k \text{ terms}}, \overbrace{f_k(A),\dots,f_k(A)}^{k \text{ terms}} \right) \right\}_{\substack{k=1\\(4.17)}}^{\infty}.$$

Now, because of the way that the supports $S(\mu_j)$ and $S(\nu_j)$ are ordered, we can apply Corollary 3.3.3 of [15] to deduce that

$$P^{1,\dots,1,1,\dots,1}_{\mu_1,\dots,\mu_k,\nu_1,\dots,\nu_k}\left(\overbrace{f_k(B),\dots,f_k(B)}^{k \text{ terms}},\overbrace{f_k(A),\dots,f_k(A)}^{k \text{ terms}}\right) = [f_k(B)f_k(A)]^k$$
(4.18)

$$=\left[\left(I+\frac{it}{k}B\right)^{-1}\left(I+\frac{it}{k}A\right)^{-1}\right]$$

e

We can now obtain the modified Feynman integral. Let $V : \mathbb{R}^d \to \mathbb{R}$ be Lebesgue measurable. The modified Feynman integral associated with the potential V, denoted by $\mathscr{F}_M^t(V)$, is defined as the strong operator limit

$$\mathscr{F}_{M}^{t}(V) = \lim_{k \to \infty} \left(\left[I + \frac{it}{k} H_0 \right]^{-1} \left[I + \frac{it}{k} V \right]^{-1} \right)^{k}$$
(4.19)

k

in $\mathcal{L}(L^2(\mathbb{R}^d))$ when this limit exists. (See [14, Definition 11.4.4].)

Following [14, Theorem 11.4.2 and Corollary 11.4.5], we assume that $V : \mathbb{R}^d \to \mathbb{R}$ is Lebesgue measurable with $V_+ \in L^1_{loc}(\mathbb{R}^d)$ and assume that V_- is relatively form bounded with respect to H_0 with relative bound less than 1. With our function f_k , we note that, since the spectrum of H_0 is $[0, \infty)$, $f_k \in \mathscr{G}(H_0)$ and because V is real-valued (though possibly unbounded), $f_k \in \mathscr{G}(V)$ (see Subsection 2.4) for all $k \in \mathbb{N}$ and $t \ge 0$. Then $\mathscr{F}^t_M(V)$ exists for all $t \in \mathbb{R}$ and

$$\mathcal{F}_{M}^{t}(V) = e^{-it(H_{0}+V)}$$

$$= \lim_{k \to \infty} P^{1,\dots,1,1,\dots,1}_{\mu_{1},\dots,\mu_{k},\nu_{1},\dots,\nu_{k}} \left(\underbrace{f_{k}(H_{0}),\dots,f_{k}(H_{0})}_{k \text{ terms}}, \underbrace{f_{k}(V),\dots,f_{k}(V)}_{k \text{ terms}} \right)$$

$$(4.20)$$

where the (strong operator) limit is uniform in t on all bounded subsets of \mathbb{R} . We note that the hypothesis on V allows highly singular potentials – see Remark 11.4.6 of [14] and Example 11.4.7 of [14]. While it would be advantageous to be able to address this limit using an aspect of the stability theory for Feynman's operational calculus, the current state of the art of the stability theory does not allow this to be done.

4.3. Stability with respect to the functions f_j . In this subsection, we will investigate the stability (with respect to the functions f_j) of the operational calculus for unbounded operators obtained above using A. E. Taylor's ideas. As remarked on in Subsection 3.3, there are two varieties of stability to consider, with respect to the time-ordering measures and with respect to the operators (or operator-valued functions). (Actually, there is a third type of stability, joint stability, which is the stability of the operational calculus with respect to both the operators (operator-valued functions) and the measures. See Chapter 7 of [15].) Here, we will not consider the stability with respect to the time-ordering measures, as the paper [28] addresses this aspect of the stability theory for this setting.

By considering the stability of the operational calculus with respect to the functions f_j , we are essentially looking at the stability of the operational calculus with respect to the operators (since the operators change with the choice of the function from $\mathscr{G}(T)$.)

We will begin our discussion by considering an arbitrary unbounded, closed and densely defined operator T on X. (The ideas we use throughout this subsection are introduced and discussed in Subsection 2.4.) We next choose a $f \in \mathscr{G}(T)$. We obtain a corresponding Cauchy domain D_f with $\sigma(T) \subseteq D_f \subseteq \overline{D_f} \subseteq \Delta(f)$ with $\Delta(f)$ open in \mathbb{C} . (See Subsection 2.4.) Moreover, the finitely many components of D_f are bounded by finitely many rectifiable Jordan curves, no two of which intersect. (The closures of the components of D_f are disjoint, by definition; see Subsection 2.4.) Finally, D_f has at most one unbounded component and the number of components is fixed by the number of components of $\sigma(T)$.

Definition 4.1. Given $f \in \mathscr{G}(T)$ and its associated Cauchy domain D_f as in the previous paragraph, we say that a sequence $\{f_k\}_{k=1}^{\infty}$ from $\mathscr{G}(T)$ converges to $f \in \mathscr{G}(T)$ if

(1)
$$\Delta(f_k) \supseteq \Delta(f)$$
, for all $k \in \mathbb{N}$ (4.21)

and

(2) $\{f_k\}_{k=1}^{\infty}$ converges uniformly on compact subsets of $\Delta(f)$. (4.22)

Choose such a sequence $\{f_k\}_{k=1}^{\infty}$ from $\mathscr{G}(T)$. For each $k \in \mathbb{N}$, f_k has an associated Cauchy domain D_{f_k} . From the definition of Cauchy domain (see Subsection 2.4, above), each Jordan curve bounding D_{f_k} is homotopic to the corresponding Jordan curve of D_f . Consequently, for each $k \in \mathbb{N}$,

$$\frac{1}{2\pi i} \int_{\partial D_{f_k}} f_k(\xi) R(\xi; T) \, d\xi = \frac{1}{2\pi i} \int_{\partial D_f} f_k(\xi) R(\xi; T) \, d\xi.$$
(4.23)

(Each integral is, in fact, a finite sum of integrals over rectifiable Jordan arcs.) Now, denote by C_1^f, \ldots, C_N^f the components of D_f and assume, without loss of generality, that C_1^f is the unbounded component. For each $k \in \mathbb{N}$, we may then write

$$\frac{1}{2\pi i} \int_{\partial D_f} f_k(\xi) R(\xi;T) \, d\xi = \sum_{j=1}^N \frac{1}{2\pi i} \int_{\partial C_j^f} f_k(\xi) R(\xi;T) \, d\xi. \tag{4.24}$$

Now, for each $j \in \{1, \ldots, N\}$,

$$\left\| \frac{1}{2\pi i} \int_{\partial C_j^f} f(\xi) R(\xi; T) \, d\xi - \frac{1}{2\pi i} \int_{\partial C_j^f} f_k(\xi) R(\xi; T) \, d\xi \right\|_{\mathcal{L}(X)}$$

$$\leq \frac{1}{2\pi} \int_{\partial C_j^f} |f(\xi) - f_k(\xi)| \, \|R(\xi; T)\|_{\mathcal{L}(X)} d\xi \qquad (4.25)$$

$$\leq \frac{1}{2\pi} \sup \left\{ |f(\xi) - f_k(\xi)| : \xi \in \partial C_j^f \right\} \cdot \sup \left\{ \|R(\xi; T)\|_{\mathcal{L}(X)} : \xi \in \partial C_j^f \right\} \ell \left(\partial C_j^f\right),$$

where $\ell(\Sigma)$ denotes the length of the rectifiable curve Σ . Next, since ∂C_j^f is a Jordan arc, it is compact for each j = 1, ..., n. So

$$\sup\left\{\left|f_k(\xi) - f(\xi)\right| : \xi \in C_j^f\right\} \xrightarrow[k \to \infty]{} 0.$$

We also know that the supremum of $||R(\xi;T)||_{\mathcal{L}(X)}$ is bounded on ∂C_j^f . Therefore, because the sum in (4.24) is finite,

$$\lim_{k \to \infty} \left\| \frac{1}{2\pi i} \int_{\partial D_f} f(\xi) R(\xi; T) \, d\xi - \frac{1}{2\pi i} \int_{\partial D_f} f_k(\xi) R(\xi; T) \, d\xi \right\|_{\mathcal{L}(X)} = 0.$$
(4.26)

We have therefore determined that the sequence

$$\left\{\frac{1}{2\pi i} \int_{\partial D_f} f_k(\xi) R(\xi; T) \, d\xi\right\}_{k=1}^{\infty} \tag{4.27}$$

of bounded linear operators converges in norm to the bounded linear operator

$$\frac{1}{2\pi i} \int_{\partial D_f} f(\xi) R(\xi; T) \, d\xi. \tag{4.28}$$

We now turn our attention back to the operators A_1, \ldots, A_n and the functions f_1, \ldots, f_n chosen from $\mathscr{G}(A_1), \ldots, \mathscr{G}(A_n)$, respectively. As we know, these functions determine bounded linear operators $f_1(A_1), \ldots, f_n(A_n)$. Associate the continuous Borel probability measure λ_j to $f_j(A_j)$ for each $j = 1, \ldots, n$. (We will use continuous time-ordering measures here mostly for convenience. The same stability result obtained at the end of this section can be stated and proved for the case where our time-ordering measures have a finitely supported discrete part, although combinatorial issues greatly complicate the arguments.) Fix, momentarily, $j \in \{1, \ldots, n\}$ and select a sequence $\{f_{j,k}\}_{k=1}^{\infty}$ from $\mathscr{G}(A_j)$ converging to f_j in the sense defined above. We have a corresponding family $\{D_{f_{j,k}}\}_{k=1}^{\infty}, j = 1, \ldots, n$, of Cauchy domains. We have observed above that

$$\frac{1}{2\pi i} \int_{\partial D_{f_{j,k}}} f_{j,k}(\xi) R(\xi; A_j) \, d\xi = \frac{1}{2\pi i} \int_{\partial D_{f_j}} f_{j,k}(\xi) R(\xi; A_j) \, d\xi.$$
(4.29)

It follows that

$$|f_{j,k}(A_j)||_{\mathcal{L}(X)} \leq |f_{j,k}(\infty)| + \frac{1}{2\pi} \int_{\partial D_{f_j}} |f_{j,k}(\xi)| \, ||R(\xi; A_j)||_{\mathcal{L}(X)} \, d\xi$$

$$\leq |f_{j,k}(\infty)| + \frac{M_j}{2\pi} \int_{\partial D_{f_j}} |f_{j,k}(\xi)| \, d\xi,$$
 (4.30)

where

$$M_j := \sup \left\{ \|R(\xi; A_j)\|_{\mathcal{L}(X)} : \xi \in \partial D_{f_j} \right\};$$

see (4.3) above. Based on the calculation just above, for each $k \in \mathbb{N}$ we construct the commutative Banach algebra

$$\mathbb{A}_k := \mathbb{A}\left(R_{1,k}, \dots, R_{n,k}\right),\tag{4.31}$$

where

$$R_{j,k} := |f_{j,k}(\infty)| + \frac{M_j}{2\pi} \int_{\partial D_f} |f_{j,k}(\xi)| \, d\xi, \qquad (4.32)$$

for $j = 1, \ldots, n$. We also define

$$\mathbb{A}_0 := \mathbb{A} \left(R_1, \dots, R_n \right), \tag{4.33}$$

where, as above,

$$R_{j} = |f_{j}(\infty)| + \frac{M_{j}}{2\pi} \int_{\partial D_{f_{j}}} |f_{j}(\xi)| \, d\xi, \qquad (4.34)$$

for $j = 1, \ldots, n$. We now define

$$\mathbb{A}_{\oplus} := \bigoplus_{k=0}^{\infty} \mathbb{A}_k. \tag{4.35}$$

Then \mathbb{A}_{\oplus} is a commutative Banach algebra with norm

$$\|\{g_k\}_{k=0}^{\infty}\|_{\oplus} := \sup_{k \in \mathbb{N} \cup \{0\}} \|g_k\|_{\mathbb{A}_k}.$$
(4.36)

For each $k \in \mathbb{N}$ we take $\pi_k : \mathbb{A}_{\oplus} \longrightarrow \mathbb{A}_k$ to be the canonical projection. Given $\theta := \{g_k\}_{k=0}^{\infty} \in \mathbb{A}_{\oplus}$, the disentangling map

$$\mathcal{T}^{\mathbb{A}_{\oplus}}_{\lambda_{1},\dots,\lambda_{n}}:\mathbb{A}_{\oplus}\longrightarrow \bigoplus_{k=0}^{\infty}\mathcal{L}(X)$$

is defined by

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n}^{\mathbb{A}_{\oplus}} \theta := \left\{ (g_k)_{\lambda_1,\dots,\lambda_n} \left(f_{1,k}(A_1),\dots,f_{n,k}(A_n) \right) \right\}_{k=0}^{\infty}.$$
(4.37)

Remark 4.2. The reader may have noticed that we are working here with the algebras \mathbb{A}_k of analytic functions. This is for convenience; the algebras \mathbb{A}_k and the disentangling algebras \mathbb{D}_k (if they were constructed here) are, as noted in Subsection 2.2, isometrically isomorphic.

It is straightforward to show that $\mathcal{T}_{\lambda_1,\ldots,\lambda_n}^{\mathbb{A}_{\oplus}}$ is a bounded linear operator (see [15, Chapters 7, 8], [24], [28], [29]).

Now, let $\theta := \{g_k\}_{k=0}^{\infty} \in \mathbb{A}_{\oplus}$ be such that $g_k \equiv g$ for all $k \in \mathbb{N} \cup \{0\}$. We consider

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n}^{\mathbb{A}_{\oplus}}\left(\{g_k\}_{k=0}^{\infty}\right) = \{\mathcal{T}_{\lambda_1,\dots,\lambda_n}g\left(f_{1,k}\left(A_1\right)^{\sim},\dots,f_{n,k}\left(A_n\right)^{\sim}\right)\}_{k=0}^{\infty} \quad (4.38)$$

and claim that

 $\|g_{\lambda_{1},\dots,\lambda_{n}}\left(f_{1,k}\left(A_{1}\right),\dots,f_{n,k}\left(A_{n,k}\right)\right)-g_{\lambda_{1},\dots,\lambda_{n}}\left(f_{1}\left(A_{1}\right),\dots,f_{n}\left(A_{n}\right)\right)\|_{\mathcal{L}(X)}$ $\to 0 \quad (4.39)$

as $k \to \infty$. To show that this is indeed the case, we start with

$$\left\| P_{\lambda_{1},\dots,\lambda_{n}}^{m_{1},\dots,m_{n}}\left(f_{1,k}\left(A_{1}\right),\dots,f_{n,k}\left(A_{n}\right)\right)-P_{\lambda_{1},\dots,\lambda_{n}}^{m_{1},\dots,m_{n}}\left(f_{1}\left(A_{1}\right),\dots,f_{n}\left(A_{n}\right)\right)\right\|_{\mathcal{L}(X)} \right)$$

$$\leq \sum_{\pi\in S_{m}} \int_{\Delta_{m}(\pi)} \left\| C_{\pi(m)}^{k}\left(s_{\pi(m)}\right)\cdots C_{\pi(1)}^{k}\left(s_{\pi(1)}\right)-\left(4.40\right)\right\|_{\mathcal{L}(X)} \left(s_{\pi(m)}\right)\cdots C_{\pi(1)}\left(s_{\pi(1)}\right)\right\|_{\mathcal{L}(X)} \left(\lambda_{1}^{m_{1}}\times\dots\times\lambda_{n}^{m_{n}}\right)\left(ds_{1},\dots,ds_{m}\right).$$

(Note: We are making use, here, of the notation introduced in Subsection 2.1 and we use the notation $C_j^k(\cdot)$ to denote the particular operator $f_{j,k}(A_j)$. We have also applied Theorem 2.8 to write out the disentangling using continuous time-ordering measures.) We know that, in operator norm on $\mathcal{L}(X), f_{j,k}(A_j) \longrightarrow f_j(A_j)$ as $k \to \infty$. Also, there is no time-dependence in our operators. A standard argument shows that

$$C^k_{\pi(m)}\cdots C^k_{\pi(1)} - C_{\pi(m)}\cdots C_{\pi(1)} \xrightarrow[k \to \infty]{} 0$$

in operator norm, for any $\pi \in S_m$. Therefore

$$\sum_{\pi \in S_m} \int_{\Delta_m(\pi)} \left\| C_{\pi(m)}^k \left(s_{\pi(m)} \right) \cdots C_{\pi(1)}^k \left(s_{\pi(1)} \right) - C_{\pi(m)} \left(s_{\pi(m)} \right) \cdots C_{\pi(1)} \left(s_{\pi(1)} \right) \right\|_{\mathcal{L}(X)} \left(\lambda_1^{m_1} \times \cdots \times \lambda_n^{m_n} \right) (ds_1, \dots, ds_m)$$

$$= \sum_{\pi \in S_m} \left\| C_{\pi(m)}^k \cdots C_{\pi(1)}^k - C_{\pi(m)} \cdots C_{\pi(1)} \right\|_{\mathcal{L}(X)} \left(\lambda_1^{m_1} \times \cdots \times \lambda_n^{m_n} \right) (\Delta_m(\pi))$$

$$\xrightarrow{k \to \infty} 0$$

and so

$$\left\|P_{\lambda_{1},\dots,\lambda_{n}}^{m_{1},\dots,m_{n}}\left(f_{1,k}\left(A_{1}\right),\dots,f_{n,k}\left(A_{n}\right)\right)\right\|_{\mathcal{L}(X)} \xrightarrow{(4.41)} 0.$$

To verify (4.39), for arbitrary g (where $\{g_k\}_{k=0}^{\infty} \in \mathbb{A}_{\oplus}$, $g_k = g$ for all $k \in \mathbb{N}$), we apply the dominated convergence theorem. To see how this works, we first note that, given $\epsilon > 0$, there is a positive integer k_0 such that

$$\|\{g_k\}_{k=0}^{\infty}\|_{\mathbb{A}_{\oplus}} = \sup_{k \in \mathbb{N} \cup \{0\}} \|g\|_{\mathbb{A}_k} < \|g\|_{\mathbb{A}_{k_0}} + \epsilon.$$
(4.42)

Next, note that for any $k \in \mathbb{N}$,

It follows that, writing

$$g(z_1,\ldots,z_n) = \sum_{m_1,\ldots,m_n=0}^{\infty} a_{m_1,\ldots,m_n}^g z_1^{m_1}\cdots z_n^{m_n},$$

we have

$$\begin{split} \|g_{\lambda_{1},\dots,\lambda_{n}}\left(f_{1,k}\left(A_{1}\right),\dots,f_{n,k}\left(A_{n}\right)\right)-g_{\lambda_{1},\dots,\lambda_{n}}\left(f_{1}\left(A_{1}\right),\dots,f_{n}\left(A_{n}\right)\right)\|_{\mathcal{L}(X)} \\ &\leq \sum_{m_{1},\dots,m_{n}=0}^{\infty}|a_{m_{1},\dots,m_{n}}^{g}|R_{1,k}^{m_{1}}\cdots R_{n,k}^{m_{n}}+\sum_{m_{1},\dots,m_{n}=0}^{\infty}|a_{m_{1},\dots,m_{n}}^{g}|R_{1}^{m_{1}}\cdots R_{n}^{m_{n}} \\ &<\sum_{m_{1},\dots,m_{n}=0}^{\infty}|a_{m_{1},\dots,m_{n}}^{g}|R_{1,k_{0}}^{m_{1}}\cdots R_{n,k_{0}}^{m_{n}}+\epsilon \\ &+\sum_{m_{1},\dots,m_{n}=0}^{\infty}|a_{m_{1},\dots,m_{n}}^{g}|R_{1}^{m_{1}}\cdots R_{n}^{m_{n}} \\ &=\|g\|_{\mathbb{A}_{k_{0}}}+\epsilon+\|g\|_{\mathbb{A}_{0}}\,. \end{split}$$

Based on this computation, we define $\Lambda : \mathbb{N}^n \cup \{(0, \dots, 0)\} \longrightarrow \mathbb{R}$ by

$$\Lambda(m_1, \dots, m_n) := |a_{m_1, \dots, m_n}^g| \left[R_{1, k_0}^{m_1} \cdots R_{n, k_0}^{m_n} + R_{1, k}^{m_1} \cdots R_{n, k}^{m_n} \right] + \frac{\epsilon}{2^{m_1 + \dots + m_n + n}}.$$
(4.45)

The function Λ is clearly a summable dominating function for

$$\sum_{m_1,\dots,m_n=0}^{\infty} |a_{m_1,\dots,m_n}^g| \left\| P_{\lambda_1,\dots,\lambda_n}^{m_1,\dots,m_n} \left(f_{1,k} \left(A_1 \right),\dots,f_{n,k} \left(A_n \right) \right) - P_{\lambda_1,\dots,\lambda_n}^{m_1,\dots,m_n} \left(f_1 \left(A_1 \right),\dots,f_n \left(A_n \right) \right) \right\|_{\mathcal{L}(X)}.$$

We may therefore apply the dominated convergence theorem (for Bochner integrals) as well as (4.41) to obtain

$$\lim_{k \to \infty} \|g_{\lambda_{1},...,\lambda_{n}} (f_{1,k} (A_{1}), ..., f_{n,k} (A_{n})) -g_{\lambda_{1},...,\lambda_{n}} (f_{1} (A_{1}), ..., f_{n} (A_{n})) \|_{\mathcal{L}(X)}$$

$$\leq \lim_{k \to \infty} \sum_{m_{1},...,m_{n}=0}^{\infty} |a_{m_{1},...,m_{n}}^{g}| \left\| P_{\lambda_{1},...,\lambda_{n}}^{m_{1},...,m_{n}} (f_{1,k} (A_{1}), ..., f_{n,k} (A_{n})) - P_{\lambda_{1},...,\lambda_{n}}^{m_{1},...,m_{n}} (f_{1} (A_{1}), ..., f_{n} (A_{n})) \right\|_{\mathcal{L}(X)}$$

$$= 0. \qquad (4.46)$$

We have therefore established the following theorem.

Theorem 4.3. Let A_1, \ldots, A_n be closed densely defined linear operators in $\mathscr{C}(X)$, X a Banach space. Choose, for each $j \in \{1, \ldots, n\}$, an $f_j \in$

 $\mathscr{G}(A_j)$ (notation as above). Associate to each bounded operator $f_j(A_j)$, $j = 1, \ldots, n$, the continuous Borel probability measure λ_j on [0, T]. Furthermore, for each $j \in \{1, \ldots, n\}$, choose a sequence $\{f_{j,k}\}_{k=1}^{\infty}$ converging to f_j in the sense of (4.21) and (4.22). Construct the commutative Banach algebras \mathbb{A}_k , $k \in \mathbb{N} \cup \{0\}$ and \mathbb{A}_{\oplus} as in (4.31), (4.33) and (4.35). Let $\{g_k\}_{k=0}^{\infty} \in \mathbb{A}_{\oplus}$, $g_k \equiv g$, for all $k \in \mathbb{N} \cup \{0\}$. Then

$$\lim_{k \to \infty} \left\| \pi_k \left[\mathcal{T}_{\lambda_1,\dots,\lambda_n}^{\mathbb{A}_{\oplus}} \left(\left\{ g_k \left(f_{1,k} \left(A_1 \right)^{\sim},\dots,f_{n,k} \left(A_n \right)^{\sim} \right) \right\} \right) \right] - \mathcal{T}_{\lambda_1,\dots,\lambda_n} g \left(f_1 \left(A_1 \right)^{\sim},\dots,f_n \left(A_n \right)^{\sim} \right) \right\|_{\mathcal{L}(X)} = 0.$$
(4.47)

Remark 4.4. We remind the reader that this theorem also holds when the time-ordering measures are allowed to have a nonzero, finitely supported, discrete part. Due to combinatorial issues, the disentanglings are much more complicated and so the argument above is more involved, though reasonably straightforward. See, for example, Chapter 8 of [15] for some stability arguments using time-ordering measures with nonzero discrete parts.

4.4. Comments on other stability properties. The stability theorem, Theorem 4.3, addresses the stability of the operational calculus with respect to the functions used in the Taylor calculus approach to the use of unbounded operators. It is also possible to investigate the stability of the operational calculus with respect to the time-ordering measures – see [29, 28, 23, 22, 24, 27, 25, 21, 16] as well as [13], [14] for stability in a different, but similar, setting.

Though we will not delve into the details of the stability theory of the operational calculus with respect to the time-ordering measures, we will take the time to sketch out the essential ideas. Given operators (or operator-valued functions) A_1, \ldots, A_n and associated Borel (typically probability) time-ordering measures $\lambda_1, \ldots, \lambda_n$ on [0, T], we construct the disentangling algebra $\mathbb{D}(A_1^{\sim}, \ldots, A_n^{\sim})$ and obtain the disentangling map $\mathcal{T}_{\lambda_1, \ldots, \lambda_n} : \mathbb{D}(A_1^{\sim}, \ldots, A_n^{\sim}) \longrightarrow \mathcal{L}(X)$. To investigate the stability of the operational calculus with respect to the time-ordering measures, we select sequences $\{\lambda_{j,k}\}_{k=1}^{\infty}, j = 1, \ldots, n,$ of Borel (typically probability) measures on [0, T] converging weakly to λ_j . (By definition, this means that $\int_{[0,T]} f \, d\lambda_{j,k} \longrightarrow \int_{[0,T]} f \, d\lambda_j$ for all bounded continuous functions f on [0,T].) Since each n-tuple ($\lambda_{1,k}, \ldots, \lambda_{n,k}$, stability of FOC with respect to the time-ordering measures means that the family $\{\mathcal{T}_{\lambda_{1,k},\ldots,\lambda_{n,k}}\}_{k=1}^{\infty}$ of operational calculi has a limiting operational calculus; i.e.,

$$\lim_{k \to \infty} \mathcal{T}_{\lambda_{1,k},\dots,\lambda_{n,k}} f = \mathcal{T}_{\lambda_{1},\dots,\lambda_{n}} f$$

for all $f \in \mathbb{D}$. (This is stated somewhat informally.)

We can also study the stability of the operational calculus with respect to both the operators (operator-valued functions) and the time-ordering measures. This is the so-called *joint stability* of the operational calculus. In

Subsection 3.2, sequences $\{f_{j,k}\}_{k=1}^{\infty}$, j = 1, ..., n, give rise to $\{f_{j,k}(A_j)\}_{k=1}^{\infty}$, j = 1, ..., n, of bounded operators. If we also select weakly convergent sequences $\{\lambda_{j,k}\}_{k=1}^{\infty}$ of measures, joint stability asks if

 $\lim_{k \to \infty} \mathcal{T}_{\lambda_{1,k},\dots,\lambda_{n,k}} g\left(f_{1,k}(A_1),\dots,f_{n,k}(A_n)\right) = \mathcal{T}_{\lambda_1,\dots,\lambda_n} g\left(f_1(A_1),\dots,f_n(A_n)\right).$

Joint stability of the operational calculus is studied for the abstract approach to the operational calculus found in [15] and the results therein can be applied in this setting where the sequences of operators are $\{f_{j,k}(A_j)\}_{k=1}^{\infty}$, $j = 1, \ldots, n$. It should be noted that the setting of [15] allows only bounded operators (with the exception of the generator of a (C_0) semigroup). In the next section, we study joint stability for the operational calculus when using analytic families of closed unbounded operators. Just as with the operational calculus in the setting of [15], we select sequences of operators as well as sequences of time-ordering measures. However, in the setting of this paper, the sequences of operators are sequences of closed and unbounded operators and so we use the idea of generalized convergence. This complicates the study of joint stability somewhat, giving the result a slightly different flavor than the stability theory in the "standard" setting of [15]. The discussion of joint stability for the operational calculus using analytic families found in Subsection 5.2, below, relies very much on the material found in Subsections 3.2–3.4, where the "heavy lifting" is carried out. Indeed, with the formalism introduced in these subsections, the discussion of joint stability is quite straightforward and brief.

5. FOC for unbounded operators using analytic families

In this section we turn to the operational calculus using analytic families of unbounded closed operators. (See Subsection 2.7 for necessary definitions.) To this end, let D_0 be an open neighborhood of $0 \in \mathbb{C}$. Let $T_1(z), \ldots, T_n(z)$, $z \in D_0$, be analytic families of unbounded closed operators on the Banach space X. Furthermore, for each $j \in \{1, \ldots, n\}$ we will assume that, in the definition from Subsection 2.7, the Banach space Z equals X and so there are bounded analytic families $U_j(z), V_j(z) \in \mathcal{L}(X)$ with $U_j(z)$ injective onto $\mathcal{D}(T_j(z))$ and

$$T_j(z)U_j(z) = V_j(z).$$
 (5.1)

It is the operator-valued functions $V_j(z)$ which will be used in the operational calculus. To this end, we choose T > 0 such that $[0,T] \subset D_0 \cap \mathbb{R}$. For each $j = 1, \ldots, n$, associate Borel probability measures (time-ordering measures) λ_j on [0,T] to the operator-valued function $V_j(\cdot) : [0,T] \longrightarrow \mathcal{L}(X)$. (We assume, as always, that the discrete parts of the time-ordering measures, when present, are finitely supported.) As in Subsection 2.2, we construct the Banach algebra $\mathbb{A}(R_1, \ldots, R_n)$ with the radii $R_j, j = 1, \ldots, n$, given by

$$R_j := \int_{[0,T]} \|V_j(s)\|_{\mathcal{L}(X)} \lambda_j(ds), \qquad (5.2)$$

which is assumed finite for each j. We also construct the disentangling algebra $\mathbb{D}(V_1(\cdot)^{\sim}, \ldots, V_n(\cdot)^{\sim})$, again following Subsection 2.2. Given $g \in \mathbb{D}(V_1(\cdot)^{\sim}, \ldots, V_n(\cdot)^{\sim})$, we write, in the usual way,

$$g\left(V_1(\cdot)^{\sim},\ldots,V_n(\cdot)^{\sim}\right)$$
$$=\sum_{m_1,\ldots,m_n=0}^{\infty}a_{m_1,\ldots,m_n}^g\left[V_1(\cdot)^{\sim}\right]^{m_1}\cdots\left[V_n(\cdot)^{\sim}\right]^{m_n}$$

The disentangling map is defined exactly as in Subsection 2.2. Indeed,

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n}g\left(V_1(\cdot)^{\sim},\dots,V_n(\cdot)^{\sim}\right)$$

= $g_{\lambda_1,\dots,\lambda_n}\left(V_1(\cdot),\dots,V_n(\cdot)\right)$
= $\sum_{m_1,\dots,m_n=0}^{\infty} a_{m_1,\dots,m_n}^g P_{\lambda_1,\dots,\lambda_n}^{m_1,\dots,m_n}\left(V_1(\cdot),\dots,V_n(\cdot)\right)$ (5.3)

exactly as in Subsection 2.2. Consequently, the operational calculus in this setting has all of the properties outlined in Subsection 2.2. The reader will note that, just as with the use of Taylor's calculus, using analytic families is ultimately very easy, at least in the definition of the operational calculus. This is due to the fact that we are "hiding" the unbounded operators T(z) using the bounded operators V(z). This leads us to the following comments.

While unbounded operators do not appear explicitly in the disentangled operator, they are present in the definition of the operator-valued functions (or, more carefully, the boundedly analytic families) $V_j(\cdot)$, j = 1, ..., n. The injective operator-valued functions $U_j(\cdot)$ serve to take a vector from X and map it to the domain of the operator $T_j(\cdot)$. To be a bit more precise, take $j_1, \ldots, j_m \in \{1, \ldots, n\}$. We may write, for $\phi \in X$,

$$V_{j_m}(s_m)\cdots V_{j_1}(s_1)\phi = [T_{j_m}(s_m)U_{j_m}(s_m)]\cdots [T_{j_1}(s_1)U_{j_1}(s_1)]\phi.$$

Since $U_{j_1}(s_1)\phi \in \mathcal{D}(T_{j_1}(s_1))$, it makes sense to apply $T_{j_1}(s_1)$. Similarly,

$$U_{j_2}(s_2) (T_{j_1}(s_1)U_{j_1}(s_1)) \in \mathcal{D}(T_{j_2}(s_2))$$

and so we can apply $T_{j_2}(s_2)$ to the vector $U_{j_2}(s_2) (T_{j_1}(s_1)U_{j_1}(s_1)\phi)$. This process continues through the entire operator product. So, even though the operator products in the disentangling series do not involve the unbounded operators alone, the structure of the analytic families allow the unbounded operators to appear in a natural way in the operator products present in the disentangling series – the disentangled monomials in (5.3).

5.1. Stability results for FOC using analytic families. We investigate here the stability of FOC when using analytic families of operators. Fix, for the moment, $j \in \{1, ..., n\}$. Select a sequence $\{T_{j,k}(z)\}_{k=1}^{\infty}$ of analytic families in $D_0 \subseteq \mathbb{C}$ (D_0 is as above). We take Z = X in the definition of analytic family, Definition 2.32. Furthermore, we will assume that all of our analytic families $T_n(z)$, T(z) satisfy Definition 2.34. Then there are, for each $k \in \mathbb{N}$, bounded analytic families $U_{j,k}(z), V_{j,k}(z) \in \mathcal{L}(X)$, with $U_{j,k}(z)$ injective onto $\mathcal{D}(T_{j,k}(z))$ and

$$T_{j,k}(z)U_{j,k}(z) = V_{j,k}(z).$$
 (5.4)

Fix, along with our j, a $z \in D_0$. Then

$$\mathcal{G}(T_{j,k}(z)) = \{ (U_{j,k}(z)\phi, V_{j,k}(z)\phi) : \phi \in X \}$$
(5.5)

and

$$\mathcal{G}(T_j(z)) = \left\{ (U_j(z)\phi, V_j(z)\phi) : \phi \in X \right\}.$$
(5.6)

We assume that $T_{j,k}(z) \longrightarrow T_j(z)$ as $k \to \infty$ in the generalized sense of Subsection 2.5. Then

$$\hat{\delta}\left(\mathcal{G}(T_{j,k}(z)), \mathcal{G}(T_j(z))\right) \longrightarrow 0$$
(5.7)

as $k \to \infty$. It follows at once that

$$\delta\left(\mathcal{G}(T_{j,k}(z),\mathcal{G}(T_j(z))) \xrightarrow[k \to \infty]{} 0 \right)$$
(5.8)

and

$$\delta\left(\mathcal{G}(T_j(z)), \mathcal{G}(T_{j,k}(z))\right) \xrightarrow[k \to \infty]{} 0.$$
 (5.9)

We will use (5.8). Define, for each $k \in \mathbb{N}$ and $z \in D_0$,

$$\delta_k(z) := \delta\left(\mathcal{G}(T_{j,k}(z)), \mathcal{G}(T_j(z))\right).$$
(5.10)

Then $\delta_k(z)$ is the smallest positive real number for which

$$\operatorname{dist}\left(u, \mathcal{G}(T_{j,k}(z))\right) \le \delta_k(z) \|u\|_{\mathcal{G}(T_j(z))}$$
(5.11)

for all $u \in \mathcal{G}(T_j(z))$ (see (2.25)). (Note that the graph norm for an operator $T \in \mathscr{C}(X,Y)$ is $\|(f,Tf)\|_{\mathcal{G}(T)} := (\|f\|_X^2 + \|Tf\|_Y^2)^{1/2}$. Indeed, for Banach spaces X and Y, the product $X \times Y$ becomes a Banach space under the norm $\|(x,y)\|_{X \times Y} = (\|x\|_X^2 + \|y\|_Y^2)^{1/2}$.) Fix $u_j(z) \in \mathcal{G}(T_j(z))$ and, for each $k \in \mathbb{N}$, choose $v_{j,k}(z) \in \mathcal{G}(T_{j,k}(z))$ with

$$\|u_{j}(z) - v_{j,k}(z)\|_{X \times X} < \operatorname{dist}\left(u(z), \mathcal{G}(T_{j,k}(z))\right) + \frac{1}{k} \le \delta_{k}(z)\|u(z)\|_{\mathcal{G}(T_{j}(z))} + \frac{1}{k}$$
(5.12)

We obtain a sequence $\{v_{j,k}(z)\}_{k=1}^{\infty}$ in $X \times X$ for which

$$\|u_j(z) - v_{j,k}(z)\|_{X \times X} \xrightarrow[k \to \infty]{} 0, \qquad (5.13)$$

because $\delta_k(z) \to 0$ as $k \to \infty$. Now, using (5.5) and (5.6), we may write

$$u_j(z) = (U_j(z)\phi_j(z), V_j(z)\phi_j(z))$$
 (5.14)

for some $\phi_i(z) \in X$ and

$$v_{j,k}(z) = (U_{j,k}(z)\psi_{j,k}(z), V_{j,k}(z)\psi_{j,k}(z))$$
(5.15)

for some $\psi_{j,k}(z) \in X$. Hence

$$\|U_j(z)\phi_j(z) - U_{j,k}(z)\psi_{j,k}(z)\|_{X \times X} \xrightarrow[k \to \infty]{} 0$$
(5.16)

and

$$\|V_j(z)\phi_j(z) - V_{j,k}(z)\psi_{j,k}(z)\|_{X \times X} \xrightarrow[k \to \infty]{} 0.$$
(5.17)

We remind the reader that these limits are for our fixed $z \in D_0$; i.e., these limits are pointwise limits.

We now address stability of the operational calculus for this setting. In order to do so, we need to switch our attention to the formalism of Feynman's operational calculus in arbitrary Banach algebras which was developed in Subsection 3.2. Therefore, we will start by assuming that X is a Banach algebra, and, to avoid unnecessary (in the view of the author) technicalities, we will assume that X is separable. (See Subsection 3.1, above.)

To continue the development of the stability theory in the current setting, we define, for each $j \in \{1, ..., n\}$,

$$W_j : [0,T] \longrightarrow X,$$

 $s \longmapsto V_j(s)\phi_j(s),$ (5.18)

and

$$W_{j,k} : [0,T] \longrightarrow X,$$

$$s \longmapsto V_{j,k}(s)\psi_{j,k}(s).$$
(5.19)

Because our analytic families satisfy Definition 2.34, we now appeal to the discussion contained in Subsection 2.8 to conclude that the maps $W_j(\cdot)$ and $W_{j,k}(\cdot)$ are continuous on D_0 and so on $[0,T] \subseteq D_0 \cap \mathbb{C}$. Also, it follows from Equation (5.17) that

$$W_{j,k}(s) \xrightarrow[k \to \infty]{} W_j(s)$$
 (5.20)

in X-norm for each $s \in [0, T]$ and $j = 1, \ldots, n$.

We wish to apply the ideas of Theorem 8.5.13 of [15] concerning the stability of the operational calculus with respect to the operator-valued functions. In order to use the ideas of this theorem, we need our functions $W_j(\cdot)$ and $W_{j,k}(\cdot)$ to be λ_j -measurable in the sense discussed in Subsection 3.1. Indeed, because our functions $W_j(\cdot)$ and $W_{j,k}(\cdot)$ are continuous and since X is separable, it follows that our functions are all measurable in the appropriate sense. The analogue of Theorem 8.5.13 which we will use is stated as follows.

Theorem 5.1 ([15, Theorem 8.5.13 Analog]). Let X be a separable Banach algebra. Let $W_j : [0,T] \to X$, j = 1, ..., n, and associate to each $W_j(\cdot)$ the Borel probability measure λ_j on [0,T] where $\lambda_j = \mu_j + \eta_j$, with μ_j a continuous Borel measure and with η_j a discrete finitely supported measure on [0,T]. Let $\{\tau_1, \ldots, \tau_h\}$ with $0 < \tau_1 < \cdots < \tau_h < T$ be the union of the supports of the discrete measures η_1, \ldots, η_n , and write

$$\eta_j = \sum_{i=1}^h p_{ji} \delta_{\tau_i}$$

for each j = 1, ..., n. We assume that $W_j(\cdot)$ is λ_j -measurable for each j = 1, ..., n. Furthermore, choose sequences $\{W_{j,k}(\cdot)\}_{k=1}^{\infty}$, j = 1, ..., n, of λ_j -measurable functions from [0,T] to X such that, for each j = 1, ..., n, $W_{j,k}(\cdot) \to W_j(\cdot) \lambda_j$ -almost everywhere in X-norm as $k \to \infty$. Assume that, for each j = 1, ..., n,

$$\sup \{ \|W_{j,k}(s)\|_X : k \in \mathbb{N}, s \in [0,T] \} < \infty$$
(5.21)

and

$$\sup \{ \|W_j(s)\|_X : s \in [0, T] \} < \infty.$$
(5.22)

Define real numbers r_j , $r_{j,k}$ for j = 1, ..., n and $k \in \mathbb{N}$ by

$$r_j := \int_{[0,T]} \|W_j(s)\|_X \,\lambda_j(ds)$$
(5.23)

and

$$r_{j,k} := \int_{[0,T]} \|W_{j,k}(s)\|_X \,\lambda_j(ds).$$
(5.24)

Construct the commutative Banach algebras $\mathbb{A}(r_{1,k},\ldots,r_{n,k})$, $\mathbb{A}(r_1,\ldots,r_n)$ and the associated disentangling algebras

$$\mathbb{D}_k := \mathbb{D}\left((W_{1,k}(\cdot), \lambda_1)^{\sim}, \dots, (W_{n,k}(\cdot), \lambda_n)^{\sim} \right)$$
(5.25)

and

$$\mathbb{D}_0 := \mathbb{D}\left((W_1(\cdot), \lambda_1)^{\sim}, \dots, (W_n(\cdot), \lambda_n)^{\sim} \right).$$
(5.26)

Finally, let

$$\mathbb{D}_{\oplus} := \bigoplus_{k \in \mathbb{N} \cup \{0\}} \mathbb{D}_k \tag{5.27}$$

be the direct-sum Banach algebra. Let π_k , $k \in \mathbb{N} \cup \{0\}$, be the canonical projection of \mathbb{D}_{\oplus} onto \mathbb{D}_k .

Then, for any $\theta_f \in \mathbb{D}_{\oplus}$, $\theta_f = (f, f, \ldots)$, we have

$$\mathcal{T}_{\lambda_1,\dots,\lambda_n}\left(\pi_k(\theta_f)\right) \to \mathcal{T}_{\lambda_1,\dots,\lambda_n}\left(\pi_0(\theta_f)\right)$$
(5.28)

in norm on X as $k \to \infty$. More transparently, we may write the above as

$$\lim_{k \to \infty} \mathcal{T}_{\lambda_1, \dots, \lambda_n} f\left(W_{1,k}(\cdot), \dots, W_{n,k}(\cdot)\right) = \mathcal{T}_{\lambda_1, \dots, \lambda_n} f\left(W_1(\cdot), \dots, W_n(\cdot)\right)$$
(5.29)

in norm on X.

The proof of this theorem proceeds in exactly the same way as in [15], working in the Banach algebra X instead of the Banach algebra $\mathcal{L}(X)$.

To use Theorem 5.1, we note that the functions $W_j(\cdot)$ and $W_{j,k}(\cdot)$ defined in (5.18) and (5.19), respectively are, continuous on [0, T]. Since X is separable, each $W_j(\cdot)$ is λ_j -measurable and each $W_{j,k}(\cdot)$ is λ_j -measurable. It has been observed in Equation (5.20) that $W_{j,k}(s) \to W_j(s)$ in X-norm for every $s \in [0, T]$ and each $j = 1, \ldots, n$. Furthermore, from the continuity of $W_j(\cdot)$ and $W_{j,k}(\cdot)$ on [0, T], the norm-boundedness conditions (5.21) and

(5.22) are clearly satisfied. By a direct application of Theorem 5.1 with $\theta_f = (f, f, f, \ldots) \in \mathbb{D}_{\oplus}$,

$$\lim_{k \to \infty} f_{\lambda_1, \dots, \lambda_n} \left(W_{1,k}(\cdot), \dots, W_{n,k}(\cdot) \right) = f_{\lambda_1, \dots, \lambda_n} \left(W_1(\cdot), \dots, W_n(\cdot) \right)$$

in norm on X. (Recall that $f_{\lambda_1,\ldots,\lambda_n} =: \mathcal{T}_{\lambda_1,\ldots,\lambda_n} f$.)

5.2. Joint stability of F.O.C. in the setting of analytic families. In this section, we address the stability of the operational calculus with respect to generalized convergence of sequences of analytic families of operators and weak convergence of sequences of time-ordering measures. Of course, in the previous subsection, we addressed the stability of the operational calculus in the setting of analytic families of closed operators and so this subsection adds weakly convergent sequences of time-ordering measures. We will use the notation introduced in Subsection 2.8 and in Subsection 5.1, just above.

To begin, we will follow the general setup detailed at the beginning of the current section. Hence, we let $D_0 \subseteq \mathbb{C}$ be an open neighborhood of $0 \in \mathbb{C}$. We also take $T_1(z), \ldots, T_n(z)$ to be analytic families of unbounded closed operators on the separable Banach space X. For each $j = 1, \ldots, n$, we will assume, in the definition of analytic family found in Subsection 2.7, that the Banach space Z equals X and so there are boundedly analytic functions $U_j(\cdot), V_j(\cdot)$ mapping D_0 into $\mathcal{L}(X)$ with $U_j(\cdot)$ injective onto the domain $\mathcal{D}(T_j(z))$ of $T_j(z), z \in D_0$ and

$$T_j(z)U_j(z) = V_j(z),$$

for $z \in D_0$. We now, just as before, choose T > 0 such that $[0,T] \subseteq D_0 \cap \mathbb{R}$ and for each $j = 1, \ldots, n$, associate to each $V_j(\cdot)$ a Borel probability measure λ_j on [0,T]. (As always, if a discrete part of a time-ordering measure is present in any of the λ_j , we assume it to be finitely supported.) Define, as before,

$$R_j := \int_{[0,T]} \|V_j(s)\|_{\mathcal{L}(X)} \,\lambda_j(ds)$$

which we assume finite for each j = 1, ..., n. Construct the disentangling algebra $\mathbb{D}(V_1(\cdot)^{\sim}, ..., V_n(\cdot)^{\sim})$ and obtain, given $g \in \mathbb{D}(V_1(\cdot)^{\sim}, ..., V_n(\cdot)^{\sim})$, the disentangled operator $g_{\lambda_1,...,\lambda_n}(V_1(\cdot), ..., V_n(\cdot))$ as in (3.1), above.

the disentangled operator $g_{\lambda_1,\ldots,\lambda_n}(V_1(\cdot),\ldots,V_n(\cdot))$ as in (3.1), above. We now select, for each $j = 1,\ldots,n$, a sequence $\{T_{j,k}(\cdot)\}_{k=1}^{\infty}$ of analytic families in D_0 , D_0 being just as above. Also, as above, take the Banach space Z in the definition of analytic families to be equal to X. We will also assume that our analytic families $T_1(\cdot),\ldots,T_n(\cdot)$ and $T_{1,k}(\cdot),\ldots,T_{n,k}(\cdot)$ satisfy Definition 2.34; i.e., they are uniform on D_0 . There are, for each $k \in \mathbb{N}$, boundedly analytic families $U_{j,k}(\cdot), V_{j,k}(\cdot) \in \mathcal{L}(X)$ with $U_{j,k}(\cdot)$ injective onto $\mathcal{D}(T_{j,k}(z))$ and

$$T_{j,k}(z)U_{j,k}(z) = V_{j,k}(z)$$

for each $z \in D_0$.

We now follow the exposition at the beginning of Subsection 5.1 to obtain the maps

$$W_j : [0,T] \to X,$$

 $s \mapsto V_j(s)\phi_j(s)$

and

$$W_{j,k}: [0,T] \to X,$$

$$s \mapsto V_{j,k}(s)\psi_{j,k}(s)$$

(See equations (5.18) and (5.19).) By virtue of our assumed uniformity for the analytic families we appeal to the discussion of Subsection 2.8 to conclude that $W_j(\cdot)$ and $W_{j,k}(\cdot)$ are continuous and, from (5.17), $W_{j,k}(\cdot) \longrightarrow W_j(\cdot)$ pointwise in X-norm as $k \to \infty$.

Now that we have our sequences $\{W_{j,k}(\cdot)\}_{k=1}^{\infty}$ of continuous X-valued functions, we turn to our time-ordering measures. For each $j = 1, \ldots, n$, select a sequence $\{\lambda_{j,k}\}_{k=1}^{\infty}$ of Borel probability measures on [0, T] such that each measure is allowed to have a discrete part with finite support (if present) and $\lambda_{j,k} \rightarrow \lambda_j$ as $k \rightarrow \infty$. For each $j = 1, \ldots, n$ we associate the Borel measures $\lambda_{j,k}$ and λ_j to the X-valued functions $W_{j,k}(\cdot), W_j(\cdot), k \in \mathbb{N}$.

To obtain our joint stability result, we let our continuous functions $W_j(\cdot)$, $j = 1, \ldots, n$, play the role of the functions $Y_j(\cdot)$, $j = 1, \ldots, n$, of Subsection 3.4. This being done, we can apply Theorem 3.16 to conclude that, for $f \in \mathbb{A}(M_1, \ldots, M_n)$ $(M_1, \ldots, M_n$ are defined just as in (3.57)),

$$\lim_{k \to \infty} f_{\lambda_{1,k},\dots,\lambda_{n,k}} \left(W_{1,k}(\cdot),\dots,W_{n,k}(\cdot) \right) = f_{\lambda_1,\dots,\lambda_n} \left(W_1(\cdot),\dots,W_n(\cdot) \right).$$
(5.30)

This is our joint stability result for the operational calculus in the setting of analytic families of operators. The reader will observe that, as has been the case throughout our investigations of the operational calculus using unbounded operators, the unbounded operators here are "hiding" in the functions $W_j(\cdot)$ and $W_{j,k}(\cdot)$ via the boundedly analytic functions $U_{j,k}(\cdot), U_j(\cdot), V_{j,k}(\cdot)$ and $V_j(\cdot)$. Moreover, since $W_j(s) = V_j(s)\phi_j(s)$ and $W_{j,k}(s) = V_{j,k}(s)\psi_{j,k}(s)$ and because the vectors $\phi_j(s)$ and $\psi_{j,k}(s)$ are most certainly not arbitrary (see (5.14) and (5.15), respectively), the joint stability result stated here is not as general as one may hope. Nevertheless, it is satisfying to find an analogue of joint stability in the general setting of closed unbounded operators.

References

 CONWAY, JOHN B. A course in functional analysis. Second edition. Graduate Texts in Mathematics, 96. Springer-Verlag, New York, 1990. xvi + 399 pp. ISBN: 0-387-97245-5. MR1070713 (91e:46001), Zbl 0706.46003, doi: 10.1007/978-1-4757-3828-5. 414, 415

- [2] DUNFORD, NELSON. Spectral theory. I. Convergence to projections. Trans. Amer. Math. Soc. 54 (1943), 185–217. MR0008642 (5,39c), Zbl 0063.01185, doi:10.2307/1990329.379
- [3] DUNFORD, NELSON. Spectral theory. Bull. Amer. Math. Soc. 49 (1943), 637–651.
 MR0008643 (5,39d), Zbl 0063.01184, doi: 10.1090/S0002-9904-1943-07965-7. 379
- [4] DUNFORD, NELSON; SCHWARTZ, JACOB T. Linear Operators. Part I. General theory. With the assistance of William G. Bade and Robert G. Bartle. Reprint of the 1958 original. Wiley Classics Library. A Wiley-Interscience Publication. John Wiley & Sons, Inc., New York, 1988. xiv+858 pp. ISBN: 0-471-60848-3. MR1009162 (90g:47001a), Zbl 0635.47001. 392, 403, 422
- [5] FEYNMAN, RICHARD P. An operator calculus having applications in quantum electrodynamics. *Phys. Rev.* 84 (1951), 108–128. MR0044379 (13,410e), Zbl 0044.23304, doi: 10.1103/PhysRev.84.108. 378, 381
- [6] HEWITT, EDWIN; STROMBERG, KARL. Real and abstract analysis. A modern treatment of the theory of functions of a real variable. Springer-Verlag, New York-Heidelberg, 1965. x+476 pp. ISBN: 978-0-387-90138-1. MR00367121 (51 #3363), Zbl 0307.28001, doi: 10.1007/978-3-662-29794-0. 404
- [7] HYTÖNEN, TUOMAS; VAN NEERVEN, JAN; VERAAR, MARK; WEIS, LUTZ. Analysis in Banach spaces, Vol. I. Martingales and Littlewood–Paley Theory. Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge/ A Series of Modern Surveys in Mathematics, 63. Springer, Cham, 2016. xvi+614 pp. ISBN: 978-3-319-48519-5; 978-3-319-48520-1. MR3617205, Zbl 1366.46001, doi: 10.1007/978-3-319-48520-1. 401, 402, 403
- [8] JEFFERIES, BRIAN; JOHNSON, GERALD W. Feynman's operational calculi for noncommuting operators: definitions and elementary properties. *Russ. J. Math. Phys.* 8 (2001), no. 2, 153–171. MR1928726 (2003i:47019), Zbl 1186.81082. 379, 382, 383, 384, 385, 391
- JEFFERIES, BRIAN; JOHNSON, GERALD W. Feynman's operational calculus for families of noncommuting operators: tensor products, ordered supports, and the disentangling of an exponential factor. *Mat. Zametki* **70** (2001), no. 6, 815–838.
 MR1887249) (2003a:47037), Zbl 1030.47008, doi:10.1023/A:1012903732597. 379, 382, 383, 384, 385
- [10] JEFFERIES, BRIAN; JOHNSON, GERALD W. Feynman's operational calculi for noncommuting operators: The monogenic calculus. Adv. Appl. Clifford Algebras 11 (2001), no.2, 239–264. MR1956400 (2003k:47011), Zbl 1096.47501, doi:10.1007/BF03042315.379, 382, 383, 384, 385
- [11] JEFFERIES, BRIAN; JOHNSON, GERALD W. Feynman's operational calculi for noncommuting operators: spectral theory. *Infin. Dimens. Anal. Quantum Probab. Relat. Top.* 5 (2002), no. 2, 171–199. MR1914834 (2003g:47031), Zbl 1096.47502, doi:10.1142/S021902570200078X. 379, 382, 383, 384, 385
- [12] JEFFERIES, BRIAN; JOHNSON, GERALD W., NIELSEN, LANCE. Feynman's operational calculi for time dependent noncommuting operators. J. Korean Math. Soc. 38 (2001), no. 2, 193–226. MR01817617 (2001m:81154), Zbl 0984.47013. 379, 382, 384, 385
- [13] JOHNSON, GERALD W.; LAPIDUS, MICHEL L. Generalized Dyson series, generalized Feynman diagrams, the Feynman integral and Feynman's operational calculus. *Mem. Amer. Math. Soc.* 62 (1986), no. 351, vi+78 pp. ISBN: 978-1-4704-0767-4. MR0849943 (88f:81034), Zbl 0638.28009, doi: 10.1090/memo/0351. 380, 435
- [14] JOHNSON, GERALD W., LAPIDUS, MICHEL L. The Feynman integral and Feynman's operational calculus. Oxford Mathematical Monographs. Oxford Science Publications. The Clarendon Press, Oxford University Press, New York, 2000.

xviii+771 pp. ISBN: 0-19-853574-0. MR1771173 (2001i:58015), Zbl 1027.46002. 378, 380, 387, 393, 394, 395, 426, 429, 435

- [15] JOHNSON, GERALD W., LAPIDUS, MICHEL L., NIELSEN, LANCE. Feynman's operational calculus and beyond. Noncommutativity and time-ordering. Oxford Mathematical Monographs. Oxford University Press, Oxford, 2015. xiv+369 pp. ISBN: 978-0-19-870249-8. MR3381096, Zbl 1330.81010, doi:10.1093/acprof:oso/9780198702498.001.0001. 379, 380, 382, 384, 385, 386, 387, 388, 390, 391, 413, 425, 426, 427, 428, 429, 432, 435, 436, 439, 440
- [16] JOHNSON, GERALD W., NIELSEN, LANCE. A stability theorem for Feynman's operational calculus. Stochastic Processes, Physics and Geometry: New Interplays. II. (Leipzig, 1999), 351–365. CMS Conference Proceedings, 29, American Mathematical Society, Providence, RI, 2000. MR1803428 (2001k:47020), Zbl 0983.46056. 380, 435
- [17] JOHNSON, GERALD W., NIELSEN, LANCE. Feynman's operational calculi: blending instantaneous and continuous phenomena in Feynman's operational calculi. Stochastic Analysis and Mathematical Physics (SAMP/ANESTOC 2002), 229-254. World Sci. Publ., River Edge, NJ, 2004. MR2115939 (2005j:81096), doi: 10.1142/9789812702364_0012. 385, 388, 390
- [18] KADISON, RICHARD V.; RINGROSE, JOHN R. Fundamentals of the theory of operator algebras. I. Elementary theory. Reprint of the 1983 original. Graduate Studies in Mathematics, 15. American Mathematical Society, Providence, RI, 1997. xvi+398 pp. ISBN: 0-8218-0819-2 MR1468229 (98f:46001a), Zbl 0888.46039, doi: 10.1090/gsm/015 427
- [19] KATO, TOSIO. Perturbation theory for linear operators. Reprint of the 1980 edition. Classics in Mathematics. Springer-Verlag, Berlin, 1995. xxii+619 pp. ISBN: 3-540-58661-X. (Originally published as Vol. 132 of the Grundlehren der mathematischen Wissenschaften.) MR1335452 (96a:47025), Zbl 0836.47009. 378, 380, 381, 392, 393, 397, 398, 399
- [20] LAPIDUS, MICHEL L. Product formula for imaginary resolvents with application to a modified Feynman integral. J. Functional Anal. 63 (1985), 261–275. MR0808263 (87c:47059), Zbl 0601.47025, doi: 10.1016/0022-1236(85)90088-6. 378, 380, 426
- [21] NIELSEN, LANCE. Stability Properties for Feynman's Operational Calculus. Thesis (Ph.D.), The University of Nebraska-Lincoln. 1999. 102 pp. ISBN: 978-0599-37910-7 MR2699342. 380, 384, 385, 435
- [22] NIELSEN, LANCE. Stability properties of Feynman's for exponential functions of noncommuting operators. Acta Appl. Math. 74 (2002), no. 3, 265–292. MR1942532 (2003k:58013), Zbl 1041.46053, doi: 10.1023/A:1021155700823. 380, 435
- [23] NIELSEN, LANCE. Stability properties for Feynman's operational calculus in the combined continuous/discrete setting. Acta Appl. Math. 88 (2005), no. 1, 47–79. MR2163746 (2006d:47031), Zbl 1089.47018, doi:10.1007/s10440-005-6699-0. 380, 435
- [24] NIELSEN, LANCE. Time dependent stability for Feynman's operational calculus. *Rocky Mountain J. Math.* 35 (2005), no. 4, 1347–1368. MR2178990 (2006m:47025), Zbl 1101.47011, doi: 10.1216/rmjm/1181069689. 380, 432, 435
- [25] NIELSEN, LANCE. Weak convergence and vector-valued functions: improving the stability theory of Feynman's operational calculi. *Math. Phys. Anal. Geom.* 10 (2007), no. 4, 271–295. MR2386254 (2009k:47016), Zbl 1143.47014, doi:10.1007/s11040-007-9033-4. 380, 435
- [26] NIELSEN, LANCE. Feynman's operational calculi: using Cauchy's integral formula. New York J. of Math. 16 (2010), 463–488. MR2740586 (2012a:32001), Zbl 1228.32005. 415, 418

- [27] NIELSEN, LANCE. Weak convergence and Banach space-valued functions: improving the stability theory of Feynman's operational calculi. *Math. Phys. Anal. Geom.* 14 (2011), no. 4, 279–294. MR2863554 (2012j:60008), Zbl 1246.60011, doi:10.1007/s11040-011-9097-z. 380, 435
- [28] NIELSEN, LANCE. Towards a comprehensive stability theory for Feynman's operational calculus: the time independent setting. Acta. Appl. Math. 138 (2015), 59–79. MR3365581, Zbl 1319.81062, doi: 10.1007/s10440-014-9957-1. 380, 429, 432, 435
- [29] NIELSEN, LANCE. Towards a comprehensive stability theory for Feynman's operational calculus: the time-dependent setting. Acta. Appl. Math. 152 (2017), 1–31. MR3721655, Zbl 1382.44005, doi:10.1007/s10440-017-0109-2. 380, 432, 435
- [30] NIELSEN, LANCE. Combining continuous and discrete phenomena for Feynman's operational calculus in the presence of a (C_0) semigroup and Feynman–Kac formulas with Lebesgue–Stieltjes measures. *Integral Equations Operator Theory* **90** (2018), no. 1, paper no. 12, 61 pp. MR3769722, Zbl 1388.58006, doi: 10.1007/s00020-018-2428-8. 379, 392
- [31] REED, MICHAEL; SIMON, BARRY. Methods of modern mathematical physics.
 I. Functional analysis. Second edition. Academic Press, Inc. [Harcourt Brace Jovanovich, Publishers], New York, 1980. xv+400 pp. ISBN: 0-12-585050-6. MR0751959 (85e:46002), Zbl 0459.46001. 392, 398
- [32] SIMON, BARRY. Operator theory. A comprehensive course in analysis, part 4. American Mathematical Society, Providence, R. I., 2015. MR3364494, Zbl 1334.00003, doi: 10.1090/simon/004. 392
- [33] TAYLOR, ANGUS E. Analysis in complex Banach spaces. Bull. Amer. Math. Soc.
 49 (1943), 652–669. MR0008644 (5,39e), Zbl 0063.07311, doi: 10.1090/S0002-9904-1943-07968-2. 379
- [34] TAYLOR, ANGUS E. Spectral theory of closed distributive operators. Acta. Math. 84 (1951), 189–224. MR0040582 (12,717c), Zbl 0042.12103, doi:10.1007/BF02414854.

378, 379, 384, 395, 396, 423, 424

(Lance Nielsen) DEPARTMENT OF MATHEMATICS, CREIGHTON UNIVERSITY, OMAHA, NE 68178, USA

lnielsen@creighton.edu

This paper is available via http://nyjm.albany.edu/j/2020/26-20.html.